

August 7, 2014

Oregon Public Utility Commission Attention: Filing Center P.O. Box 2148 550 Capitol Street N.E., Suite 215 Salem, OR 97301-2551 Fax (503) 373-7752

RE: UM 1431 Compliance Letter - Order No. 10-067

Dear Commission:

In compliance with Appendix A, condition # 2 of the above referenced order, Frontier Communications Northwest Inc is providing a copy of its Quarterly 10-Q report filing and inter-company receivables, payables and dividends report for the second quarter 2014.

The quarterly inter-company receivables, payables and dividends report is considered confidential and should be treated as such. Pursuant to Order 09-197, the "Confidential" document is being sent via US Mail in a sealed envelope with the proper labeling.

If you have any questions, please contact Renee Willer at (503) 629-2459 or Renee.Willer@ftr.com or me.

Sincerely,

øseph D. Chicome

Manager, Compliance and Reporting

916-686-3588

joe.chicoine@ftr.com

Enclosures

cc: Renee Willer

Frontier Communications Northwest Inc. Intercompany Receivables/Payables and Dividends Second Quarter 2014

(Amounts in thousands, except per share amount)

A	۱s	S	ets	()=cr

Liabilities ()=dr Intercompany Receivables/Payables Balance Mar 31, 2014 Activity 2Q14

Balance Jun 30, 2014

Accounts Receivable - Affiliates
Notes Receivable - Affiliates
Interest Receivable - Affiliates
Accounts Payable - Affiliates
Notes Payable - Affiliates
Accrued Interest - Affiliates

Cash Dividends Declared

Frontier Northwest to Parent Parent to Shareholders (total) Parent to Shareholders (per share) 2Q14



FRONTIER COMMUNICATIONS CORP

FORM 10-Q (Quarterly Report)

Filed 08/07/14 for the Period Ending 06/30/14

Address HIGH RIDGE PK BLDG 3

STAMFORD, CT 06905

Telephone 2036145600

CIK 0000020520

Symbol FTR

SIC Code 4813 - Telephone Communications, Except Radiotelephone

Industry Communications Services

Sector Services

Fiscal Year 12/31



FRONTIER COMMUNICATIONS CORPORATION

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d)

OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2014

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One)
QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly peri	od ended <u>June 30, 2014</u>
	<u>or</u>
TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE	ACT OF 1934
For the transition period fi	omto
Commission file	number: <u>001-11001</u>
	CATIONS CORPORATION tt as specified in its charter)
Delaware	06-0619596
(State or other jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
3 High Ridge Park Stamford, Connecticut	06905
(Address of principal executive offices)	(Zip Code)
(Registrant's telephone n	203) 614-5600 umber, including area code) N/A ner fiscal year, if changed since last report)
Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 the registrant was required to file such reports), and (2) has been subject to such filing requirements for the pas	(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that
Yes <u>X</u>	. No
Indicate by check mark whether the registrant has submitted electronically and posted on its corporate We Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to s	
Yes X	_ No
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerat "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):	ed filer or a smaller reporting company. See definition of "accelerated filer," "large accelerated filer" and
Large accelerated filer	Non-accelerated filer Smaller reporting company
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange A	ct).
Yes	No X_
The number of shares outstanding of the registrant's Common Stock as of July 25, 2014 was 1,002,103,000.	

FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES

Index

Part I. Financial Information (Unaudited)	Page No.
Item 1. Financial Statements	
Consolidated Balance Sheets as of June 30, 2014 and December 31, 201 3	2
Consolidated Statements of Op erations for the three and six months ended June 30, 2014 and 2013	3
Consolidated Statements of Comprehensive Income (Loss) for the three and six months ended June 30, 2014 and 2013	3
Consolidated Statements of Equity for the six months ended June 30 , 2013 , the six months ended December 31 , 2013 and the six months ended June 30 , 2014	4
Consolidated Statements of Cash Flows for the six months ended June 30, 2014 and 2013	5
Notes to Consolidated Financial Statements	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	21
Item 3. Quantitative and Qualitative Disclosures about Market Risk	40
Item 4. Controls and Procedures	41
Part II. Other Information	
Item 1. Legal Proceedings	42
Item 1A. Risk Factors	42
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	42
Item 4. Mine Safety Disclosure	42
Item 6. Exhibits	43
Signature	44

1

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (\$ in thousands)

	 (Unaudited) June 30, 2014	D	December 31, 2013		
<u>ASSETS</u>					
Current assets:					
Cash and cash equivalents	\$ 801,697	\$	880,039		
Accounts receivable, less allowances of \$64,401 and \$71,362, respectively	463,932		479,210		
Restricted cash	3,200		11,411		
Prepaid expenses	64,596		68,573		
Income taxes and other current assets	 77,830		179,606		
Total current assets	1,411,255		1,618,839		
Restricted cash	2,000		2,000		
Property, plant and equipment, net	7,162,649		7,255,762		
Goodwill	6,337,719		6,337,719		
Other intangibles, net	1,063,150		1,214,932		
Other assets	 202,505		206,232		
Total assets	\$ 16,179,278	\$	16,635,484		
LIABILITIES AND EQUITY					
Current liabilities:					
Long-term debt due within one year	\$ 262,527	\$	257,916		
Accounts payable	275,119		327,256		
Advanced billings	133,560		137,319		
Accrued taxes	107,379		66,276		
Accrued interest	185,130		188,639		
Other current liabilities	 318,539		324,181		
Total current liabilities	1,282,254		1,301,587		
Deferred income taxes	2,355,494		2,417,108		
Pension and other postretirement benefits	697,708		725,333		
Other liabilities	249,839		262,308		
Long-term debt	7,650,833		7,873,667		
Equity:					
Common stock, \$0.25 par value (1,750,000,000 authorized shares,					
1,002,131,000 and 999,462,000 outstanding, respectively, and					
1,027,986,000 issued, at June 30, 2014 and December 31, 2013)	256,997		256,997		
Additional paid-in capital	4,086,548		4,321,056		
Retained earnings	153,062		76,108		
Accumulated other comprehensive loss, net of tax	(254,601)		(260,530)		
Treasury stock	(298,856)		(338,150)		
Total equity	 3,943,150		4,055,481		
Total liabilities and equity	\$ 16,179,278	\$	16,635,484		
	 ·				

The accompanying Notes are an integral part of these Consolidated Financial Statements.

PART I. FINANCIAL INFORMATION (Continued)

FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS FOR THE THREE AND SIX MONTHS ENDED JUNE 30, 2014 AND 2013 (\$ in thousands, except for per-share amounts) (Unaudited)

	For the three months ended June 30,					For the six months ended June 30,			
		2014		2013		2014		2013	
Revenue	\$	1,147,265	\$	1,190,533	\$	2,301,311	\$	2,395,929	
Operating expenses: Network access expenses		106,224		107,114		213,316		216,512	
Other operating expenses		523,385		534,015		1,052,311		1,075,514	
Depreciation and amortization		273,463		297,849		554,870		601,524	
Acquisition and integration costs		19,851				30,447			
Total operating expenses		922,923		938,978		1,850,944		1,893,550	
Gain on sale of Mohave partnership interest				14,601				14,601	
Operating income		224,342		266,156		450,367		516,980	
Investment and other income (expense), net Losses on early extinguishment of debt Interest expense		(17) - 167,611		2,956 159,780 166,547		1,378 - 338,568		7,610 159,780 337,967	
Income (loss) before income taxes Income tax expense (benefit)		56,714 19,034		(57,215) (18,755)		113,177 36,223		26,843 14,520	
Net income (loss) Less: Income attributable to the noncontrolling interest in a partnership		37,680		(38,460)		76,954		12,323 2,643	
Net income (loss) attributable to common shareholders of Frontier	\$	37,680	\$	(38,460)	\$	76,954	\$	9,680	
Basic and diluted net income (loss) per common share attributable to common shareholders of Frontier	\$	0.04	\$	(0.04)	\$	0.08	\$	0.01	

CONSOLIDATED STATEMENT S OF COMPREHENSIVE INCOME (LOSS) FOR THE THREE AND SIX MONTHS ENDED JUNE 30 , 2014 AND 2013 (\$ in thousands) (Unaudited)

	For the three months ended June 30,					For the six months ended June 30,			
	20	014		2013		2014		2013	
Net income (loss) Other comprehensive income, net of tax (see Note 13) Comprehensive income (loss)	\$	37,680 2,890 40,570	\$	(38,460) 6,746 (31,714)	\$	76,954 5,929 82,883	\$	12,323 13,492 25,815	
Less: Income attributable to the noncontrolling interest in a partnership		<u>-</u>		<u>-</u> .		<u>-</u> ,		2,643	
Comprehensive income (loss) attributable to the common shareholders of Frontier	\$	40,570	\$	(31,714)	\$	82,883	\$	23,172	

The accompanying Notes are an integral part of these Consolidated Financial Statements.

PART I. FINANCIAL INFORMATION (Continued)

FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY FOR THE SIX MONTHS ENDED JUNE 30, 2013, THE SIX MONTHS ENDED DECEMBER 31, 2013 AND THE SIX MONTHS ENDED JUNE 30, 2014 (\$ and shares in thousands) (Unaudited)

				Equity of Frontier					
	Common	Stock	Additional Paid-In	Retained	Accumulated Other Comprehensive	Treasur	y Stock	Noncontrolling	Total
	Shares	Amount	Capital	Earnings	Loss	Shares	Amount	Interest	Equity
Balance January 1, 2013	1,027,986	\$ 256,997	\$ 4,639,563	\$ 63,205	\$ (483,576)	(29,576)	\$ (368,593)	\$ 11,675	\$ 4,119,271
Stock plans	-,,		(31,045)	,	. (,)	2,113	35,148	,	4.103
Dividends on common stock			(199,866)	_	_	2,115	55,140		(199,866)
Net income	_	_	-	9,680	_	_	-	2,643	12,323
Other comprehensive income, net									
of tax	_	_	_	_	13,492	_	-		13,492
Distributions	_	_	_	_		_	-	(6,400)	(6,400)
Sale of Mohave partnership interest	-	-	-	-	-	-	-	(7,918)	(7,918)
Balance June 30, 2013	1,027,986	256,997	4,408,652	72,885	(470,084)	(27,463)	(333,445)		3,935,005
Stock plans	-	-	12,374	-	-	(1,061)	(4,705)		7,669
Dividends on common stock	-	-	(99,970)	(99,932)	-	-	-	-	(199,902)
Net income	-	-	-	103,155	-	-	-		103,155
Pension settlement costs, net of tax	-	-		-	27,381	-	-		27,381
Other comprehensive income, net									
of tax		-			182,173				182,173
Balance December 31, 2013	1,027,986	256,997	4,321,056	76,108	(260,530)	(28,524)	(338,150)	-	4,055,481
Stock plans	-	-	(34,071)	-	-	2,669	39,294	-	5,223
Dividends on common stock	-	-	(200,437)	-	-	-	-	-	(200,437)
Net income	-	-	-	76,954	-	-	-	-	76,954
Other comprehensive income, net									
of tax					5,929				5,929
Balance June 30, 2014	1,027,986	\$ 256,997	\$ 4,086,548	\$ 153,062	\$ (254,601)	(25,855)	\$ (298,856)	<u>s</u> -	\$ 3,943,150

The accompanying Notes are an integral part of these Consolidated Financial Statements.

PART I. FINANCIAL INFORMATION (Continued)

FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS FOR THE SIX MONTHS ENDED JUNE 30 , 2014 AND 2013 (\$ in thousands) (Unaudited)

		2014		2013
Cash flows provided by (used in) operating activities:				
Cash flows provided by (used in) operating activities. Net income	\$	76,954	s	12,323
Adjustments to reconcile net income to net cash provided by	Ψ	70,751	Ψ	12,323
operating activities:				
Depreciation and amortization		554,870		601,524
Losses on early extinguishment of debt		· -		159,780
Pension/OPEB costs		(417)		8,608
Stock based compensation expense		11,892		8,927
Gain on sale of Mohave partnership interest		-		(14,601)
Other non-cash adjustments		19,567		5,568
Deferred income taxes		(66,493)		(19,148)
Change in accounts receivable		15,278		43,202
Change in accounts payable and other liabilities		(34,881)		(75,159)
Change in prepaid expenses, income taxes and other current assets		64,349		(50,317)
Net cash provided by operating activities		641,119		680,707
Cash flows provided from (used by) investing activities:				
Capital expenditures - Business operations		(260,595)		(326,522)
Capital expenditures - Integration activities		(41,575)		-
Network expansion funded by Connect America Fund		(24,568)		(9,233)
Grant funds received for network expansion from Connect America Fund		3,748		5,998
Proceeds on sale of Mohave partnership interest		-		17,755
Cash transferred from escrow		8,211		21,790
Other assets purchased and distributions received, net		21,986		1,721
Net cash used by investing activities		(292,793)		(288,491)
Cash flows provided from (used by) financing activities:				
Long-term debt borrowings		10,801		750,000
Financing costs paid		(6,140)		(19,360)
Long-term debt payments		(229,626)		(1,534,074)
Premium paid to retire debt		-		(159,429)
Dividends paid		(200,437)		(199,866)
Other financing activities		(1,266)		(7,389)
Net cash used by financing activities		(426,668)		(1,170,118)
Decrease in cash and cash equivalents		(78,342)		(777,902)
Cash and cash equivalents at January 1,		880,039		1,326,532
Cash and cash equivalents at June 30,	\$	801,697	\$	548,630
Supplemental cash flow information:				
Cash paid during the period for:				
Interest	\$	319,326	\$	348,459
Income taxes, net	\$	14,408	\$	83,462

The accompanying Notes are an integral part of these Consolidated Financial Statements

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(1) Summary of Significant Accounting Policies:

(a) Basis of Presentation and Use of Estimates:
Frontier Communications Corporation and its subsidiaries are referred to as "we," "us," "our," "Frontier," or the "Company" in this report. Our interim unaudited consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America (U.S. GAAP) and should be read in conjunction with the consolidated financial statements and notes included in our Annual Report on Form 10-K for the year ended December 31, 2013. All significant intercompany balances and transactions have been eliminated in consolidation. These interim unaudited consolidated financial statements include all adjustments (consisting of normal recurring accruals) considered necessary, in the opinion of Frontier's management, to present fairly the results for the interim periods shown. Revenues, net income and cash flows for any interim periods are not necessarily indicative of results that may be expected for the full year. For our interim financial statements as of and for the period ended June 30, 2014, we evaluated subsequent events and transactions for potential recognition or disclosure through the date that we filed this quarterly report on Form 10-Q with the Securities and Exchange Commission (SEC).

Frontier had a 33 1/3 % controlling general partner interest in a partnership entity, the Mohave Cellular Limited Partnership (Mohave). Mohave's results of operations were included in our consolidated Frontier had a 33 % Contoming general partner interests in a partnership entity, the Monave Centual Entitle Partnership (Monave), Monave Stessins of operations were included in our consolidated from Consolidated statements through its date of disposal on April 1, 2013. The minority interest of the li mited partners was reflected in the consolidated statements of operations as "Income attributable to the noncontrolli ng interest in a partnership." On April 1, 2013, the Company sold its partnership interest in Mohave and received proceeds of \$17.8 million. The Company recognized a gain on sale of approximately \$14.6 million before taxes in the second quarter of 2013.

The preparation of our interim financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect (i) the reported amounts of assets and liabilities at the date of the financial statements, (ii) the disclosure of contingent assets and liabilities, and (iii) the reported amounts of revenue and expense s during the reporting period. Actual results may differ from those estimates. Estimates and judgments are used when accounting for revenue recognition, including the allowance for doubtful accounts, impairment of long-lived assets, intangible assets, depreciation and amortization, income taxes, contingencies, and pension and other postretirement benefits, among others. Certain information and footnote disclosures have been excluded and/or condensed pursuant to SEC rules and regulations.

(b) Revenue Recognition:
Revenue is recognized when services are provided or when products are delivere d to customers. Revenue that is billed in advance includes: monthly recurring network access services (including data services), special access services and monthly recurring voi ce, video and related charges. The unearned portion of these fees is initially deferred as a component of "Advanced billings" on our consolidated balance sheet and recognized as revenue over the period that the services are provided. Revenue that is billed in arrears includes: non-recurring network access services (including data services), switched access services, non-recurring voice and video services. The unearned portion of these fees is initially deferred as a component of "Advanced billings" on our consolidated balance sheet and recognized as revenue over the period that the services, switched access services, non-recurring voice and video services. The earned but unbilled portion of these fees is recognized as revenue in our consolidated statements of operations and accrued in accounts receivable in the period that the services are provided. Excise taxes are recognized as a liability when billed. Installation fees and their related direct and incremental costs are initially deferred and recognized as revenue and expense over the average te rm of a customer relationship. We recognize as current period expense the portion of installation costs that exceeds installation fee revenue.

As required by law, the Company collects various taxes from its customers and subsequently remits these taxes to governmental authorities. Substantially all of these taxes are recorded through the consolidated balance sheet and presented on a net basis in our consolidated statements of operations. We also collect Universal Service Fund (USF) surcharges from customers (primarily federal USF) that we have recorded on a gross basis in our consolidated statements of operations and included within "Revenue" and "Other operating expenses" of \$30.7 million and \$28.1 million, and \$60.8 million and \$58.0 million, for the three and six months ended June 30, 2014 and 2013, respectively.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(c) Goodwill and Other Intangibles:

Goodwill represent sthe excess of purchase price over the fair value of identifiable tangible and intangible net assets acquired. We undertake studies to determine the fair values of assets and liabilities acquired and allocate purchase prices to assets and liabilities, including property, plant and equipment, goodwill and o ther identifiable intangibles. We annually (during the fourth quartner) or more frequently, if appropriate, examine the carrying value of our goodwill and trade name to determine whether there are any impairment 1 osses. We test for goodwill impairment at the "operating segment" level, as that term is defined in U.S. GAAP. Our four operating segments consist of the following regions: Cen tral, East, National and West. Our regional operating segments are aggregated into one

The Company amortizes finite-lived intangible assets over their estimated useful lives and reviews such intangible assets at least annually (during the fourth quarter) to assess whether any potential impairment exists and whether factors exist that would necessitate a change in useful life and a different amortization period.

(2) Recent Accounting Literature

In May 2014, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) No. 2014-09, "Revenue from Contracts with Customers." This standard requires companies to recognize revenue to depict the transfer of promised goods or services to customers in a amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. This new standard is effective for annual and interim reporting periods beginning after December 15, 2016. Early adoption is not permitted. Companies are permitted to either apply the requirements retrospectively to all prior periods presented, or apply the requirements in the year of adoption, through a cumulative adjustment. The Company is currently evaluating the impact of adopting the new standard on January 1, 2017, but has not yet selected a transition method or determined the impact of adoption on its consolidated financial statements.

<u>Discontinued Operations</u>
In April 2014, the FASB issu ed ASU No. 2014-08, "Reporting Discontinued Operations and Disclosures of Dispo sals of Components of an Entity" which raises the threshold for a disposal to qualify as discontinued operations and requires new disclosures for individually material disposal transactions that do not meet the definition of a discontinue d operation. Under the new standard, companies report discontinued operations when they have a disposal that represents a strategic shift that has or will have a major impact on operations or financial results. This new standard will be a pplied prospectively and will be effective for the Company on January 1, 2015. This new standard is not expected to have a material impact on the Company's consolidated financial statements .

Internal Control – Integrated Framework
In May 2013, the Committee of Sponsoring Organizations of the Treadway Commission (COSO) issued its updated Internal Control – Integrated Framework (the 2013 Framework) and related illustrative documents. COSO will continue to make available its original Framework during the transition period extending to December 15, 2014. The Company currently utilizes COSO's original Framework, which was published in 1992 and is recognized as the leading guidance for designing, implementing and conducting internal controls over external financial reporting and assessing its effectiveness. The 2013 Framework is expected to help organizations design and implement internal control in light of many changes in business and operating environments since the issuance of the original Framework, broaden the application of internal control in addressing operations and reporting objectives, and clarify the requirements for determining what constitutes effective internal control. We plan to adopt the 2013 Framework during 2014 and do not expect that it will have a significant impact on the Company.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(3) The Transaction s

The 2010 Transaction

On July 1, 2010, the Company acquired the defined assets and liabilities of the local exchange business and related landline activities of Verizon Communications Inc. (Verizon) in 14 states (the Acquired Territories), including Internet access, long distance services and broadband video provided to designated customers in the Acquired Territories (the 20 10 Acquired Business). Frontier was considered the acquirer of the 2010 Acquired Business for accounting purposes. All integration activities for the 2010 Transaction were completed as of the end of 2012.

The AT&T Transaction

On December 17, 2013, the Company announced that it entered into an agreement to acquire the wireline properties of AT&T Inc. (AT&T) in Connecticut (the AT&T Transaction) for a purchase price of \$2.0 billion in cash, with ad justments for working capital. Upon completion of the AT&T Transaction, Frontier will operate AT&T's wireline business and fiber optic network that provides services to residential, commercial and whole sale customers in Connecticut. The Company will also acquire AT&T's U-verse® video and satellite TV customers in Connecticut.

The Company will account for the AT&T Transaction in accordance with the guidance included in U.S. GAAP for business combinations . We incurred \$ 19.9 million and \$ 30.4 million of acquisition and integrat ion related costs in connection with the AT&T Transaction during the three and six months ended June 30, 2014, respectively .

On July 25, 2014, the Federal Communications Commission (FCC) approved the AT&T Transaction. The Company is currently also seeking approval from the Connecticut Public Utilities Regulatory Authority, and has already received Justice Department approval.

(4) <u>Accounts Receivable</u>: The components of accounts receivable, net are as follows:

(\$ in thousands)	 June 30, 2014	 December 31, 2013
Retail and Wholesale	\$ 487,268	\$ 498,717
Other	41,065	51,855
Less: Allowance for doubtful accounts	(64,401)	(71,362)
Accounts receivable, net	\$ 463,932	\$ 479,210

We maintain an allowance for doubtful accounts based on our estimate of our ability to collect accounts receivable. Bad debt expense, which is recorded as a reduction to revenue, was \$11.8 million and \$ 16.1 million , and \$24.7 million and \$32.3 million , for the three and six months ended June 30 , 2014 and 2013 , respectively.

(5) <u>Property, Plant and Equipment</u>: Property, plant and equipment, net is as follows:

(\$ in thousands)	June	30, 2014	Dec	ember 31, 2013
Property, plant and equipment	\$	15,139,272	\$	14,850,697
Less: Accumulated depreciation		(7,976,623)		(7,594,935)
Property, plant and equipment, net	\$	7,162,649	\$	7,255,762

Depreciation expense is principally based on the composite group method. Depreciation expense was \$197.6 million and \$210.8 million and \$403.1 million and \$427.5 million, for the three and six months ended June 30, 2014 and 2013, respect ively. As a result of an independent study of the estimated remaining useful lives of our plant assets, we adopted

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

new estimated remaining useful lives for certain pl ant assets as of October 1, 2013, with an insignificant impact to depreciation expense.

(6) Goodwill and Other Intangibles:

The components of goodwill by the reporting units in effect as of both June 30, 2014 and December 31, 2013 are as follows:

(\$ in thousands)

Central	\$ 1,815,498
East	2,003,574
National	1,218,113
West	1,300,534
Total Goodwill	\$ 6,337,719

The components of other intangibles are as follows:

(\$ in thousands)	<u></u>			June 30, 2014				December 31, 2013							
	Gr	Gross Carrying Amount		Accumulated Amortization		Net Carrying Amount		Gross Carrying Amount		Accumulated Amortization		Net Carrying Amount			
Other Intangibles:															
Customer base	\$	2,427,648	\$	(1,488,634)	\$	939,014	\$	2,427,648	\$	(1,336,852)	\$	1,090,796			
Trade name and license		124,136		-		124,136		124,136		-		124,136			
Total other intangibles	\$	2,551,784	\$	(1,488,634)	\$	1,063,150	\$	2,551,784	\$	(1,336,852)	\$	1,214,932			

Amortization expense was \$ 75.9 million and \$ 87.0 million , and \$ 151.8 million and \$174.0 million, for the three and six months ended June 30 , 2014 and 2013 , respectively. Amortization expense represents the amortization of intangible assets (primarily customer base) that were acquired in the 2010 Transaction based on a useful life of nine years for the residential customer base and 12 years for the business customer base, amortized on an accelerated method.

(7) Fair Value of Financial Instruments:

The following table summarizes the carrying amounts and estimated fair values for long-term debt at June 30, 2014 and December 31, 2013. For the other financial instruments including cash, accounts receivable, long-term debt due within one year, accounts payable and other current liabilities, the carrying amounts approximate fair value due to the relatively short ma turities of those instruments. Other equity method investments, for which market values are not readily available, are carried at cost, which approximates fair value.

The fair value of our long-term debt is estimated based upon quoted market prices at the reporting date for those financial instruments.

(\$ in thousands)	_	June 3	014	_	December 31, 2013							
	_	Carrying Amount	_	Fair Value	_	Carrying Amount	Fair Value					
Long-term debt	\$	7,650,833	\$	8,419,064	\$	7,873,667	\$	8,191,744				

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

 $\begin{array}{ll} \hbox{(8)} & \underline{\text{Long-Term Debt}}: \\ \hbox{The activity in our long-term debt from December 31, 201 3 to June 30 , 2014 is summarized as follows:} \end{array}$

					Interest Rate at				
(\$ in thousands)	De	December 31, 2013		Payments and Retirements		New Borrowings		June 30, 2014	June 30, 2014 *
Senior Unsecured Debt	\$	8,107,066	\$	(228,750)	\$	-	\$	7,878,316	7.93%
Other Secured Debt		13,550		(669)		10,801		23,682	3.62%
Rural Utilities Service Loan Contracts		8,930		(207)		-		8,723	6.15%
Total Long-Term Debt	\$	8,129,546	\$	(229,626)	\$	10,801	\$	7,910,721	7.91%
Less: Debt (Discount)/Premium		2,037						2,639	
Less: Current Portion		(257,916)						(262,527)	
	\$	7,873,667					\$	7,650,833	

^{*} Interest rate includes amortization of debt issuance costs a nd debt premiums or discounts. The interest rates at June 30, 2014 represent a weighted average of multiple issuances.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Additional information regarding our Senior Unsecured Debt is as follows:

(\$ in thousands)	June 30,	2014	December	December 31, 2013				
	Principal Outstanding	Interest Rate	Principal Outstanding	Interest Rate				
Senior Notes and Debentures Due:								
5/1/2014	-	-	200,000	8.250%				
3/15/2015	105,026	6.625%	105,026	6.625%				
4/15/2015	96,872	7.875%	96,872	7.875%				
10/14/2016 *	431,250	3.025% (Variable)	460,000	3.045% (Variable)				
4/15/2017	606,874	8.250%	606,874	8.250%				
10/1/2018	582,739	8.125%	582,739	8.125%				
3/15/2019	434,000	7.125%	434,000	7.125%				
4/15/2020	1,021,505	8.500%	1,021,505	8.500%				
7/1/2021	500,000	9.250%	500,000	9.250%				
4/15/2022	500,000	8.750%	500,000	8.750%				
1/15/2023	850,000	7.125%	850,000	7.125%				
4/15/2024	750,000	7.625%	750,000	7.625%				
11/1/2025	138,000	7.000%	138,000	7.000%				
8/15/2026	1,739	6.800%	1,739	6.800%				
1/15/2027	345,858	7.875%	345,858	7.875%				
8/15/2031	945,325	9.000%	945,325	9.000%				
10/1/2034	628	7.680%	628	7.680%				
7/1/2035	125,000	7.450%	125,000	7.450%				
10/1/2046	193,500	7.050%	193,500	7.050%				
	7,628,316		7,857,066					
Subsidiary Senior Notes and Debentures Due:								
2/15/2028	200,000	6.730%	200,000	6.730%				
10/15/2029	50,000	8.400%	50,000	8.400%				
Total	\$ 7,878,316	7.78% **	\$ 8,107,066	7.78% **				

During the first six months of 2014, we entered into secured financings totaling \$10. 8 million with four year terms and no stated interest rate for certain equipment purchases.

On December 16, 2013, we signed a commitment letter for a bridge loan facility (the Bridge Facility) and recognized interest expense related to this commitment of \$7.5 million and \$15.0 million during the three and six months ended June 30, 2014. On January 29, 2014, we entered into a bridge loan agreement (the Bridge Loan Agreement) with the Lenders party thereto and JP Morgan Chase Bank, N.A., as administrative agent, pursuant to which the Lenders have agreed at closing of the AT&T Transaction to provide to us an unsecured bridge loan facility for up to \$1.9 billion for the purposes of funding (i) substantially all of the purchase price for the AT&T Transaction and (ii) the fees and expenses incurred in connection with the transactions contemplated by the stock purchase agreement for the AT&T Transaction. Pursuant to the Bridge Loan Agreement, if and to the extent we do not, or are unable to, issue debt securities yielding up to \$1.9 billion in gross cash proceeds on or prior to the closing of the AT&T Transaction, we shall draw down up to \$1.9 billion, less the amount of the debt securities, if any, issued by us on or prior to the closing of the AT&T Transaction, in aggregate principal amount of loans under the Bridge Facility to fund the purchase price of the AT&T Transaction.

Represents borrowings under the 2011 CoBank Credit Agreement, as defined below.
 Interest rate represents a weighted average of the stated interest rates of multiple issuances.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

On July 16, 2014, the Bridge Facility was reduced by \$350.0 million following execution of the 2014 CoBank Credit Agreement, as defined below. All other terms and conditions remain unchanged.

On June 2, 2014, the Company entered into a credit agreement with CoBank, ACB, as administrative agent, lead arranger and a lender, and the other lenders party thereto, for a \$350.0 million senior unsecured delayed draw term loan facility (the 2014 CoBank Credit Agreement). The facility will be drawn upon closing of the AT&T Transaction with proceeds used to partially finance the acquisition. The maturity date is the earlier of the fifth anniversary of the draw date and December 15, 2019. Repayment of the outstanding principal balance will be made in quarterly installments in the amount of \$8.8 million, commencing one full fiscal quarter after the draw date, with the remaining outstanding principal balance to be repaid on the maturity date. Borrowings under the 2014 CoBank Credit Agreement will bear interest based on the margins over the Base Rate (as defined in the 2014 CoBank Credit Agreement) or LIBOR, at the election of the Company. Interest rate margins under the facility (ranging from 0.875% to 2.875% for Base Rate borrowings and 1.875% to 3.875% for LIBOR borrowings) are subject to adjustments based on the Total Leverage Ratio of the Company, as such term is defined in the 2014 CoBank Credit Agreement.

On June 2, 2014, the Company entered into a new revolving credit agreement with JPMorgan Chase Bank, N.A., as administrative agent, the lenders party thereto and the other parties named therein (the New Revolving Credit Agreement), for a \$750.0 million revolving credit facility (the New Revolving Credit Facility) with a scheduled termination date of May 31, 2018 and terminated its existing revolving credit facility (the Prio r Revolving Credit Facility) under the Credit Agreement, dated as of May 3, 2013, among the Company, JPMorgan Chase Bank, N.A., as administrative agent, the lenders party thereto and the other parties named therein (the Prior Revolving Credit Agreement). As of June 30, 2014, a the New Revolving Credit Facility in the Seven Sevolving Credit Facility will vary from time to time depending on the Company's debt rating (as defined in the New Revolving Credit Agreement) and were 0.400 % per annum as of June 30, 2014. During the term of the New Revolving Credit Facility, the Company may borrow, repay and reborrow funds, and may obtain letters of credit, subject to customary borrowing conditions. Loans under the New Revolving Credit Facility will bear interest based on the alternate base rate or the adjusted LIBO R ate (each as determined in the New Revolving Credit Agreement), at the Company's election, plus a margin based on the Company's debt rating (ranging from 0.50% to 1.50% for alternate base rate borrowings and 1.50% to 2.50% for adjusted LIBO R ate borrowings). The current pricing on this facility would have been 1.0 0 % or 2.0 0 %, respectively, as of June 30, 2014. Letters of credit issued under the New Revolving Credit Facility will also be subject to fees that vary depending on the Company's debt rating. The New Revolving Credit Facility is available for general corporate purposes but may not be u sed to fund dividend payments. The terms of the New Revolving Credit Facility are substantially similar to the terms of the Prior Revolving Credit Facility are substantially similar to the te

The Company has a credit agreement with CoBank, ACB, as administrative agent, lead arranger and a lender, and the other lenders party thereto, for a \$575.0 million senior unsecured term loan facility with a final maturity of October 14, 2016 (the 2011 CoBank Credit Agreement). The entire facility was drawn upon execution of the 2011 CoBank Credit Agreement in October 2011. Repayment of the outstanding principal balance is made in quarterly installments in the amount of \$14.4 million, which commenced on March 31, 2012, with the remaining outstanding principal balance to be repaid on the final maturity date. Borrowings under the 2011 CoBank Credit Agreement bear interest based on the margins over the Base Rate (as defined in the 2011 CoBank Credit Agreement) or LIBOR, at the election of the Company. Interest rate margins under the facility (ranging from 0.875% to 2.875% for Base Rate borrowings and 1.875% to 3.875% for LIBOR borrowings) are subject to adjustments based on the Total Leverage Ratio of the Company, as such term is defined in the 2011 CoBank Credit Agreement. The current pricing on this facility is LIBOR plus 2.875%.

As of June 30, 2014, we were in compliance with all of our debt and credit facility financial covenants.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

P rincipal payments for our currently outstanding debt for the next five years are as follows as of June 30, 2014:

(\$ in thousands)	Principal Payments						
2014 (remaining six months)	\$	30,315					
2015	\$	262,541					
2016	\$	348,167					
2017	\$	610,075					
2018	\$	583,948					
2019	\$	434,565					

(9) Income Taxes:
The following is a reconciliation of the provision for income taxes computed at federal statutory rates to the effective rates:

	For the three mor		For the six months ended June 30,			
	2014	2013	2014	2013		
Consolidated tax provision at federal statutory rate	35.0 %	35.0 %	35.0 %	35.0 %		
State income tax provisions, net of federal income						
tax benefit	1.9	9.2	1.1	(8.2)		
Noncontrolling interest	=	(0.6)	-	-		
Tax reserve adjustment	(3.2)	7.7	(1.6)	(15.4)		
Changes in certain deferred tax balances	=	(10.6)	(2.5)	22.6		
IRS audit adjustments	=	(9.0)	-	19.3		
All other, net	(0.1)	1.1	<u> </u>	0.8		
Effective tax rate	33.6 %	32.8 %	32.0 %	54.1 %		

Income taxes for the three and six months ended June 30, 2014 include a \$1.9 million benefit from the net reversal of reserves for uncertain tax positions. Income taxes for the six months ended June 30, 2014 include the impact of a \$2.9 million benefit arising from state tax law changes.

Income taxes for the three and six months ended June 30, 2013 include the impact of a charge of \$5.2 million resulting from the settlement of the 2010 IRS audit and a \$6.0 million charge resulting from the adjustment of deferred tax balances, partially offset by a \$4.4 million benefit from the net reversal of reserves for uncertain tax positions.

As of June 30, 2014, there were no uncertain tax positions for which the statutes of limitations are expected to expire during the next twelve months and which would affect our effective tax rate .

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(10) Net Income (Loss) Per Common Share:
The reconciliation of the net income per common share calculation is as follows:

(<u>\$ and shares in thousands, except per share amounts</u>)	1	For the three mo June 3	For the six months ended June 30,				
		2014	2013		2014		2013
Net income (loss) used for basic and diluted earnings per common share:							
Net income (loss) attributable to common shareholders of Frontier	\$	37,680 \$	(38,460)	\$	76,954	\$	9,680
Less: Dividends paid on unvested restricted stock awards	_	(746)	(755)	_	(1,513)	_	(1,276)
Total basic and diluted net income (loss) attributable to common shareholders of Frontier	\$	36,934 \$	(39,215)	\$	75,441	\$	8,404
Basic earnings per common share: Total weighted average shares and unvested restricted stock		1 002 101	000 224		1 001 474		000 770
awards outstanding - basic Less: Weighted average unvested restricted stock awards		1,002,191 (7,563)	999,234 (6,623)		1,001,474 (7,189)		998,770 (6,606)
Total weighted average shares outstanding - basic	_	994,628	992,611	_	994,285	_	992,164
Basic net income (loss) per share attributable to common							
shareholders of Frontier	\$	0.04 \$	(0.04)	\$	0.08	\$	0.01
Diluted earnings per common share:							
Total weighted average shares outstanding - basic		994,628	992,611		994,285		992,164
Effect of dilutive shares	_	2,157	603	_	2,645		449
Total weighted average shares outstanding - diluted	_	996,785	993,214	-	996,930	_	992,613
Diluted net income (loss) per share attributable to common shareholders of Frontier	\$	0.04 \$	(0.04)	\$	0.08	\$	0.01

Stock Options

For the three and six months ended June 30, 2014, options to purchase 83,000 shares (at exercise prices ranging from \$ 12.50 to \$ 14.15) and for the three and six months ended June 30, 2013, options to purchase 93,000 shares (at exercise prices ranging from \$ 12.50 to \$ 14.15), issuable under employee compensation plans were excluded from the computation of diluted earnings per share (EPS) for those periods because the exercise prices were greater than the average market price of our common stock and, therefore, the effect would be antidilutive. In calculating diluted EPS, we apply the treasury stock method and include future unearned compensation as part of the assumed proceeds.

Stock Units
At June 30, 2014 and 2013, we had 1, 372,163 and 1,044,128 stock units, respectively, issued under our Non-Employee Directors' Deferred Fee Equity Plan (Directors' Equity Plan). These securities have not been included in the diluted income per share of common stock cal culation for the three and six months ended June 30, 2014 and 2013 because their inclusion would have an antidilutive effect.

(11) Stock Plans:
At June 30, 2014, we had six stock-based compensation plans under which grants were made and awards remained outstanding. No further awards may be granted under four of the plans: the 1996 Equity Incentive Plan (the 1996 EIP), the Amended and Restated 2000 Equity Incentive Plan (the 2000 EIP), the 2009 Equity Incentive Plan (the 2009 E IP) and the Deferred Fee Plan. At June 30, 2014, there were 22,540,761 shares authorized for grant and 14,732,186 shares available for grant under the 2013 Equity Incentive Plan (the 2013 EIP and together with the 1996 EIP, the 2000 EIP and the 2009 EIP,

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

the EIPs) and the Directors' Equity Plan. Our general policy is to issue shares from treasury upon the grant of restricted shares and the exercise of options.

Performance Shares
On February 15, 2012, the Company's Compensation Committee, in consultation with the other non-management directors of the Company's Board of Directors and the Committee's independent executive compensation consultant, adopted the Frontier Long-Term Incentive Plan (the LTIP). LTIP awards are granted in the form of performance shares. The LTIP is currently offered under the Company's 2009 EIP and 2013 EIP, and participants consist of senior vice presidents and above. The LTIP awards have performance, market and time-vesting conditions.

Beginning in 2012, during the first 90 days of a three -year performance period (a M easurement Period), a target number of performance shares are awarded to each LTIP participant with respect to the Measurement Period. The performance metrics under the LTIP are (1) annual targets for operating cash flow based on a goal set during the first 90 days of each year in the three-year Measurement Period and (2) an overall performance "modifier" set during the first 90 days of the Measurement Period, based on the Company's total return to stockholders (i.e., Total Shareholder Return or TSR) relative to the Integrated Telecommunications Services Group (GICS Code 50101020) for the three-year Measurement Period. Operating cash flow performance is determined at the end of each year and the annual results will be averaged at the end of the three-year Measurement Period to determine the preliminary number of shares earned under the LTIP award. The TSR performance measure is then applied to decrease or increase payouts based on the Company's three year relative TSR performance. LTIP awards, to the extent earned, will be paid out in the form of common stock shortly following the end of the three-year

In 2012, the Compensation Committee granted approximately 979, 000 performance shares under the LTIP and set the operating cash flow performance goal for the first year in the 2012-2014 Measurement Period and the TSR modifier for the three-year Measurement Period. In 2013, the Compensation Committee granted approximately 1,124,000 performance shares under the LTIP and set the operating cash flow performance goal for 2013, which applies to the first year of the 2013-2015 Measurement Period and the second year of the 2012-2014 Measurement Period. On February 17, 2014, the Compensation Committee granted 10/3/2019 performance phase under the LTIP and set the operation of the 2012-2014 Measurement Period. On February 17, 2014, the Compensation Committee granted 10/3/2019 performance phase under the LTIP and set the operation of the 2012-2014 Measurement Period. Committee granted 1,028,000 performance shares under the LTIP and set the operating cash flow performance goal for 2014, which applies to the first year in the 2014-2016 Measurement Period, the second year of the 2013-2015 Measurement Period and the third year of the 2012-2014 Measurement Period. The number of shares of common stock earned at the end of each three-year Measurement Period may be more or less than the number of target performance shares granted as a result of operating c ash flow and TSR performance. An executive must maintain a satisfactory performance rating during the Measurement Period and must be employed by the Company at the end of the three-year Measurement Period in order for the award to vest. The Compensation Committee will determine the number of shares earned for each three year Measurement Period in February of the year following the end of the Measurement Period.

The following summary presents information regarding LTIP target performance shares as of June 30, 2014 and changes during the six months then ended with regard to LTIP shares awarded under the 2009 EIP and the 2013 EIP

	Number of
	Shares
Balance at January 1, 2014	1,749,000
LTIP target performance shares granted	1,028,000
LTIP target performance shares forfeited	(80,000)
Balance at June 30, 2014	2,697,000

For the six months ended June 30, 2014 and 2013, the Company recognized an expense of \$ 2.0 million and \$ 0.7 million, respectively, for the LTIP.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Restricted Stock
The following summary presents information regarding unvested restricted stock as of June 30, 2014 and changes during the six months then ended with regard to restricted stock under the 2009 EIP and the 2013 EIP:

	Weighted										
		Average									
	Number of	Grant Date		Aggregate							
	Shares	Fair Value	Fair Value								
Balance at January 1, 2014	6,234,000 \$	4.80	\$	28,988,000							
Restricted stock granted	3,766,000 \$	4.71	\$	21,995,000							
Restricted stock vested	(2,282,000) \$	5.23	\$	13,329,000							
Restricted stock forfeited	(242,000) \$	4.50									
Balance at June 30, 2014	7,476,000 \$	4.64	\$	43,661,000							

For purposes of determining compensation expense, the fair value of each restricted stock grant is estimated based on the average of the high and low market price of a share of our common stock on the date of grant. Total remaining unrecognized compensation cost associated with unvested restricted stock awards at June 30 , 2014 was \$ 28.9 million and the weighted average period over which this cost is expected to be recognized is approximately two years.

Shares granted during the first six months of 2013 totaled 3,340,000. The total fair value of shares granted and vested at June 30, 2013 was approximately \$13.5 million and \$7.6 million, respectively. The total fair value of unvested restricted stock at June 30, 2013 was \$ 31.2 million . The weighted average grant date fair value of restricted shares granted during the six month s ended June 30, 2013 was \$

We have granted restricted stock awards to employees in the form of our common stock. None of the restricted stock awards may be sold, assigned, pledged or otherwise transferred, voluntarily or involuntarily, by the employees until the restrictions lapse, subject to limited exceptions. The restrictions are time-based. Compensation expense, recognized in "Other operating expenses", of \$ 9.6 million and \$ 8.4 million for the six months ended June 30, 2014 and 2013, respectively, has been recorded in connection with these grants.

Stock Options
There has been no activity with regard to stock options during the first six months of 2014. There was no intrinsic value for the stock options outstanding and exercisable at June 30, 2014 and 2013.

(12) Segment Information:
We operate in one reportable segment. Frontier provides both regulated and unregulated voice, data and video services to residential, business and wholesale customers and is typically the incumbent voice services provider in its service areas.

As permitted by U.S. GAAP, we have utilized the aggregation criteria to combine our operating segments because all of our properties share similar economic characteristics, in that they provide the same products and services to similar customers using comparable technologies in all of t he states in which we operate. The regulatory s tructure is generally similar. Differences in the regulatory regime of a particular state do not significantly impact the economic characteristics or operating results of a particular property.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(13) Comprehensive Income:
Comprehensive income consists of net income and other gains and losses affecting shareholders' investment and pension/postretirement benefit (OPEB) liabilities that, under U.S. GAAP, are excluded from net income.

The components of accumulated other comprehensive loss, net of tax at June 30, 2014 and 2013, and changes for the six months then ended, are as follows:

(\$ in thousands)	Pension Costs			OPEB Costs	Deferred taxes on pension and OPEB costs		All other	Total	
Balance at January 1, 2014	\$	(411,432)	\$	(5,230)	\$ 156,280	\$	(148)	\$	(260,530)
Other comprehensive income before reclassifications		-		-	-		148		148
Amounts reclassified from accumulated other comprehensive loss	_	9,660	_	(336)	(3,543)	_	-		5,781
Net current-period other comprehensive income (loss) Balance at June 30, 2014	\$	9,660 (401,772)	\$	(336) (5,566)	\$ (3,543) \$ 152,737	\$	148	\$	5,929 (254,601)
(§ in thousands) Balance at January 1, 2013	\$	Pension Costs (697,874)	\$	OPEB Costs (74,264)	Deferred taxes on pension and OPEB costs \$ 288,712	\$	All other (150)	\$	Total (483,576)
Other comprehensive income before reclassifications		(230)		-	356		2		128
Amounts reclassified from accumulated other comprehensive loss	_	20,148		1,406	(8,190)	_	<u> </u>		13,364
Net current-period other comprehensive income (loss) Balance at June 30, 2013	\$	19,918 (677,956)	\$	1,406 (72,858)	\$ 280,878	\$	(148)	\$	13,492 (470,084)

The significant items reclassified from each component of accumulated other comprehensive loss for the three and six months ended June 30, 2014 and 2013 are as follows:

(\$ in thousands)											
Details about Accumulated Other Comprehensive		For the three months ended June 30, For the six months ended June 30,							Affected Line Item in the Statement Where		
Loss Components		2014		2013		2014		2013	Net Income is Presented		
Amortization of Pension Cost Items 60											
Prior-service costs	\$	(11)	S	(2)	\$	(22)	\$	(4)			
Actuarial gains/(losses)		(4,819)		(10,072)		(9,638)		(20,144)			
		(4,830)		(10,074)		(9,660)		(20,148)	Income (loss) before income taxes		
Tax impact		1,835		3,828		3,671		7,656	Income tax (expense) benefit		
	\$	(2,995)	S	(6,246)	\$	(5,989)	\$	(12,492)	Net income (loss)		
Amortization of OPEB Cost Items											
Prior-service costs	\$	890	S	1,525	\$	1,780	\$	3,050			
Actuarial gains/(losses)		(722)		(2,228)		(1,444)		(4,456)			
		168		(703)		336		(1,406)	Income (loss) before income taxes		
Tax impact	_	(64)		267	_	(128)	_	534	Income tax (expense) benefit		
	\$	104	S	(436)	\$	208	\$	(872)	Net income (loss)		

⁽a) Amounts in parentheses indicate losses.
(b) These accumulated other comprehensive loss components are included in the computation of net periodic pension and OPEB cost s (see Note 1 4 - Retirement Plans for additional details).

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

(14) Retirement Plans:

The following tables provide the components of net periodic benefit cost:

	Pension Benefits									
	For the three months ended June 30.					For the six months ended June 30.				
	2014			2013		2014		2013		
(<u>\$ in thousands</u>)		2014		2013	_	2014		2013		
Components of net periodic pension benefit cost										
Service cost	\$	9,940	\$	12,834	\$	19,880	\$	25,668		
Interest cost on projected benefit obligation		19,824		18,880		39,648		37,760		
Expected return on plan assets		(23,512)		(24,590)		(47,024)		(49,180)		
Amortization of prior service cost /(credit)		11		2		22		4		
Amortization of unrecognized loss		4,819		10,072		9,638		20,144		
Net periodic pension benefit cost	\$	11,082	\$	17,198	\$	22,164	\$	34,396		

				Postretireme	nt E	Benefits				
	Other Than Pensions (OPEB)									
	For the three months ended June 30,					For the six months ended June 30,				
		2014		2013		2014		2013		
(\$ in thousands)										
Components of net periodic postretirement benefit cost										
Service cost	\$	2,553	\$	3,179	\$	5,106	\$	6,358		
Interest cost on projected benefit obligation		4,788		4,440		9,576		8,880		
Expected return on plan assets		(7)		(43)		(14)		(86)		
Amortization of prior service cost/(credit)		(890)		(1,525)		(1,780)		(3,050)		
Amortization of unrecognized loss		722		2,228		1,444		4,456		
Net periodic postretirement benefit cost	\$	7,166	\$	8,279	\$	14,332	\$	16,558		

During the first six months of 201 4 and 201 3, we capitalized \$ 7.9 million and \$10.0 million, respectively, of pension and OPEB expense into the cost of our capital expenditures, as the costs relate to our engineering and plant construction activities. Based on current assumptions and plan asset values, we estimate that our 201 4 pension and OPEB expenses will be approximately \$65 million to \$ 85 million for our current business operations, excluding amounts capitalized into the cost of capital expenditures, as compared to \$ 97.1 million in 2013, excluding the impact of pension settlement costs and amounts capitalized into the cost of capital expenditures. We made total cash contributions to our pension plan during the six months ended June 30, 2014 of \$ 31.2 million . An additional cash contribution of \$19.6 million was made on July 15, 2014. We expect that we will make total contributions of cash and/or other assets to our pension plan of approximately \$100 million in 2014.

The Compa ny's pension plan assets have in creased from \$1,216.5 million at December 31,2013 to \$1,276.5 million at June 30,2014, an in crease of \$60.0 million, or 5%. This in crease is a result of positive investment returns of \$95.5 million and cash contributions of \$31.2 million, offset by benefit payments of \$66.7 million during the first six months of 2014.

(1 5) Commitments and Contingencies:

We anticipate total capital expenditures for our current business operations of approximately \$575 million to \$625 million for 2014, excluding the expenditure of funds previously received from the Connect America Fund program. Although we from time to time make short-term purchasing commitments to vendors with respect to these expenditures, we generally do not enter into firm, written contracts for such activities.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

In connection with the pending AT&T Transaction, the Company currently expects to incur operating expenses and capital expenditures of approximately \$225 million to \$275 million in 2014 related to integration in itatives. The Company incurred \$30.4 million of operating expenses, consisting of \$1.6 million and \$28.8 million of acquisition and integration costs, respectively, and \$41.6 million in capital expenditures related to the AT&T Transaction during the six months ended June 30, 2014.

In connection with the 2010 Transaction, the FCC and certain state regulatory commissions, in connection with granting their approvals of the 2010 Transaction, specified certain capital expenditure and operating requirements for the Acquired Territories for specified periods of time post-closing. These requirements focus primarily on certain capital investment commitments to expand broadband availability to at least 85% of the households throughout the Acquired Territories with minimum download speeds of 3 megabits per second (Mbps) by the end of 2013. We are required to provide download speeds of 4 Mbps to at least 75%, 80% and 85% of the households throughout the Acquired Territories by the end of 2013, 2014 and 2015, respectively. As of December 31, 2013, we met our FCC requirement to provide 3 Mbps coverage to 85% of the households in the Acquired Territories by the end of 2013. As of December 31, 2012, we also met our FCC requirement to provide 4 Mbps coverage to 75% and 80% of the households in the Acquired Territories. As of June 30, 2014, we expanded broadband availability in excess of 4 Mbps to 83.9 % of the households throughout the Acquired Territories.

To satisfy all or part of certain capital investment commitments to three state regulatory commissions in connection with the 2010 Transaction , we placed a total of \$115.0 million in cash into escrow accounts and obtained a letter of credit for \$190 million in 2010. In September 2013, the letter of credit expired. Another \$72.4 milli on of cash in an escrow account was acquired in connection with the 2010 Transaction to be used for service quality initiatives in the state of West Virginia. As of June 30 , 2014 , \$184.5 million had been released from the escrow accounts and the Company had a restricted cash balance in the remaining escrow account in the aggregate amount of \$3.2 milli on , including interest earned. The aggregate amount of this escrow account will continue to decrease over time as we make the required capital expenditures in West Virginia.

In our normal course of business, we have obligations under certain non-cancela ble arrangements for services. During 2012, we entered into a "take or pay" arrangement for the purchase of future long distance and carrier services. Our remaining commitments under the arrangement are \$145.5 million and \$140.8 million for the years ending December 31, 2014 and 2015, respectively. As of June 30, 2014, we expect to utilize the services included within the arrangement and no liability for the "take or pay" provision has been recorded.

We are party to various legal proceedings (including individual, class and putative class actions) arising in the normal course of our business covering a wide range of matters and types of claims including, but not limited to, general contracts, billing disputes, rights of access, taxes and surcharges, consumer protection, trademark and patent infringement, employment, regulatory, tort, claims of competitors and disputes with other carriers.

We accrue an expense for pending litigation when we determine that an unfavorable outcome is probable and the amount of the los s can be reasonably estimated. Legal defense costs are expensed as incurred. None of our existing accruals, after considering insurance coverage, for pending matters is material. We monitor our pending litigation for the purpose of adjusting our accruals and revising our disclosures accordingly, when required. Litigation is, however, subject to uncertainty, and the outcome of any particular matter is not predictable. We will vigorously defend our interests for pending litigation, and as of this date, we believe that the ultimate resolution of all such matters, after considering insurance coverage or other indemnities to which we are entitled, will not have a material adverse effect on our consolidated financial position, results of operations, or our cash flows.

We sold all of our utility b usinesses as of April 1, 2004. However, we have retained a potential payment obligation associated with our previous electric utility activ ities in the State of Vermont. The Vermont Joint Owners (VIO), a consortium of 14 Vermont utilities, including us, entered into a purchase power agree ment with Hydro-Quebec in 1987. The agreement contains "step-up" provisions that state if any VIO member defaults on its purchase obligation under the contract to purchase power from Hydro-Quebec, then the other VIO participants will assume responsibility for the defaulting part y's share on a prorata basis. Our pro-rata share of the purchase power obligation is 10%. If any member of the VIO defaults on its obligations under the Hydro-Quebec agreement, then the remaining members of the VIO.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

including us, may be required to pay for a substantially larger share of the VJO's total purchase power obligation for the remainder of the agreement (which runs through 2015). U.S. GAAP rules require that we disclose "the maximum potential amount of future payments (undiscounted) the guarantor could be required to make under the guarantee." U.S. GAAP rules also state that we must make such disclosure "... even if the likelihood of the guarantor's having to make any payments u nder the guarantee is remote..." As noted above, our obligation only arises as a result of default by another VIO me mber, such as upon bankruptcy. Therefore, to satisfy the "maximum potential amount" disclosure requirement we must assume that all members of the VJO simultaneously default, an unlikely scenario given that all VJO members are regulated utility providers with regulated cost recovery. Despite the remote chance that such an event could occur, or that the State of Vermont could or would allow such an event, assuming that all the members of the VJO defaulted on January 1, 2014 and remained in default for the duration of the contract (another 2 years), we estimate that our undiscounted purchase obligation for 2014 through 2015 would be approximately \$2.874, million. In such a scenario, the Company would then own the power and could seek to recover its costs. We would do this by seeking to recover our costs from the defaulting members and/or reselling the power to other utility providers or the northeast power grid. There is an active market for the sale of power. We could potentially lose money if we were una ble to sell the power at cost. We caution that we cannot predict with any degree of certainty any potential outcome.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Forward-Looking Statements

This quarterly report on Form 10-Q contains forward-looking statements that are subject to risks and uncertainties that could cause actual results to differ material ly from those expressed or implied in the statements. Statements that are not historical facts are forward-looking statements made pursuant to the safe harbor provisions of The Private Securities Litigation Reform Act of 1995. Words such as "believe," "anticipate," "expect" and similar expressions are intended to identify forward-looking statements. Forward-looking statements (including oral representations) are only predictions or statements of current plans, which we review continuously. Forward-looking statements may differ from actual future results due to, but not limited to, and our future results may be material ly affected by, the following potential risks or uncertainties. You should understand that it is not possible to predict or identify all potential risks or uncertainties. We note the following as a partial list:

- · Our ability to complete the acquisition of the Connecticut operations from AT&T on the terms or timeline currently contemplated, or at all;
- The ability to successfully integrate the Connecticut operations into our existing operations and the diversion of management's attention from ongoing business and regular business responsibilities to effect such integration;
- The effects of increased expenses incurred due to activities related to the AT&T Transaction;
- The risk that the cost savings from the AT&T Transaction may not be fully realized or may take longer to realize than expected or that our actual integration costs may exceed our estimates;
- · The sufficiency of the assets to be acquired from AT&T to enable the combined company to operate the acquired business;
- · Failure to enter into or obtain, or delays in entering into or obtaining, certain agreements and consents necessary to operate the acquired business as planned;
- · The failure to obtain, delays in obtaining or adverse conditions contained in any required regulatory approvals for the AT&T Transaction;
- · Disruption from the AT&T Transaction making it more difficult to maintain relationships with customers or suppliers of the Connecticut operations;
- The effects of greater than anticipated competition from cable, wireless and other wireline carriers that could require us to implement new pricing, marketing strategies or new product or service offerings and the risk that we will not respond on a timely or profitable basis;
- · Reductions in the number of our voice customers that we cannot offset with increases in broadband subscribers and sales of other products and services;
- · Our ability to maintain relationships with customers, employees or suppliers;
- The effects of ongoing changes in the regulation of the communications industry as a result of federal and state legislation and regulation, or changes in the enforcement or interpretation of such legislation and regulation;
- · The effects of any unfavorable outcome with respect to any current or future legal, governmental or regulatory proceedings, audits or disputes;
- · The effects of changes in the availability of federal and state universal service funding or other subsidies to us and our competitors;
- · Our ability to successfully adjust to changes in the communications industry and to implement strategies for growth;

- Continued reductions in switched access revenues as a result of regulation, competition or technology substitutions;
- · Our ability to effectively manage service quality in our territories and meet mandated service quality metrics;
- · Our ability to successfully introduce new product offerings, including our ability to offer bundled service packages on terms that are both profitable to us and attractive to customers;
- The effects of changes in accounting policies or practices adopted voluntarily or as required by generally accepted accounting principles or regulations;
- · Our ability to effectively manage our operations, operating expenses and capital expenditures, and to repay, reduce or refinance our debt;
- The effects of changes in both general and local economic conditions on the markets that we serve, which can affect demand for our products and services, customer purchasing decisions, collectability of revenues and required levels of capital expenditures related to new construction of residences and businesses;
- The effects of technological changes and competition on our capital expenditures, products and service offerings, including the lack of assurance that our network improvements in speed and capacity will be sufficient to meet or exceed the capabilities and quality of competing networks;
- The effects of increased medical expenses (including as a result of the impact of the Patient Protection and Affordable Care Act) and pension and postemployment expenses, such as retiree medical and severance costs, and related funding requirements;
- The effects of changes in income tax rates, tax laws, regulations or rulings, or federal or state tax assessments;
- · Our ability to successfully renegotiate union contracts;
- · Changes in pension plan assumptions and/or the value of our pension plan assets, which could require us to make increased contributions to the pension plan in 2014 and beyond;
- · The effects of economic downturns which could result in difficulty in collection of revenues and loss of customers;
- Adverse changes in the credit markets or in the ratings given to our debt securities by nationally accredited ratings organizations, which could limit or restrict the availability, or increase the cost, of financing to us;
- · Our cash flow from operations, amount of capital expenditures, debt service requirements, cash paid for income taxes and liquidity may affect our payment of dividends on our common shares;
- . The effects of state regulatory cash management practices that could limit our ability to transfer cash among our subsidiaries or dividend funds up to the parent company; and
- The effects of severe weather events such as hurricanes, tornadoes, ice storms or other natural or man-made disasters, which may increase our operating expenses or adversely impact customer revenue.

Any of the foregoing events, or other events, could cause financial information to vary from management's forward-looking statement is included in this report. You should c onsider these important factors in evaluating any statement in this report on Form 10-Q or otherwise made by us or on our behalf. The following information is unaudited and should be read in conjunction with the consolidated financial statements and related notes included in this report. We have no obligation to update or revise these forward-looking statements and do not undertake to do so.

Investors should also be aware that while we do, at various times, communicate with securities analysts, it is against our policy to disclose to them selectively any material non-public information or other confidential information. Accordingly, investors should not assume that we agree with any statement or report issued by an analyst irrespective of the content of the

statement or report. To the extent that reports issued by securities analysts contain any projections, forecasts or opinions, such reports are not our responsibility.

Overview

tote 3 of the Notes to Consolidated Financial Statements included in Part I, Item 1 of t his report for a discussion of T he Transaction s

The Company is the largest communications company providing services predominantly to rural areas and small and medium-sized towns and cities in the United States. The Company currently operates in 27 states and will also operate in Connecticut after consummation of the AT&T Transaction . The Company is the nation's fourth largest Incumbent Local Exchange Carrier (ILEC), with 3.0 million customers, 1.9 million broadband connections and 13,900 employees as of June 30, 2014.

Regulatory Developments

In October 2009, the FCC issued a proposed rulemaking looking at rules to "Preserve a Free and Open Internet," (i.e., net neutrality), including a reconsideration of the legal classification of broadband and proposed restrictions on broadband network management practices. On December 21, 2010, the FCC adopted an order imposing some regulations on Internet service providers. These regulations affect fixed and mobile br oadband providers differently. These regulations became effective November 20, 2011, and we are in compliance with these regulations. On January 14, 2014, the U.S. Court of Appeals for the D.C. Circuit vacated the portions of the FCC's rules that prohibited blocking and required non-discriminatory treatment. On May 15, 2014, the FCC adopted a Notice of Proposed Rulemaking which sought comment on how it can preserve the spirit of the vacated rules. The Notice also sought comment on what other regulatory actions the FCC could take to implement "Open Internet" policies, in particular reclassifying broadband as a Title II Telecommunications service. The future state of these regulations and their effect on us is unknown.

On November 18, 2011, the FCC adopted an Order to reform the Federal Universal Service High-Cost Fund (USF) and Intercarrier C ompensation (the 2011 Order). The Order changed how federal subsidies are calculated and disbursed, with these changes being phased-in beginning in July 2012. These changes transition the USF, which supports voice services in high-cost areas, to the Connect America Fund (CAF), which supports broadband deployment in high-cost areas. CAF Phase I, implemented in 2012 to provide interim support while the FCC developed CAF Phase II, provide d for ongoing USF support for price cap carriers capped at the 2011 amount. In addition, the FCC in the FCC in the FCC in the first round of CAF Phase I made available for price cap ILECs an additional \$\$300 million in incremental high-cost broadband support between the provide and support of the price cap carriers capped at the 2011 amount. In addition, the FCC in the FCC in the first round of CAF Phase I made available for price cap ILECs an additional \$\$300 million in incremental high-cost broadband support by the provide and initially recorded such funds as increases to Cash and Other liabilities in the balance sheet. The \$71.9 million in the first round of CAF Phase I support is expected to enable an incremental 92,877 households for broadband service and was accounted for as Contribution s in Aid of Construction. We are required to implement, spend and enable these 92,877 households no later than July 24, 2015. We spent \$ 13.5 million of the first round of CAF Phase I funds on network expansion during the first six months of 2014 and \$51.1 million of these CAF Phase I funds to date through June 30, 2014, enabling 74,000 unserved households.

On May 21, 2013, the FCC released a Report and Order authorizing a second round of CAF Phase I (the 2013 Order). As part of this May 2013 Report and Order, the FCC expanded the areas eligible for funding to include those that lack service of 3 Mbps download and 768 kbps upload. In 2013, Frontier applied for funds from the second round of CAF Phase I and in December 2013 the FCC awarded \$57.6 million to us to serve 94,899 locations in our high-cost, unserved and underserved areas across multiple states. The FCC also announced on January 10, 2014, that Frontier was eligible for an additional \$3.7 million in fun ding to serve 6,815 locations. We received these funds in February 2014. The second round of CAF Phase I includes certain accompanying spending requirements from the Company, and our capital expenditure plans take this into account. We spent \$ 11. 1 million of the second round of CAF Phase I funds on network expansion during the first six months of 2014, enabling 32,300 unserved and underserved

In the aggregate, we spent \$24.6 million of the CAF Phase I funds received on network expansion and upgrades during the first six months of 2014 and \$62. 2 million of all CAF Phase I funds received through June 30, 2014, enabling 106,300 households in total

The FCC continues to finalize its CAF Phase II program, which is intended to provide long-term support for broadband in high-cost areas that were unserved by a competitor. The 2013 Order established that price cap carriers would have the right of first refusal to accept support across the high-cost footprint of their state in exchange for committing to provide broadband and voice services to those areas for five years. On June 10, 2014, the FCC released an Order and Notice of Proposed Rulemaking

seeking comment on potential changes to the CAF Phase II structure. The Notice seeks comment on whether the FCC should raise the broadband speed obligation for CAF Phase II from 4 Mbps download to 10 Mbps download. Recognizing that such a change in speed may also be more costly, the FCC seeks comment on whether it should lengthen the term of CAF Phase II support available to price cap carriers and also whether it should relax price cap carriers' obligation to deploy to 100% of the supported locations in exchange for a commensurate reduction in funding. These changes to the CAF Phase II program have not been finalized.

The 2011 Order also reformed Intercarrier Compensation, which is the payment framework that governs how carriers compensate each other for the exchange of interstate traffic, and it began a multi-year transition in July 2012, with the second step implemented in July 2013 and the third step in July 2014. The transition will move terminating traffic to a near zero rate by 2017. Frontier expects to be able to recover a significant portion of those revenues through end user rates and other replacement support mechanisms. The 2011 Order has been challenged by certain parties in court and certain parties have also petitioned the FCC to reconsider various aspects of the 2011 Order. On May 23, 2014, the U.S. Court of Appeals for the 10 *Circuit issued a decision denying all challenges to the 2011 Order. Certain parties have continued to appeal that decision. Accordingly, we cannot predict the long-term impact at this time but believe that the 2011 Order will provide a stable regulatory framework to facilitate our ongoing focus on the deployment of broadband into our rural markets.

Effective December 29, 2011, the 2011 Order required providers to pay interstate access rates for the termination of VoIP toll traffic. On April 25, 2012, the FCC, in an Order on Reconsideration, specified that changes to originating access rates for VoIP traffic will not be implemented until July 2014. The 2011 Order has been challenged by certain parties in court and certain parties have also petitioned the FCC to reconside r various aspects of the 2011 Order. The net impact of the 2011 Order during the period from July 2012 through June 2014 was insignificant. The net effect of this change to originating access after July 1, 2014 is dependent upon the percentage of VoIP traffic.

Certain states also have their own open proceedings to address reform to intrastate access charges and other intercarrier compensation and state universal service funds. Although the FCC has pre-empted state jurisdiction on most access charges, many states could consider moving forward with their proceedings. We cannot predict when or how these matters will be decided or the effect on our subsid y or switched access revenues.

The following should be read in conjunction with Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in our Annual Report on Form 10-K for the year ended December 31, 2013

(a) Liquidity and Capital Resources

As of June 30, 2014, we had cash and cash equivalents aggregating \$801.7 million, excluding total restricted cash, primarily representing funds escrowed for future broadband expansion, of \$5.2 million. Our primary source of funds continued to be cash generated from operations. For the six months ended June 30, 2014, we used cash flow from operations, cash on hand and debt proceeds to fund principally all of our cash investing and financing activities, primarily capital expenditures, dividends and debt repayments.

We have a revolving credit facility with a line of credit of \$750.0 million that we believe provides sufficient flexibility to meet our liquidity needs. As of June 30, 2014, this credit facility was fully available and we had not made any borrowings under it.

At June 30, 2014, we had a working capital surplus of \$ 129.0 million. We believe our operating cash flows, existing cash balances, and existing revolving credit facility will be adequate to finance our working capital requirements, fund capital expenditures, make required debt payments, pay taxes, pay dividends to our stockholders, and support our short-term and long-term operating strategies for the next twelve months. However, a number of factors, including but not limited to, losses of customers, pricing pressure from increased competition, lower subsidy and switched access revenues, and the impact of the current economic environment may negatively impact our cash generated from operations. In addition, based on information available to us, we believe that the financial institutions syndicated under our revolving credit facility would be able to fulfill their commitments to us, but this could change in the future. As of June 30, 2014, we had \$ 30.3 million of debt maturing during the last six months of 2014; \$ 262.5 million and \$ 348.2 million of debt multimature in 2015 and 2016, respectively.

Cash Flows provided by Operating Activities

Cash flows provided by operating activities declined \$ 39.6 million, or 6 %, for the six months ended June 30, 2014, as compared with the prior year period. The decrease was primarily the result of lower revenue and net income before depreciation and amortization.

We paid \$ 14.4 million in net cash tax es during the first six months of 201 4 as compared to \$ 83.5 million during the first six months of 2013. Our 2013 cash taxes paid reflect ed the continued impact of bonus depreciation in accordance with the American Taxpayer Relief Act of 2012. Absent any legislative changes, we expect that in 2014 our cash tax payments for the full year will be approximately \$1.30 million to \$16.0 million for our current business operations, taking into account our estimated pre-close integration expenditures relating to the AT&T Transaction.

In connection with the pending AT&T Transaction, the Company recognized \$ 30.4 million of acquisition and integration costs incurred during the first six months of 2014 and \$ 15.0 million of interest expense related to the Bridge Facility commitment. The Company expects to incur operating expenses of approximately \$140 million to \$170 million in 2014 related to integration activities for the pending AT&T Transaction.

Cash Flows used by Investing Activities

Capital Expenditures

For the six months ended June 30, 2014 and 2013, our capital expenditures were \$ 302.2 million and \$ 326.5 million (including \$ 41.6 million of integration-related capital expenditures for the six months ended June 30, 2014 for the AT&T Transaction), respectively. In addition to the capital expenditures mentioned above, n etwork expansion funded by previously received CAF funds amounted to \$ 24.6 million and \$ 9.2 million for the six months ended June 30, 2014 and 2013, respectively. We anticipate capital expenditures for business operations to decrease in 2014 related to our currently owned properties to approximately \$ 575 million to \$62.5 million, as compared to \$ 634.7 million in 2013. We anticipate an additional \$85 million to \$105 million of capital expenditures in 2014 related to the integration activities of the pending AT&T Transaction.

Cash Flows used by and provided from Financing Activities

Debt Financings

During the first six months of 2014, we entered into secured financings totaling \$ 10.8 million with four year terms and no stated interest rate for certain equipment purchases.

Debt Reduction

During the first six months of 2014 and 2013, we retired an aggregate principal amount of \$ 229.6 million and \$ 1,534.1 million, respectively, of debt consisting of \$ 228. 7 million and \$ 1,533.9 million, secured debt and \$0.2 million of rural utilities service loan contracts in each period. Additionally, we retired \$0.7 million of other secured debt during the first six months of 2014

We may from time to time repurchase our debt in the open market, through tender offers, exchanges of debt securities, by exercising rights to call or in privately negotiated transactions. We may also refinance existing debt or exchange existing debt for newly issued debt obligations.

Bridge Facility
On December 16, 2013, we signed a commitment letter for a bridge loan facility (the Bridge Facility) and recognized interest expense related to this commitment of \$ 15.0 million during the six months ended June 30, 2014. On January 29, 2014, we entered into a bridge loan agreement with the Lenders party thereto and JPMorgan Chase Bank, N.A., as administrative agent (the Bridge Loan Agreement), pursuant to which the Lenders have agreed at closing of the AT&T Transaction to provide to us an unsecured bridge loan facility for up to \$1.9 billion for the purposes of funding (i) substantially all of the purchase price for the transaction of the AT&T Transaction. Pursuant to the Bridge Loan Agreement, if AT&T Transaction and (ii) the fees and expenses incurred in connection with the transactions contemplated by the stock purchase agreement for the AT&T Transaction. Pursuant to the Bridge Loan Agreement, if and to the extent we do not, or are unable to, issue debt securities yielding up to \$1.9 billion in gross cash proceeds on or prior to the closing of the AT&T Transaction, we shall draw down up to \$1.9 billion, less the amount of the debt securities, if any, issued by us on or prior to the closing of the AT&T Transaction, in aggregate principal amount of loans under the Bridge Facility to fund the purchase price of the AT&T

On July 16, 2014, the Bridge Facility was reduced by \$350.0 million following execution of the 2014 CoBank Credit Agreement, as defined below. All other terms and conditions remain unchanged.

Bank Financing

On June 2, 2014, the Company entered into a credit agreement with CoBank, ACB, as administrative agent, lead arranger and a lender, and the other lenders party thereto, for a \$350.0 million senior unsecured delayed draw term loan facility (the 2014 CoBank Credit Agreement). The facility will be drawn upon closing of the AT&T Transaction with proceeds used to partially finance the acquisition. The maturity date is the earlie r of the fifth anniversary of the draw date and December 15, 2019. Repayment of the outstanding principal balance will be made in quarterly installments in the amount of \$8.8 million, commencing one full fiscal quarter after the draw date, with the remaining outstanding principal balance to be repaid on the maturity date. Borrowings under the 2014 CoBank Credit Agreement will bear interest based on the margins over the Base Rate (as defined in the 2014 CoBank Credit Agreement) or LIBOR, at the election of the Company. Interest rate margins under the facility (ranging from 0.875% to 2.875% for Base Rate borrowings and 1.875% to 3.875% for LIBOR borrowings) are subject to adjustments based on the Total Leverage Ratio of the Company, as such term is defined in the 2014 CoBank Credit Agreement.

The Company has a credit agreement with CoBank, ACB, as administrative agent, lead arranger and a lender, and the other lenders party thereto, for a \$575.0 mil lion senior unsecured term loan facility with a final maturity of October 14, 2016 (the 2011 CoBank Credit Agreement). The entire facility was drawn upon execution of the 2011 CoBank Credit Agreement in October 2011. Repayment of the outstanding principal balance is made in quarterly installments in the amount of \$14.4 million, which commenced on March 31, 2012, with the remaining outstanding principal balance to be repaid on the final maturity date. Borrowings under the 2011 CoBank Credit Agreement bear interest based on the margins over the Base Rate (as defined in the 2011 CoBank Credit Agreement) or LIBOR, at the election of the Company. Interest rate margins under the facility (ranging from 0.875% to 2.875% for Base Rate borrowings and 1.875% to 3.875% for LIBOR borrowings) are subject to adjustments based on the Total Leverage Ratio of the Company, as such term is defined in the 2011 CoBank Credit Agreement. The current pricing on this facility is LIBOR plus 2.875%.

Revolving Credit Facility

No June 2, 2014, the Company entered into a new revolving credit agreement with JPMorgan Chase Bank, N.A., as administrative agent, the lenders party thereto and the other parties named therein (the New Revolving Credit Agreement), for a \$750.0 million revolving credit facility (the New Revolving Credit Facility) with a scheduled termination date of May 31, 2018 and terminated its existing revolving credit facility (the Prior Revolving Credit Facility) under the Credit Agreement, dated as of May 3, 2013, among the Company, JPMorgan Chase Bank, N.A., as administrative agent, the lenders party thereto and the other parties named therein (the Prior Revolving Credit Agreement). As of June 30, 2014, the New Revolving Credit Facility was fully available and no borrowing band been made thereunder. Associated commitment fees under the New Revolving Credit Facility will vary from time to time depending on the Company's debt rating (as defined in the New Revolving Credit Agreement) and were 0.400% per annum as of June 30, 2014. During the term of the New Revolving Credit Facility, the Company may borrow, repay and reborrow funds, and may obtain letters of credit, subject to customary borrowing conditions. Loans under the New Revolving Credit Facility will bear interest based on the alternate base rate or the adjusted LIBO Rate (each as determined in the New Revolving Credit Agreement), at the Company's election, plus a margin based on the Company's debt rating (ranging from 0.50% to 1.50% for alternate base rate borrowings and 1.50% to 2.50% for adjusted LIBO Rate borrowings). The current pricing on this facility would have been 1.00% or 2.00%, respect tively, as of June 30, 2014. Letters of credit issued under the New Revolving Credit Facility will also be subject to fees that vary depending on the Company's debt rating. The New Revolving Credit Facility is available for general corporate purposes but may not be u sed to fund dividend payments. The terms of the New Revolving Credit Facility are substantially similar to the terms the Prior Revolving Credit Facility .

<u>Covenants</u>
The terms and conditions contained in our indentures, the 2011 CoBank Credit Agreement, the 2014 CoBank Credit Agreement, the New Revolving Credit Agreement and the Bridge Loan Agreement include the timely payment of principal and interest when due, the maintenance of our corporate existence, keeping proper books and records in accordance with U.S. GAAP, restrictions on the incurrence of liens on our assets, and restrictions on asset sales and transfers, mergers and other changes in corporate control. We are not subject to restrictions on the payment of dividends either by contract, rule or regulation, other than that imposed by the General Corporation Law of the State of Delaware. However, we would be restricted under the 2011 CoBank Credit Agreement, the 2014 CoBank Credit Agreement, the New Revolving Credit Agreement and the Bridge Loan Agreement from declaring dividends if an event of default occurred and was continuing at the time or would result from the dividend declaration.

The 2011 CoBank Credit Agreement , the 2014 CoBank Credit Agreement and the New Revolving Credit Agreement each contain a maximum leverage ratio covenant. Under those covenants, we are required to maintain a ratio of (i) total indebtedness minus cash and cash equivalents (including restricted cash) in excess of \$50.0 million to (ii) consolidated adjusted EBITDA (as defined in the agreements) over the last four quarters not to exceed 4.50 to 1.

The 2011 CoBank Credit Agreement, the 2014 CoBank Credit Agreement, the New Revolving Credit Agreement, the Bridge Loan Agreement and certain indentures for our senior unsecured debt obligations limit our ability to create liens or merge or consolidate with other companies and our subsidiaries' ability to borrow funds, subject to important exceptions and qualifications.

As of June 30, 2014, we were in compliance with all of our debt and credit facility covenants.

Dividends

We currently intend to pay regular quarterly dividends. Our ability to fund a regular quarterly dividend will be impacted by our ability to generate cash from operations. The declarations and payment of future dividends is at the discretion of our Board of Directors, and will depend upon many factors, including our financial condition, results of operations, growth prospects, funding requirements, applicable law, restrictions in agreements governing our indebtedness and other factors our Board of Directors deem relevant.

Off-Balance Sheet Arrangements
We do not maintain any off-balance sheet arrangements, transactions, obligations or other relationships with unconsolidated entities that would be expected to have a material current or future effect upon our financial statements

Future Commitments

In our normal course of business we have obligations under certain non-cancelable arrangements for services. During 2012, we entered into a "take or pay" arrangement for the purchase of future long distance and carrier services. Our remaining commitments under the arrangement are \$145. 5 million and \$140.8 million for the years ending December 31, 2014 and 2015, respectively. As of June 30, 2014, we expect to utilize the services included within the arrangement and no liability for the "take or pay" provision has been recorded.

To satisfy all or part of certain capital investment commitments to three state regulatory commissions in connection with the 2010 Transaction , we placed a total of \$115.0 million in cash into several escrow accounts and obtained a letter of credit for \$190.0 million in 2010. In September 2013, the letter of credit t expired. Another \$72.4 million of cash in an escrow account was acquired in connection with the 2010 Transaction to be used for service quality initiatives in the state of West Virginia. As of June 30 , 2014, \$ 184.5 million had been released from the escrow a cocunts. As of June 30 , 2014, the Company had a restricted cash balance in the remaining escrow account of \$ 3.2 million , including interest earned. The aggregate amount of this escrow account will continue to decrease over time as we make the required capital expenditures in West Virginia.

The FCC and certain state regulatory commissions, in connection with granting their approvals of the 2010 Transaction, specified certain capital expenditure and operating requirements for the Acquired Territories for specified periods of time post-closing. These requirements focus primarily on certain capital investment commitments to expand broadband availability to at least 85% of the households throughout the Acquired Territories with minimum download speeds of 3 Mbps by the end of 2013. We are required to provide download speeds of 4 Mbps to at least 75%, 80% and 85% of the households throughout the Acquired Territories by the end of 2013, 2014 and 2015, respectively. As of December 31, 2012, we also met our FCC requirement to provide 3 Mbps coverage to 85% of the households in the Acquired Territories by the end of 2013. As of December 31, 2012, we also met our FCC requirement to provide 4 Mbps coverage to 75% and 80% of the households in the Acquired Territories by the end of 2013 and 2014, respectively. As of June 30, 2014, we expanded broadband availability in excess of 4 Mbps to 83.9 % of the households throughout the Acquired Territories.

As of June 30, 2014 and December 31, 2013, we had expanded our broadband availability to the households throughout the Company's territories as follows:

(In excess of)	As of June 30, 2014	As of December 31, 2013
1 Mbps	90%	90%
3 Mbps	88%	86%
4 Mbps	87%	83%
6 Mbps	83%	76%
12 Mbps	75%	61%
20 Mbps	54%	48%

Critical Accounting Policies and Estimates

We review all significant estimates affecting our consolidated financial statements on a recurring basis and record the effect of any necessary adjustment prior to their publication. Uncertainties with respect to such estimates and assumptions are inherent in the preparation of financial statements; accordingly, it is possible that actual results could differ from those estimates and changes to estimates could occur in the near term. The preparation of our interim financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect (i) the reported amounts of assets and liabilities at the date of the financial statements, (ii) the disclosure of contingent assets and liabilities, and (iii) the reported amounts of revenue and expense s during the reporting period. Actual results may differ from those estimates. Estimates and judgments are used when accounting for revenue recognition, including the allowance for doubtful accounts, impairment of long-lived assets, impairment of intangible assets, depreciation and amortization, pension and other postretirement benefits, in come taxes and contingencies, among others.

Management has discussed the development and selection of these critical accounting estimates with the Audit Committee of our Board of Directors and our Audit Committee has reviewed our disclosures relating to such estimates.

There have been no material changes to our critical accounting policies and estimates from the information provided in Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in our Annual Report on Form 10-K for the year ended December 31, 2013

New Accounting Pronouncements

Revenue Recognition
In May 2014, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) No. 2014-09, "Revenue from Contracts with Customers." This standard requires companies to recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. This new standard is effective for annual and interim reporting periods beginning after December 15, 2016. Early ad option is not permitted. C ompanies are permitted to either apply the requirements retrospectively to all prior periods presented, or apply the requirements in the year of adoption, through a cumulative adjustment. The Company is currently evaluating the impact of adopting the new standard on January 1, 2017, but has not yet selected a transition method or determined the impact of adoption on its consolidated financial statements.

Discontinued Operations
In April 2014, the FASB issued ASU No. 2014-08, "Reporting Discontinued Operations and Disclosures of Disposals of Components of an Entity" which raises the threshold for a disposal to qualify as discontinued operations and requires new disclosures for individually material disposal transactions that do not meet the definition of a discontinued operation. Under the new standard, companies report discontinued operations when they have a disposal that represents a strategic shift that has or will have a major impact on operations or financial results. This new standard will be applied prospectively and will be effective for the Company on January 1, 2015. This new standard is not expected to have a material impact on the Company's consolidated financial statements.

Internal Control - Integrated Framework

In May 2013, the Committee of Sponsoring Organizations of the Treadway Commission (COSO) issued its updated Internal Control – Integrated Framework (the 2013 Framework) and related illustrative documents. COSO will continue to make available its original Framework during the transition period extend ing to December 15, 2014. The Company currently utilizes COSO's original Framework, which was published in 1992 and is recognized as the leading guidance for designing, implementing and conducting internal controls over external financial reporting and assessing its effectiveness. The 2013 Framework is expected to help organizations design and implement internal control in light of many changes in business and operating environments since the issuance of the original Framework, broaden the application of internal control in addressing operations and reporting objectives, and clarify the requirements for determining what constitutes effective internal control. We plan to adopt the 2013 Framework in 2014 and do not expect that it will have a significant impact on the Company.

(b) Results of Operations

REVENUE

Revenue is generated pri marily through the provision of voice services, data services, video services, network access, carrier services and other Internet services. Such revenues are generated through either a monthly recurring fee or a fee based on usage, and revenue recognition is not dependent upon significant judgments by management, with the exception of a determination of a provision for uncollectible amounts.

Revenue for the three months ended June 30, 2014 decreased \$43.3 million, or 4 %, to \$1,147.3 million as compared with the three months ended June 30, 2013. Revenue for the six months ended June 30, 2013. The decline s during bo th the second quarter and the first six months of 2014 are primarily the result of decreases in voice services revenues and lower switched and nonswitched access revenue, partially offset by an increase in data services revenue, each as described in more detail below. Additionally, wireless revenue decreased by \$7.8 million for the six months ended June 30, 2014 due to the sale of our Mohave Cellular Limited Partnership (Mohave) interest on April 1, 2013.

Switched access and subsidy revenue of \$ 267.1 million represented 12 % of ou r revenues for the six months ended June 30, 2014. Switched access revenue was \$ 106.1 million for the six months ended June 30, 2014, or 5 % of our revenues, down from \$ 124.1 million, or 5 % of our revenues, for the six months ended June 30, 2013. Subsidy revenue was \$ 161.0 million, or 7 % of our revenues, for the six months ended June 30, 2013. We expect declining revenue trends in switched access revenue to continue during the remainder of 2014.

During the first six months of 2014, we lost 48,000 customers, net, as compared to a loss of 52,200 customers, net, during the first six months of 2013. We believe the improvement in customer retention in 2014 as compared to 2013 is due to simplified products and pricing, our local engagement strategy, improved customers service and our investments in our network. During the second quarter of 2014, we lost 34,000 customers, net, as compared to a loss of 14,000 customers, net, during the first quarter of 2014. Residential customer losses increased sequentially by 22,200 in the second quarter of 2014 as the Company focused a greater proportion of its residential marketing efforts on bundled customer additions, as compared to our Simply Broadband customer prospects.

Total residential revenue for the three and six months ended June 30, 2014 declined \$ 8.1 million, or 2 %, and \$26.7 million, or 3%, respectively, as compared to the three and six months ended June 30, 2013 pri marily as a result of decreases in voice services revenue and the sale of our interest in the Mohave partnership, partially offset by increases in data services revenue and increases in subscriber line charges due to additional access recovery charges are a result of regulatory changes related to Intercarrier Compensation reform and are expected to continue to increase, allowed by the FCC to partially offset regulatory mandated declines in switched access rates. We had approximately 2,762,100 and 2, 842,900 total residential customers as of June 30, 2014 and 2013, respectively. We lost approximately 31,800 and 16,400 residential customers, net, during the quarter s ended June 30, 2014 and 2013, respectively, and lost approximately 41,400 and 44,200 residential customers, net, during the six months ended June 30, 2014 and 2013, respectively. Average monthly residential revenue per customer (res idential ARPC) increased \$ 0.58, or 1 %, to \$ 59.64 and \$ 0.40, or 1%, to \$ 59.35, during the three and six months ended June 30, 2014 and 2013, respectively, as compared to the three and six months ended June 30, 2013. The overall increase in residen tial ARPC is due to a higher percentage of the Company's residential customer is that take broadband services, Frontier Secure products, broadband customer migration to higher speeds and certain pricing actions. The Company expects continuing increases in data services revenue, primarily driven by increased broadband subscribers, and continuing declines in voice services revenue.

Total business revenue for the three and $\,$ six months ended June 30, 2014 declined \$30.4 million, or 6%, and \$54.0 million, or 5%, respectively, as compared to the three and six months ended June 30, 2014 and 2013, principally as a result of decreases in our voice services revenue and wireless backhaul revenue. We had approximately 264,200 and 278,100 total business customers as of June 30, 2014 and 2013, respectively. We lost approximately $\,$ 2,200 and $\,$ 2,900 business customers, net, during the quarter sended June 30, 2014 and 2013, respectively, and lost approximately $\,$ 6,600 and $\,$ 8,000 business customers, net, during the six months ended June 30, 2014 and 2013, respectively. Average monthly business revenue per customer (business ARPC) decreased $\,$ 3.04 to $\,$ 648.71 and increased $\,$ 1.95 to $\,$ 649.97 during the three and six months ended June 30, 2014 as compared to the three and six months ended June 30, 2013. The decrease in business ARPC during the second quarter of 2014 as compared to 2013 is driven by the declines in our voice services revenue and wireless backhaul revenue. The overall increase in business ARPC during the six

months ended June 30, 2014 as compared to 2013 is primarily due to declining customer counts for our small business customers that carry a lower ARPC. The C ompany expects the declines in voice services revenues and wireless backhaul revenues from business customers to continue in 2014, mitigated, in part, by increases in data services revenues.

During the three and six months ended June 30, 2014, the Company added approximately 27,700 and 64,900 net broadband subscribers, respectively. During the three and six months ended June 30, 2013, the Company added approximately 29,500 and 57,700 net broadband subscri bers, respectively. As of June 30, 2014, approximately 7 2 % of our residential broadband customers subscribe d to a bundle of services. As of June 30, 2014, we were able to offer broadband to approximately 6.5 million households, or 90 % of the 7.2 million households in our markets. The increase in broadband subscribers contributed to our improved data services revenue performance. We continue to invest in network speed and capacity to support our goal of increasing broadband penetration and market share. We expect to continue to increase broadband subscribers d uring the remainder of 2014.

Management believes that customer counts and average monthly revenue per customer are important fac tors in evaluating our trends. Among the key services we provide to residential customers are voice service, data service and video service. We continue to explore the potential to provide additional services to our customer base, with the objective of meeting all of our customers' communications needs. For business customers we provide voice and data services, as well as a broad range of value-added services.

In the section "Other Financial and Operating Data" below is a table that presents customer counts, average monthly revenue per customer and customer churn. It also categorizes revenue into customer revenue (residential and business) and regulatory revenue (switched access and subsidy revenue). The decline in the number of customers was partially offset by increased penetration of additional higher revenue generating products sold to both residential and business customers, which has increased our average monthly revenue per customer. Economic conditions, a substantial loss of customers, and increasing competition could make it more difficult to sell our service offerings, which would adversely affect our revenue, profitability and cash flows.

OTHER FINANCIAL AND OPERATING DATA

Customers Broadband subscribers Video subscribers		1,9	4 026,281 931,521 393,901	As of December 31, 2 3,074 1,866 385	,280	(2	Decrease))% % %	1,8	% Incr 321,014 812,110 880,180	(3)% 7 % 4 %
		For the three months June 30, 2014	s ended	\$ Increase (Decrease)	% Increase (Decrease)		For the six months June 30, 2014	ended 2013	\$ Increase (Decrease)	% Increase (Decrease)
Revenue (in 000's): Residential Business Customer revenue Switched access and subsidy Total revenue	s	497,040 \$ 516,261 1,013,301 133,964 1,147,265 \$	505,181 546,669 1,051,850 138,683 1,190,533	\$ (8,141) (30,408) (38,549) (4,719)	(2)% (6)% (4)% (3)% (4)%	s	993,004 \$ 1,041,235 2,034,239 267,072 2,301,311 \$	1,019,706 \$ 1,095,282 2,114,988 280,941 2,395,929 \$	(26,702) (54,047) (80,749) (13,869) (94,618)	(3)% (5)% (4)% (5)% (4)%
Switched access minutes of use (in millions)		3,760	4,109		(8)%		7,703	8,399		(8)%
	_	As of or for the three months 2014	s ended June 30, 2013	% Increase (Decrease)		=	As of or for the six months a 2014	ended June 30, 2013	% Increase (Decrease)	
Residential Customer Metrics: Customers Revenue (in 000's) Average monthly residential revenue per customer '0' Customer monthly chum	s s	2,762,099 497,040 \$ 59.64 \$ 1.80%	2,842,883 505,181 59.06 1.64%	(2)%		s s	2,762,099 993,004 \$ 59.35 \$ 1.71%	2,842,883 1,019,706 58.95 1.64%	(3)% (3)% 1 % 4 %	
Business Customer Metrics: Customers Revenue (in 000's) Average monthly business	s	264,182 516,261 \$	278,131 546,669	(5)%		s	264,182 1,041,235 \$	278,131 1,095,282	(5)% (5)%	
Average monthly dustness revenue per customer	\$	648.71 \$	651.75	(0)%		S	649.97 \$	648.02	0 %	

⁽¹⁾ Calculation excludes the Mohave Cellular Limited Partnership (Mohave), which was sold to Verizon Wireless on April 1, 2013.

Note: As stated in our report for the quarterly period ended March 31, 2014, prior period revenue and certain operating statistics have been revised from the previously disclosed amounts to reflect the insignificant reclassification of certain revenues from residential to business and the related impact on average monthly revenue per customer amounts.

(\$ in thousands_)	For the three months ended June 30,						
·		2014		2013		Increase Decrease)	% Increase (Decrease)
Voice services Data and Internet services Other Customer revenue Switched access and subsidy Total revenue	\$	471,570 462,730 79,001 1,013,301 133,964 1,147,265	\$	513,800 467,428 70,622 1,051,850 138,683 1,190,533	\$	(42,230) (4,698) 8,379 (38,549) (4,719) (43,268)	(8)% (1)% 12 % (4)% (3)% (4)%
		For the six month	s ended June 30	,			
		2014		2013		Increase Decrease)	% Increase (Decrease)
Voice services Data and Internet services Other	\$	953,889 924,226 156,124	\$	1,039,744 922,264 152,980	\$	(85,855) 1,962 3,144	(8)% 0 % 2 %
Customer revenue Switched access and subsidy Total revenue	\$	2,034,239 267,072 2,301,311	\$	2,114,988 280,941 2,395,929	\$	(80,749) (13,869) (94,618)	(4)% (5)% (4)%

Voice services revenue for the three and six months ended June 30, 2014 decreased \$42.2 million, or 8 %, to \$471.6 million, and \$85.9 million, or 8%, to \$953.9 million, respectively, as compared with the Voice services revenue for the three and six months ended June 30, 2013, primarily due to the continued loss of voice extent, decreases in long distance services revenue and individual features packages, partially offset by increase d local voice charges to residential and business end users. Local and enhanced services revenue for the three and six months ended June 30, 2014 decreased \$ 3.7.8 million, or 9%, to \$ 384.1 million, or 9%, to \$ 377.9.9 million, respectively, primarily due to the continued loss of voice customers and, to a lesser extent, decreases in individual features packages, partially offset by increase d local voice charges to residential and business end users. Local and enhanced services revenue for the three and six months ended June 30, 2014 decreased \$ 3.7.8 million, or 9%, to \$ 384.1 million, or 9%, to \$ 377.9.9 million, respectively, primarily due to the continued loss of voice customers and, to a lesser extent, decreases in individual features packages, partially offset by increase d local voice charges to residential and business end users. Long distance services revenue for the three and six months ended June 30, 2014 decreased \$ 4.4 million, or 5%, to \$ 87.5 million, and \$ 11.2 million, or 6%, to \$ 174.0 million, respectively, primarily due to lower minutes of use driven by fewer customers, partially offset by increased cost recovery surcharge rates.

Data and Internet Services Pata and Internet Services revenue for the three months ended June 30, 2014 decreased \$ 4.7 million, or 1 %, to \$ 462.7 million, as compared with the three months ended June 30, 2013, primarily due to the decline in wireless backhaul revenue, partially offset by an increase in residential and business data services revenue. Data and Internet services revenue for the six months ended June 30, 2014 increased \$2.0 million to \$924.2 million, as compared with the six months ended June 30, 2013, primarily due to an increase in residential and business data services revenue, partially offset by the decline in wireless backhaul revenue. Data services revenue increased \$ 17.9 million, or 8 %, to \$ 249.3 million, and \$38.7 million, or 9%, to \$ 491.4 million, respectively, for the three and six months ended June 30, 2014, as compared with the same period s of 2013, primarily due to increases in the number of broadband subscribers and sales of Frontier Secure products. As of June 30, 2014, the number of the Company's broadband subscribers increased by approximately 119,400, or 7 %, since June 30, 2013. Data and Internet services also includes nonswitched access revenue from data transmission services to other carriers and highvolume business customers with dedicated high-capacity Internet and E thernet circuits. Nonswitched access revenue decreased \$ 22.6 million, or 10 %, to \$ 213.5 million, and \$36.7 million, or 8%, to \$432.8 million, respectively, for the three and six months ended June 30, 2014, as compared with the same period s of 2013, primarily due to lower monthly recurring charges attributable to a reduction in wireless backhaul revenue and other business revenues. C arrier servi ces excluding wireless backhaul revenue decreased slightly, as we continue to provide network bandwidth to other communications carriers.

We expect wireless data usage to continue to increase, which may drive the need for additional wireless backhaul capacity. Despite the need for additional capacity, we expect to experience declines in wireless backhaul revenue throughout the remainder of 2014, as compared to the same periods of the prior year, as our carrier partners migrate to Ethernet solutions at a lower price point.

Other

Other revenue for the three and six months ended June 30, 2014 increased \$ 8.4 million, or 12 %, to \$ 79.0 million, and \$3.1 million, or 2%, to \$156.1 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to lower bad debt expenses that are charged against revenue and increased cost recovery surcharge rates, partially offset by lower directory services revenue. For the six month period, we also experienced lower wireless revenue associated with the sale of our interest in the Mohave partnership on April 1, 2013.

Switched Access and Subsidy

Switched access and subsidy revenue for the three and dix months ended June 30, 2014 decreased \$ 4.7 million, or 3 %, to \$ 133.9 million, and \$13.9 million, or 5%, to \$267.1 million, respectively, as compared with the three and six months ended June 30, 2013. Switched access revenue decreased \$ 8.9 million, or 14%, to \$ 52.9 million, and \$18.0 million, or 15%, to \$106.1 million, respectively, for the first three and six months of 2014, as compared with the s ame period s of 2013, primarily due to the impact of a decline in minutes of use related to access line losses and the displacement of minutes of use by wireless, email and other communications services combined with a reduction due to the impact of the lower rates enacted by the FCC's intercarrier compensation reform in July 2013. Switched access and subsidy revenue includes subsidy payments we receive from federal and state agencies, including surcharges billed to customers that are remitted to universal service administrators. Subsidy revenue increased \$4.1 million, or 5%, to \$81.0 million, or 3%, to \$161.0 million, respectively, for the three and six months ended June 30, 2014, as compared with the same period s of 2013, primarily due to the higher contribution factor for end user USF in 2014.

Federal and state subsidies and surcharges (which are billed to customers and remitted to universal service administrators) for the Company were \$ 84.2 million, \$ 16.0 million and \$ 60.8 million, respectively, and \$ 161.0 million in total, or 7 % of our revenues, for the six months ended June 30, 2014 represents 4 % of our consolidated revenues. Total federal and state subsidies and surcharges were \$ 156.8 million, or 7 % of our revenues, for the six months ended June 30, 2013.

On November 18, 2011, the FCC adopted the 2011 Order to reform the USF and Intercarrier Compensation. The 2011 Order changed how federal subsidies are calculated and disbursed, with these changes being phased-in beginning in July 2012. These changes transition the USF, which supports voice services in high-cost areas, to the CAF, which supports broadband deployment in high-cost areas. CAF Phase I, implemented in 2012 to provide interim support while the FCC developed CAF Phase II, provide d for ongoing USF support for price cap carriers capped at the 2011 amount. In addition, the FCC in the first round of CAF Phase I made available for price cap ILECs an additional \$300 million in incremental high-cost broadband support to be used for broadband deployment to unserved areas. In 2012 and early 2013, we received \$71.9 million in the first round of CAF Phase I funds. The \$71.9 million in the first round of CAF Phase I support is expected to enable an incremental 92,877 households for broadband service and was accounted for as Contributions in Aid of Construction. We are required to spend the CAF Phase I funds to enable these 92,877 households no later than July 24, 2015. We initially recorded such funds as increases to Cash and Other lia bilities in the balance sheet. We spent \$13.5 million of the first round of CAF Phase I funds on network expansion during the first six months of 2014 and \$51.1 million of the se CAF Phase I funds to date through June 30, 2014, enabling 74,000 unserved households.

On May 21, 2013, the FCC released the 2013 Order authorizing a second round of CAF Phase I. As part of this May 2013 Report and Order, the FCC expanded the areas eligible for funding to incl ude those that lack service of 3 Mbps download and 768 kbps upload. In 2013, Frontier applied for funds from the second round of CAF Phase I and in December 2013 the FCC awarded \$57.6 million to us to serve 94,899 locations in our high-cost, unserved and underserved areas across multiple states. The FCC also announced on January 10, 2014, that Frontier was eligible for an additional \$3.7 million in funding to serve 6,815 locations. We received these funds in February 2014. The second round of CAF Phase I includes certain accompanying spending requirements from the Company, and our capital expenditure plans take this into account. We spent \$11.1 million of the second round of CAF Phase I funds on network expansion during the first six months of 2014, enabling 32,300 unserved and underserved households.

In the aggregate, we spent \$24.6 million of the CAF Phase I funds received on network expansion and upgrades during the first six months of 2014 and \$62. 2 million of all CAF Phase I funds received through June 30, 2014, enabling 106,300 households in total.

The FCC continues to finalize its CAF Phase II program, which is intended to provide long-term support for broadband in high-cost areas that were unserved by a competitor. The 2013 Order established that price cap carriers would have the right of first refusal to accept support across the high-cost footprint of their state in exchange for committing to provide broadband and voice services to those areas for five years. On June 10, 2014, the FCC released an Order and Notice of Proposed Rulemaking seeking comment on potential changes to the CAF Phase II structure. The Notice seeks comment on whether the FCC should raise the broadband speed obligation for CAF Phase II from 4 Mbps download to 10 Mbps download. Recognizing that such a change in speed may also be more costly, the FCC seeks comments on whether it should lengthen the term of CAF Phase II support available to price cap carriers and also whether it should relax price cap carriers' obligation to deploy to 100% of the supported locations in exchange for a commensurate reduction in funding. These changes to the CAF Phase II program have not been finalized.

The 2011 Order also makes changes to Intercarrier Compensation. Intercarrier Compensation, which is the payment framework that governs how carriers compensate each other for the exchange of interstate traffic, began a multi-year transition in July 2012, with the second step implemented in July 2013 and the third step in July 2014. The transition will move terminating traffic to a near zero rate by 2017. We expect to be able to recover a significant portion of those revenues through end user rates and other replacement support mechanisms. We do not expect these changes to have a significant impact on our revenues in 2014.

Effective December 29, 2011, the 2011 Order required providers to pay interstate access rates for the termination of VoIP toll traffic. On April 25, 2012, the FCC, in an Order on Reconsideration, specified that changes to originating access rates for VoIP traffic will not be implemented until July 2014. The 2011 Order has been challenged by certain parties in court and certain parties have also petitioned the FCC to reconsider various aspects of the 2011 Order. The net impact of the 2011 Order during the period from July 2012 through June 2014 was insignificant. The net effect of this change to originating access after July 1, 2014 is dependent upon the percentage of VoIP traffic.

Certain states also have their own open proceedings to address reform to intrastate access charges and other intercarrier compensation and state universal service funds. Although the FCC has pre-empted state jurisdiction on most access charges, many states could consider moving forward with their proceedings. We cannot predict when or how these matters will be decided or the effect on our subsidy or switched access revenues.

OPERATING EXPENSES

NETWORK ACCESS EXPENSES

(\$\sqrt{in thousands})	For the three mon				
	 2014		2013	\$ Increase (Decrease)	% Increase (Decrease)
Network access expenses	\$ 106,224	\$	107,114	\$ (890)	(1)%
	 For the six mont	hs ended June	2 30,		
	 2014		2013	 \$ Increase (Decrease)	% Increase (Decrease)
Network access expenses	\$ 213,316	\$	216,512	\$ (3,196)	(1)%

Network access expenses for the three and six months ended June 30, 2014 decreased \$ 0.9 million, or 1 %, to \$ 106.2 million, and \$ 3.2 million, or 1%, to \$ 213.3 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to decreased promotion al costs that are recorded in network access expenses, partially offset by an increase in network access expenses related to higher broadband subscriber counts.

OTHER OPERATING EXPENSES

(\$ in thousands_)	For the three mon	ths ended June 30,			
	 2014		2013	\$ Increase (Decrease)	% Increase (Decrease)
Wage and benefit expenses All other operating expenses	\$ 257,747 265,638 523,385	\$	287,317 246,698 534,015	\$ (29,570) 18,940 (10,630)	(10)% 8 % (2)%
	 For the six month	hs ended June 30,		_	
	 2014		2013	\$ Increase (Decrease)	% Increase (Decrease)
Wage and benefit expenses All other operating expenses	\$ 534,607 517,704	\$	578,891 496,623	\$ (44,284) 21,081	(8)% 4 %
The other operating expenses	\$ 1,052,311	\$	1,075,514	\$ (23,203)	(2)%

Wage and benefit expenses

Wage and benefit expenses for the three and six months ended June 30, 2014 decreased \$29.6 million, or 10 %, to \$257.7 million, and \$44.3 million, or 8%, to \$534.6 million, respectively, as compared to the three and six months ended June 30, 2013, primarily due to lower costs for compensation resulting from lower average employee headcount combined with lower costs for certain other benefits, including pension and OP EB expense, as discussed below. Wage and benefit expenses included \$1.2 million and \$6.7 million of severance costs for the six month periods in 2014 and 2013, respectively.

Pension and OPEB costs for the Company are included in our wage and benefit expenses. Pension and OPEB costs for the three months ended June 30, 2014 and 2013 were approximately \$ 14. 2 million and \$ 20.5 million, respectively. Pension and OPEB costs include pension and OPEB expense of \$ 18.2 million and \$ 25.5 million, less amounts capitalized into the cost of capital expenditures of \$ 4.0 million and \$ 5.0 million, respectively.

Pension and OPEB costs for the six months ended June 30, 2014 and 2013 were approximately \$ 28.5 million and \$ 41.0 million, respectively. Pension and OPEB costs include pension and OPEB expense of \$ 36.4 million and \$ 51.0 million, less amounts capitalized into the cost of capital expenditures of \$ 7.9 million and \$ 10.0 million, respectively.

Based on current assumptions and plan asset values, we estimate that our 201 4 pension and OPEB expenses will be approximately \$ 65 million to \$85 million for our current business operations, excluding amounts capitalized into the cost of capital expenditures, as compared to \$ 97.1 million in 201 3, excluding the impact of pension settlement costs and amounts capitalized into the cost of capital expenditures.

All other operating expenses

All other operating expenses for the three and six months ended June 30, 2014 increased \$ 18.9 million, or 8 %, to \$ 265.6 million, and \$21.1 million, or 4%, to \$517.7 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to an increase in certain litigation reserves in the first quarter of 2014, increased fleet and facilities costs and gross receipts state tax expenses.

DEPRECIATION AND AMORTIZATION EXPENSE

(\$ in thousands)	For the three months ended June 30,					
	 2014		2013		§ Increase Decrease)	% Increase (Decrease)
Depreciation expense Amortization expense	\$ 197,572 75,891 273,463	\$	210,828 87,021 297,849	\$	(13,256) (11,130) (24,386)	(6)% (13)% (8)%
	 For the six month	hs ended June 30,				
	 2014		2013		§ Increase Decrease)	% Increase (Decrease)
Depreciation expense Amortization expense	\$ 403,088 151,782	\$	427,482 174,042	\$	(24,394) (22,260)	(6)% (13)%
•	\$ 554,870	\$	601,524	\$	(46,654)	(8)%

Depreciation and amortization expense for the three and six months ended June 30, 2014 decreased \$ 24.4 million, or 8 %, to \$ 273.5 million, and \$46.7 million, or 8 %, to \$554.9 million, respectively, as compared to the three and six months ended June 30, 2013. Amortization expense decreased \$ 22.3 million in 2014, due to the amortization recognized on an accelerated method related to the customer base. Depreciation expense decreased \$ 24.4 million in 2014, primarily due to changes in the remaining useful lives of certain plant assets and a lower net asset base.

We annually commission an independent study to update the estimated remaining useful lives of our plant assets. The latest study was completed in the fourth quarter of 2013, and after review and analysis of the results, we adopted new lives for certain plant assets as of October 1, 2013. This change had an insignificant impact on depreciation expense for the first half of 2014. Our "composite depreciation rate" for plant assets was 6.24 % as a result of the study. We anticipate depreciation expense of approximately \$790 million to \$810 million for 2014 for our current business operations.

Amortization expense includes the apportionment of the cost of the intangible assets (primarily customer base) that were acquired in the 2010 Transaction based on an estimated useful life of nine years for the residential customer base and 12 years for the business customer base, amortized on an accelerated method. We anticipate amortization expense of approximately \$ 285 million for 201 4 for our current business operations.

ACQUISITION AND INTEGRATION COSTS

		For the three mont	hs ended June 30,
(\$ in thousands)	2	2014	2013
Acquisition and integration costs	\$	19,851	\$
		For the six month	is ended June 30,
	2	2014	2013
Acquisition and integration costs	\$	30,447	\$

During the fourth quarter of 2013, the Company began to incur acquisition and integration costs in connection with the AT&T Transaction. Acquisition costs include legal, financial advisory, accounting, regulatory and other related costs. Integration costs include expenses incurred in connection with the AT&T Transaction to integrate the network and information technology platforms and to enable other integration initiatives. The Company incurred \$ 30.4 million of operating expenses ,

consisting of \$ 1.6 million and \$ 28.8 million of acquisition and integration costs , respectively , related to the AT&T Transaction during the six months ended June 30 , 2014 . We also incurred \$ 41.6 million in capital expenditures related to integration activities during the six months ended June 30 , 2014. We anticipate closing the AT&T Transaction during the fourth quarter of 2014.

GAIN ON SALE OF MOHAVE PARTNERSHIP INTEREST

		and six months June 30,
(\$ in thousands)	2014	2013
Gain on sale of Mohave	•	14.601
partnership interest	\$	\$ 14,601

On April 1, 2013, the Company sold its 33 % % partnership interest in the Mohave Cellular Limited Partnership, in which Frontier was the General Partner. The Company recognized a gain on sale of approximately \$14.6 million before taxes in the second quarter of 2013.

INVESTMENT AND OTHER INCOME (EXPENSE), NET / LOSSES ON EARLY EXTINGUISHMENT OF DEBT / INTEREST EXPENSE / INCOME TAX EXPENSE (BENEFIT)

	For the three months ended June 30,						
(<u>Sin thousands</u>)		2014	2013			\$ Increase (Decrease)	% Increase (Decrease)
Investment and other income (expense), net	\$	(17)	\$	2,956	\$	(2,973)	(101)%
Losses on early extinguishment of debt	\$	-	\$	159,780	\$	(159,780)	(100)%
Interest expense	\$	167,611	\$	166,547	\$	1,064	1 %
Income tax expense (benefit)	\$	19,034	\$	(18,755)	\$	37,789	(201)%
	For the six months ended June 30,						
		2014		2013	_	\$ Increase (Decrease)	% Increase (Decrease)
Investment and other income, net	\$	1,378	\$	7,610	\$	(6,232)	(82)%
Losses on early extinguishment of debt	\$	-	\$	159,780	\$	(159,780)	(100)%
Interest expense	\$	338,568	\$	337,967	\$	601	0 %
Income tax expense	\$	36,223	\$	14,520	\$	21,703	149 %

Investment and other i ncome (expense), net
Investment and other income (expense), net for the three and six months ended June 30, 2014 decreased \$ 3.0 million and \$6.2 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to proceeds of \$2.3 million in the settlement of a split-dollar life insurance policy for a former senior executive during the second quarter of 2013, a decrease of \$1.7 million in the settlement of customer advances and investment gains associated with cash received in connection with our previously written-off investment in Adelphia of \$1.4 million during the first half of 2013.

Our average cash balances were \$878.7 million and \$917.0 million for the six months ended June 30, 2014 and 2013, respectively. Our average total restricted cash balance s were \$10.7 million and \$35.4 million for the six months ended June 30, 2014 and 2013, respectively.

Losses on Early Extinguishment of Debt

During the second quarter of 2013, we recognized a loss of \$ 159.8 million on the early extinguishment of debt in connection with various debt tender offers, privately negotiated transaction s and open market repurchases that resulted in the retirement of \$1,002.4 million of the Company's senior notes.

Interest expense

Interest expense for the three and six months ended June 30, 2014 increased \$1.1 million, or 1%, to \$167.6 million, and \$0.6 million to \$338.6 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to the commitment fees of \$15.0 million on the Bridge Loan Facility during the first half of 2014, mostly offset by the lower average debt levels resulting from the debt refinancing activities and debt retirements of \$813.0 million during 2013. Our average debt outstanding was \$8,021.9 million and \$8,272.7 million for the second quarter of 2014 and 2013, respectively, and \$8,062.4 million for the six months ended June 30, 2014 and 2013, respectively. Our composite average borrowing rate as of June 30, 2014 and 2013 was 7.91 % and 7.95 %, respectively

Income tax expense (benefit) for the three and six months ended June 30, 2014 in cre ased \$ 37.8 million to \$19.0 million, and \$21.7 million to \$36.2 million, respectively, as compared with the three and six months ended June 30, 2013, primarily due to higher pretax income in 2014, re sulting from the \$159.8 million in losses recognized in 2013 on the early extinguishment of debt. The effective tax rate for the first six months of 201 4 and 201 3 was 32.0 % and 54.1 %, respectively.

Income taxes for the three and six months ended June 30, 2014 include a \$1.9 million benefit from the net reversal of reserves for uncertain tax positions. Income taxes for the six months ended June 30, 2014 include the impact of a \$2.9 million benefit arising from state tax law changes.

Income taxes for the three and six months ended June 30, 2013 include the impact of a charge of \$5.2 million resulting from the settlement of the 2010 IRS audit and a \$6.0 million charge resulting from the adjustment of deferred tax balances, partially offset by a \$4.4 million benefit from the net reversal of reserves for uncertain tax positions.

As of June 30, 2014, there were no uncertain tax positions for which the statute s of limitations are expected to expire during the next twelve months and which would affect our effective tax rate.

We paid \$ 14.4 million and \$ 83.5 million in net cash taxes during the six months ended June 30, 2014 and 2013, respectively. Our 2013 cash taxes paid reflected the continued impact of bonus depreciation in accordance with the American Taxpayer Relief Act of 2012. Absent any legislative changes, we expect that in 2014 our cash tax payments for the full year will be approximately \$130 million to \$160 million for our current business operations, taking into account our estimated pre-close integration expenditures related to the AT&T Transaction.

$Net\ income\ (loss)\ attributable\ to\ common\ shareholders\ of\ Frontier$

Net income (toss) attributable to common shareholders of Frontier for the second quarter of 2014 was \$ 37.7 million, or \$ 0.0 4 per share, as compared to net loss of \$ 38.5 million, or \$ 0.04 per share, in the second quarter of 2013, and net income of \$77.0 million, or \$ 0.0 8 per share, as compared to \$9.7 million, or \$ 0.01 per share, for the six months ended June 30, 2013.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

We are exposed to market risk in the normal course of our business operations due to ongoing investing and funding activities, including those associated with our pension plan assets. Market risk refers to the potential change in fair value of a financial instrument as a result of fluctuations in int erest rates and equity prices. We do not hold or issue derivative instruments, derivative commodity instruments or other financial instruments for trading purposes. As a result, we do not undertake any specific actions to cover our exposure to market risks, and we are not party to any market risk management agreements other than in the normal course of business. Our primary market risk exposures from interest rate risk and equity price risk are as follows:

Interest Rate Exposure

Our exposure to market risk for changes in interest rates relates primarily to the interest-bearing portion of our pension investment portfolio and related obligations, and floating rate indebtedness. As of June 30, 2014, 95 % of our long-term debt had fixed interest rate s with minimal exposure to interest rate changes. We had no interest rate swap agreements related to our fixed rate debt in effect at June 30, 2014.

Our objectives in managing our interest rate risk are to limit the impact of interest rate changes on earnings and cash flows and to lowe r our overall borrowing costs. To achieve these objectives, all but \$431.3 million of our outstanding borrowings at June 30, 2014 have fixed interest rates. In addition, our undrawn \$750.0 million revolving credit facility has interest rates that float with the LIBO Rate, as defined. Consequently, we have limited material future earnings or cash flow exposures from changes in interest rates on our long-term debt. An adverse change in interest rates would increase the amount that we pay on our variable rate obligations and could result in fluctuations in the fair value of our fixed each obligations. Based upon our overal 1 interest rate exposure at June 30, 2014, a near-term change in interest rates would not material ly affect our consolidated financial position, results of operations or cash flows.

At June 30, 2014, the fair value of our long-term debt was estimated to be approximately \$ 8.4 billion, based on our overall weighted average borrowing rate of 7.91 % and our overall weighted average maturity of approximately nine years. As of June 30, 2014, there has been no significant change in the weighted average maturity applicable to our obligations since December 31, 2013.

Equity Price Exposure

Our exposure to market risks for changes in equity security prices as of June 30, 2014 is limited to our pension plan assets. We have no other security investments of any significant amount.

The Company's pension plan assets have in creased from \$ 1,216.5 million at December 31, 2013 to \$ 1,276.5 million at June 30, 2014, an in crease of \$ 60.0 million, or 5%. This in crease is a result of positive investment returns of \$ 95.5 million and cash contributions of \$ 31.2 million, offset by benefit payments of \$ 66.7 million. An additional cash contribution of \$ 19.6 million was made on July 15, 2014. We expect that we will make total contributions of cash and/or other assets to our pension plan of approximately \$ 100 million in 2014.

Item 4. Controls and Procedures

(a) Evaluation of disclosure controls and procedures

We carried out an evaluation, under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, regarding the effectiveness of our disclosure controls and procedures (as defined in Rule s 13a-15(e) and 15d - 15(e) under the Securities Exchange Act of 1934, as amended). Based upon this evaluation, our principal executive officer and principal financial officer concluded, as of the end of the period covered by this report, June 30, 2014, that our disclosure controls and procedures were effective.

(b) Changes in internal control over financial reporting

We reviewed our internal control over financial reporting at June 30, 2014. There have been no changes in our internal control over financial reporting identified in an evaluation thereof that occurred during the second fiscal quarter of 2014 that material ly affected, or is reasonably likely to material ly affect, our internal control over financial reporting.

In May 2013, the Committee of Sponsoring Organizations of the Treadway Commission (COSO) issued its updated Internal Control – Integrated Framework (the 2013 Framework) and related illustrative documents. COSO will continue to make available its original Framework during the transition period extending to December 15, 2014. The Company currently utilizes COSO's original Framework, which was published in 1992 and is recognized as the leading guidance for designing, implementing and conducting internal controls over external financial reporting and assessing their effectiveness. The 2013 Framework is expected to help organizations design and implement internal control in light of many changes in business and operating environments since the issuance of the original Framework, broaden the application of internal control in addressing operations and reporting objectives, and clarify the requirements for determining what constitutes effective internal control. We plan to adopt the 2013 Framework in 2014 and do not expect that it will have a significant impact on the Company.

PART II. OTHER INFORMATION FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES

I tem 1. Legal Proceedings

See Note 1 5 of the Notes to Consolidated Financial Statements included in Part I, Item 1 of this report. There have been no material changes to our legal proceedings from the information provided in Item 3. "Legal Proceedings" included in our Annual Report on Form 10-K for the year ended December 31, 2013.

We are party to various legal proceedings (including individual, class and putative class actions) arising in the normal course of our business covering a wide range of matters and types of claims including, but not limited to, general contract s, billing disputes, rights of access, taxes and surcharges, consumer protection, trademark and patent infringement, employment, regulatory, tort, claims of competitors and disputes with other carriers. Litigation is subject to uncertainty and the outcome of individual matters is not predictable. However, we believe that the ultimate resolution of all such matters, after considering insurance coverage or other indemnities to which we are entitled, will not have a material adverse effect on our financial position, results of operations, or our cash flows.

Item 1A. Risk Factors

There have been no changes to the Risk Factors described in Part 1, Item 1A. "Risk Factors" in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2013.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

There were no unregistered sales of equity securities during the quarter ended June 30, 2014.

ISSUER PURCHASES OF EQUITY SECURITIES

Period	Total Number of Shares Purchased	Average Price Paid per Share	
April 1, 2014 to April 30, 2014 Employee Transactions (1)	353	\$ 5.83	
May 1, 2014 to May 31, 2014 Employee Transactions (1)	976	\$ 5.82	
June 1, 2014 to June 30, 2014 Employee Transactions (1)	6,838	\$ 5.70	
Totals April 1, 2014 to June 30, 2014 Employee Transactions (1)	8,167	\$ 5.72	

⁽¹⁾ Includes restricted shares withheld (under the terms of grants under employee stock compensation plans) to offset minimum tax withholding obligations that occur upon the vesting of restricted shares. The Company's stock compensation plans provide that the value of shares withheld shall be the average of the high and low price of the Company's common stock on the date the relevant transaction occurs.

Item 4. Mine Safety Disclosure

Not applicable.

PART II. OTHER INFORMATION FRONTIER COMMUNICATIONS CORPORATION AND SUBSIDIARIES

Item 6. Exhibits

(a) Exhibits:

10.1	Credit Agreement dated as of June 2, 2014 with CoBank, ACB.
10.2	Offer of Employment Letter, dated January 15, 2014, between the Company and Mark D. Nielsen.
10.3	Offer of Employment Letter, dated June 9, 2014, between the Company and Donald W. Daniels, Jr.
31.1	Certification of Principal Executive Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934.
31.2	Certification of Principal Financial Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934.
32.1	Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS	XBRL Instance Document.
101.SCH	XBRL Taxonomy Extension Schema Document.
101.PRE	XBRL Taxonomy Presentation Linkbase Document.
101.CAL	XBRL Taxonomy Calculation Linkbase Document.
101.LAB	XBRL Taxonomy Label Linkbase Document.
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the R egistrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

FRONTIER COMMUNICATIONS CORPORATION (Registrant)

By: /s/ John M. Jureller

John M. Jureller

Executive Vice President and
Chief Financial Officer

Date: August 7, 2014

CREDIT AGREEMENT

by and among

FRONTIER COMMUNICATIONS CORPORATION, as the Borrower

and

THE LENDERS PARTY HERETO

and

COBANK, ACB, as Administrative Agent

Dated as of June 2, 2014

TABLE OF CONTENTS

			Page
	EXECU	TION COPY	1
I.	CERTA	IN DEFINITIONS	1
	1.1	Certain Definitions	1
	1.2	Construction	20
	1.3	Accounting Principles	20
	1.4	Rounding	21
	1.5	Holidays	21
	1.6	Covenant Compliance Generally	21
	1.7	Administration of Rates	21
II.	CREDI	T FACILITIES	2 2
	2.1	Term Loans	22
	2.2	Interest Rate Provisions	23
	2.3	Interest Periods	23
	2.4	Making of Loans	24
	2.5	Fees	25
	2.6	Notes	26
	2.7	Drawing Payments	26
	2.8	Interest Payment Dates	27
	2.9	Voluntary Prepayments and Reduction of Commitments	27
	2.10	[Intentionally omitted]	28
	2.11	Sharing of Payments by Lenders	28
	2.12	Defaulting Lenders	29
III.	INCRE.	ASED COSTS; TAXES; ILLEGALITY; INDEMNITY	30
	3.1	Increased Costs	30
	3.2	Taxes	3 2
	3.3	Illegality	3 6
	3.4	LIBOR Rate Option Unavailable; Interest After Default	36
	3.5	Indemnity	37
	3.6	Mitigation Obligations; Replacement of Lenders	37
	3.7	Survival	39
IV.	CONDI	TIONS OF LENDING	39
	4.1	Effectiveness	39

	4.2	Term Loans	41
V.	REPRE	SENTATIONS AND WARRANTIES	43
	5.1	Organization, Powers, Governmental Approvals	43
	5.2	Financial Statements	44
	5.3	No Material Adverse Change	44
	5.4	Title to Properties; Possession Under Leases	44
	5.5	Ownership of Subsidiaries	4 5
	5.6	Litigation; Compliance with Laws	4 5
	5.7	Agreements	4 6
	5.8	Federal Reserve Regulations	4 6
	5.9	Investment Company Act	46
	5.10	Use of Proceeds	46
	5.11	Tax Returns	46
	5.12	No Material Misstatements	46
	5.13	Employee Benefit Plans	46
	5.14	Insurance	47
	5.15	Patriot Act; FCPA	47
VI.	AFFIRI	MATIVE COVENANTS	47
	6.1	Existence; Businesses and Properties	4 8
	6.2	Maintaining Records	4 8
	6.3	Use of Proceeds	4 9
	6.4	CoBank Equity	4 9
VII.	NEGAT	TIVE CONVENANTS	50
	7.1	Liens; Restrictions on Sales of Receivables	50
	7.2	Ownership of the Principal Subsidiaries	51
	7.3	Asset Sales	51
	7.4	Mergers	51
	7.5	Restrictions on Dividends	5 2
	7.6	Transactions with Affiliates	52
	7.7	Subsidiary Indebtedness	52
	7.8	Anti-Terrorism Laws	5 3
VIII.	FINAN	CIAL COVENANTS AND REPORTING	53
	8.1	Total Leverage Ratio	53
	8.2	Financial Statements and Other Reports	53
IX.	EVENT	TS OF DEFAULT	5 6
	9.1	Events of Default	5 6

	9.2	Consequences of Even of Default	5 8
X.	THE ADMINISTRATIVE AGENT		
	10.1	Appointment and Authority	60
	10.2	Rights as a Lender	60
	10.3	No Fiduciary Duty	60
	10.4	Exculpation	6 1
	10.5	Reliance by the Administrative Agent	61
	10.6	Delegation of Duties	62
	10.7	Filing Proofs of Claim	62
	10.8	Resignation or Removal of the Administrative Agent	62
	10.9	Non-Reliance on the Administrative Agent and Other Lenders	63
	10.10	Enforcement	64
	10.11	No other Duties, etc.	64
	10.12	No Reliance on the Administrative Agent's Customer Identification	64
		Program	
XI.	MISCELLANEOUS		64
	11.1	Modifications, Amendments or Waivers	64
	11.2	No Implied Waivers; Cumulative Remedies	66
	11.3	Expenses; Indemnity; Damage Waiver	66
	11.4	Notices; Effectiveness; Electronic Communication	68
	11.5	Severability	70
	11.6	Duration; Survival	70
	11.7	Successors and Assigns	70
	11.8	Confidentiality	7 5
	11.9	Counterparts; Integration; Effectiveness	7 6
	11.10	CHOICE OF LAW; SUBMISSION TO JURISDICTION; WAIVER	
		OF VENUE; SERVICE OF PROCESS; WAIVER OF JURY TRIAL	76
	11.11	USA Patriot Act Notice	7 8
	11.12	Payments Set Aside	7 8
	11.13	Interest Rate Limitation	78

LIST OF SCHEDULES AND EXHIBITS

S CHEDULES

SCHEDULE 1.1(B) COMMITMENTS OF LENDERS AND ADDRESSES FOR

NOTICES

SCHEDULE 7.7 INDEBTEDNESS

SCHEDULE 11.7 VOTING PARTICIPANTS

EXHIBITS

ASSIGNMENT AND ASSUMPTION COMPLIANCE CERTIFICATE EXHIBIT A EXHIBIT B EXHIBIT C LOAN REQUEST EXHIBIT D TERM LOAN NOTE

SOLVENCY CERTIFICATE EXHIBIT E

TAX COMPLIANCE CERTIFICATES CONVERSION OR CONTINUATION NOTICE EXHIBIT F

EXHIBIT G

CREDIT AGREEMENT

THIS CREDIT AGREEMENT (this "Agreement") is dated as of June 2, 2014 and is made by and among FRONTIER COMMUNICATIONS CORPORATION, a Delaware corporation, as Borrower (as hereinafter defined), the LENDERS (as hereinafter defined), and COBANK, ACB, in its capacity as Administrative Agent (as hereinafter defined).

The Borrower has requested that the Lenders provide to the Borrower a delayed draw term loan facility in an aggregate principal amount not to exceed \$350,000,000, the proceeds of which are to be used to purchase the wireline business of AT&T Inc. in the State of Connecticut and to pay fees and expenses with respect to the foregoing transactions. In consideration of their mutual covenants and agreements hereinafter set forth and intending to be legally bound hereby, the parties hereto covenant and agree as follows:

I. CERTAIN DEFINITIONS

- 1.1 Certain Definitions. In addition to words and terms defined elsewhere in this Agreement, the following words and terms shall have the following meanings, respectively, unless the context hereof clearly requires otherwise:
 - "Acquisition" means the acquisition by the Borrower of the wireline business of AT&T Inc. in the State of Connecticut.
 - "Acquisition Agreement" means that certain Stock Purchase Agreement, dated December 16, 2013, between AT&T, Inc. and the Borrower.
- "Adjusted LIBOR Rate" means for the Interest Period for any LIBOR Rate Loan, an interest rate per annum (rounded upward, if necessary, to the next whole multiple of 1/100 of 1%) equal to (i) the LIBOR Rate for such Interest Period multiplied by (ii) the Statutory Reserve Rate for such Interest Period.
- " Administrative Agent" means CoBank, in its capacity as administrative agent under the Loan Documents and any successor in such capacity appointed pursuant to Section 10.8.
 - "Administrative Questionnaire" means an administrative questionnaire in a form supplied by the Administrative Agent.
- "Affiliate" means, with respect to any specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls, is Controlled by or is under common Control with the Person specified.
- "Alternate Base Rate" means a rate per annum determined by the Administrative Agent on the first Business Day of each week, which shall be the highest of (a) the Prime Rate, (b) the Federal Funds Effective Rate <u>plus</u> one half of one percent (0.50%) per annum, and (c) the Adjusted LIBOR Rate for an Interest Period of one month on such day <u>plus</u> one percent (1.00%) per annum; <u>provided that</u>, if the LIBOR Rate is no longer available for such Interest Period, the Adjusted LIBOR Rate shall be calculated for such Interest Period as the Administrative Agent

1

shall select in its sole discretion. Any change in the Alternate Base Rate due to a change in the calculation thereof shall be effective at the opening of business on the first Business Day of each week or, if determined more frequently, at the opening of business on the first Business Day immediately following the date of such determination and without necessity of notice being provided to the Borrower or any other Person.

- "Anti-Terrorism Laws" means any Laws relating to terrorism, "know your customer" or money laundering, including Executive Order No. 13224, the USA Patriot Act, the Laws comprising or implementing the Bank Secrecy Act, and the Laws administered by the United States Treasury Department's Office of Foreign Asset Control.
 - "Applicable Margin" means, as applicable:
- (i) the percentage spread to be added to the Alternate Base Rate applicable to Base Rate Loans based on the Total Leverage Ratio then in effect according to the Pricing Grid, or
- (ii) the percentage spread to be added to the Adjusted LIBOR Rate applicable to LIBOR Rate Loans based on the Total Leverage Ratio then in effect according to the Pricing Grid.
- "Approved Fund" means any Fund that is administered or managed by (i) a Lender, (ii) an Affiliate of a Lender or (iii) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Asset Exchange" means the exchange or other transfer of telecommunications assets between or among the Borrower and another Person or other Persons in connection with which the Borrower would transfer telecommunications assets and/or other property in consideration of the receipt of telecommunications assets and/or other property having a fair market value substantially equivalent to those transferred by the Borrower (as determined in good faith by the Borrower's Board of Directors); provided that the principal value of the assets being transferred to the Borrower shall be represented by telecommunications assets.
- "Assignment and Assumption" means an assignment and assumption agreement entered into by a Lender and an assignee permitted under Section 11.7, in substantially the form of Exhibit \underline{A} or any other form approved by the Administrative Agent.
 - "Bankruptcy Code" means title 11 of the United States Code.
 - "Base Rate Loan" means a Term Loan bearing interest calculated in accordance with the Base Rate Option.
- "Base Rate Option" means the option of the Borrower to have portions of the Term Loans bear interest at the rate and under the terms set forth in Section 2.2(a) (i).
 - "Board" means the Board of Governors of the Federal Reserve System of the United States of America.

- "Borrower" means Frontier Communications Corporation, a corporation organized and existing under the laws of the State of Delaware.
- "Borrowing" means as of any date of determination (a) with respect to LIBOR Rate Loans outstanding as of such date, a borrowing consisting of Term Loans having the same Interest Period, and (b) with respect to Base Rate Loans, all Base Rate Loans outstanding as of such date.
- "Borrowing Date" means, with respect to any Term Loan, the date for the making thereof or the renewal or conversion thereof at or to the same or a different Interest Rate Option, which shall be a Business Day.
- "Budget" means, for the Borrower and its Subsidiaries on a consolidated basis, forecasted; (A) balance sheets; (B) profit and loss statements; (C) cash flow statements; (D) operating budget; and (E) capital budget, all prepared on a consistent basis with the Borrower's historical financial statements, together with appropriate supporting details and a statement of underlying assumptions. The Budget represents and will represent as of the date thereof the good faith estimate of the Borrower and its senior management concerning the most probable course of its business.
- "Business Day" means any day other than a Saturday or Sunday or a legal holiday on which banks are authorized or required to be closed for business in Denver, Colorado or New York, New York and if the applicable Business Day relates to any LIBOR Rate Loan or Base Rate Loan determined by reference to the LIBOR Rate, such day must also be a day on which dealings in Dollar deposits by and between banks are carried on in the London interbank market.
- "Capital Lease Obligations" of any Person means the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP and, for the purposes of this Agreement, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.
- "Change in Control" shall be deemed to have occurred if (A) any Person or group (within the meaning of Rule 13d-5 of the Securities and Exchange Commission as in effect on the date hereof) shall own directly or indirectly, beneficially or of record, shares representing 50% or more of the aggregate ordinary voting power represented by the issued and outstanding capital stock of the Borrower; or (B) a majority of the seats (other than vacant seats) on the board of directors of the Borrower shall at any time have been occupied by Persons who were neither (1) nominated by the management of the Borrower, nor (2) appointed by directors so nominated; or (C) any Person or group (within the meaning of Rule 13d-5 of the Securities and Exchange Commission as in effect on the date hereof) shall otherwise directly or indirectly Control the Borrower.
- "Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (i) the adoption or taking effect of any Law, (ii) any change in any Law or in the administration, interpretation, implementation or application thereof by any Official Body or

(iii) the making or issuance of any request, rule, guideline or directive (whether or not having the force of Law) by any Official Body; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines, interpretations or directives thereunder or issued in connection therewith (whether or not having the force of Law) and (y) all requests, rules, regulations, guidelines, interpretations or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities (whether or not having the force of Law), in each case pursuant to Basel III, shall in each case be deemed to be a Change in Law regardless of the date enacted, adopted, issued, promulgated or implemented.

- "Closing Date" means the Business Day on which each of the conditions precedent in Section 4.1 has been satisfied or waived by the Requisite Lenders, which date is June 2, 2014.
 - "CoBank" means CoBank, ACB, a federally chartered instrumentality of the United States.
 - "CoBank Equities" has the meaning specified in Section 6.4.
 - " Code" means the Internal Revenue Code of 1986.
 - " Communications" has the meaning given in Section 11.4(d).
 - "Communications Act" means the Communications Act of 1934 and the rules and regulations of the FCC thereunder.
- "Communications System" means a system or business providing voice, data or video transport, connection, monitoring services or other communications and/or information services (including cable television), through any means or medium.
 - "Compliance Certificate" means a certificate of the Borrower, signed by a Financial Officer of the Borrower, substantially in the form of Exhibit B hereto.
- "Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.
- "Consolidated Net Worth" means, as at any date of determination, the consolidated stockholders' equity of the Borrower and its consolidated Subsidiaries, including redeemable preferred securities where the redemption date occurs after the Maturity Date, mandatorily redeemable convertible preferred securities, mandatorily convertible Indebtedness (or Indebtedness subject to mandatory forward purchase contracts for equity or similar securities) and minority equity interests in other persons, as determined on a consolidated basis in conformity with GAAP consistently applied.

- "Consolidated Tangible Assets" means, for any Person, total assets of such Person and its consolidated Subsidiaries, determined on a consolidated basis, less goodwill, patents, trademarks and other assets classified as intangible assets in accordance with GAAP.
- "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.
 - "Conversion or Continuation Notice" has the meaning specified in Section 2.3.
- "Debtor Relief Laws" means the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect.
 - "Default" means any event or condition that with notice or passage of time, or both, would constitute an Event of Default.
- "Defaulting Lender" means, subject to Section 2.12(b), any Lender that (i) has failed to (a) fund its Term Loans on the Business Day such Term Loan was required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's good faith determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (b) pay to the Administrative Agent or any other Lender any other amount required to be paid by it hereunder within two Business Days of the date when due, (ii) has notified the Borrower and the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund any Term Loan hereunder and states that such position is based on such Lender's good faith determination that a condition precedent to funding (which condition precedent, together written any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (iii) has failed, within three Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (iii) upon receipt of such written confirmation by the Administrative Agent and the Borrower), or (iv) has, or has a direct or indirect parent company that has, (a) become the subject of a proceeding under any Debtor Relief Law, or (b) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business

enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under any one or more of clauses (i) through (iv) above shall be conclusive and binding absent manifest error, and, subject to any cure rights expressly provided above, such Lender shall be deemed to be a Defaulting Lender (subject to Section 2.12) upon delivery of written notice of such determination to the Borrower and each Lender.

- "Default Rate" means, as of any date of determination, the following: (a) for Base Rate Loans, the rate determined in accordance with the Base Rate Option as of such date <u>plus</u> an additional margin of 2.00% per annum, (ii) for LIBOR Rate Loans, the rate determined in accordance with the LIBOR Rate Option as of such date <u>plus</u> an additional margin of 2.00% per annum, and (iii) for all other Obligations, the rate determined in accordance with the Base Rate Option as of such date <u>plus</u> an additional margin of 2.00% per annum.
 - "Delayed Draw Date" means the Business Day on which each of the conditions precedent in Section 4.2 has been satisfied or waived by the Requisite Lenders.
 - "Dollar ," "Dollars ," "U.S. Dollars " and the symbol "\$" means lawful money of the United States of America.
- "EBITDA" means, with respect to the Borrower and its Subsidiaries, on a consolidated basis, for the period of calculation, the sum of (A) (1) net income or deficit, as the case may be (excluding extraordinary gains, extraordinary non-cash losses, the write up of any assets, and any gain or loss on the sale of assets), (2) total interest expense, (3) depreciation and amortization expense, (4) dividends on preferred stock, (5) accrued income or franchise taxes, federal, state or local (whether paid or accrued as a liability), and (6) losses attributable to minority interests, investment losses and non-recurring charges for severance, restructuring and acquisition (including acquisition integration) costs, minus (B) the sum of (i) investment income, (ii) interest income, (iii) dividend and patronage income, (iv) income from unconsolidated subsidiaries, partnerships and joint ventures, and (v) other non-operating income (not otherwise included in clauses (i), (iii), (iii) and (iv)); in all cases in clauses (A) and (B) only to the extent otherwise included in calculating net income or deficit. For any period of calculation, EBITDA shall be adjusted to give pro forma effect to any Material Transaction, as determined reasonably and in good faith by a Financial Officer, during the period of calculation as if such Material Transaction occurred on the first day of such period of calculation, provided that such pro forma calculations shall only include such adjustments as are permitted under Regulation S-X of the SEC.
- "Eligible Assignee" means any Person that meets the requirements to be an assignee under Sections 11.7(b)(iii), 11.7(b)(v) and 11.7(b)(vi) (subject to such consents, if any, as may be required under Section 11.7(b)(iii)).
- "Environmental Laws" means all national, federal, state, provincial, municipal or local laws, statutes, ordinances, orders, judgments, decrees, injunctions, writs, policies and guidelines (having the force of law), directives, approvals, notices, rules and regulations and other applicable laws relating to environmental or occupational health and safety matters, including those relating to the Release or threatened Release of Specified Substances and to the generation,

use, storage or transportation of Specified Substances, each as in effect as of the date of determination.

- "ERISA" means the Employee Retirement Income Security Act of 1974.
- "ERISA Affiliate" means, at any time, any trade or business (whether or not incorporated) under common control with the Borrower such that such trade or business, together with the Borrower and all other ERISA Affiliates, are treated as a single employer under Section 4001(b)(1) of ERISA.
- "ERISA Termination Event" means (A) a "Reportable Event" described in Section 4043 of ERISA (other than a "Reportable Event" not subject to the provision for 30-day notice to the PBGC under such regulations), or (B) the withdrawal of the Borrower or any of its ERISA Affiliates from a Plan during a plan year in which it was a "substantial employer" as defined in Section 4001(a)(2) of ERISA, or (C) the filing of a notice of intent to terminate a Plan or the treatment of a Plan amendment as a termination under Section 4041 of ERISA, or (D) the institution of proceeding to terminate a Plan by the PBGC or (E) any other event or condition which might constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan.
 - "Event of Default" means any of the events described in Section 9.1 and referred to therein as an "Event of Default."
- "Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (i) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (a) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (b) that are Other Connection Taxes, (ii) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Term Loan or Term Loan Commitment pursuant to a law in effect on the date on which (a) such Lender acquires such interest in such Term Loans or Term Loan Commitment (other than pursuant to an assignment request by the Borrower under Section 3.6(b) or (b) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.2, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (iii) Taxes attributable to such Recipient's failure to comply with Section 3.2(g) and (iv) any U.S. federal withholding Taxes imposed under FATCA.
 - "Executive Order No. 13224" means the Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001.
 - "Farm Credit Lender" means a federally-chartered Farm Credit System lending institution organized under the Farm Credit Act of 1971.

- "FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof and any agreements entered into pursuant to Section 1471(b)(1) of the Code.
 - "FCC" means the Federal Communications Commission.
 - "FCPA" means the Foreign Corrupt Practices Act, 15 U.S.C. §§ 78dd-1, et seq.
- "Federal Funds Effective Rate" means, for any day, the rate of interest per annum (rounded upward, if necessary, to the nearest whole multiple of 1/100th of 1%) equal to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers on such day, as published by the Federal Reserve Bank of New York on such date, or if no such rate is so published on such day, on the most recent day preceding such day on which such rate is so published.
 - "Fee Letter" means that certain fee letter dated as of April 29, 2014 between the Borrower and the Administrative Agent.
- "Financial Officer" of any Person means the President, Chief Financial Officer, Chief Executive Officer, Vice President Finance, Executive Vice President, Chief Accounting Officer, Controller or Treasurer of such corporation. Any document delivered hereunder that is signed by a Financial Officer shall be conclusively presumed to have been authorized by all necessary corporate action on the part of the Borrower and such Financial Officer shall be conclusively presumed to have acted on behalf of the Borrower.
- "Foreign Lender" means (i) if the Borrower is a U.S. Person, a Lender that is not a U.S. Person, and (ii) if the Borrower is not a U.S. Person, a Lender that is resident or organized under the Laws of a jurisdiction other than that in which the Borrower is resident for tax purposes. For purposes of this definition, the United States of America, each State thereof and the District of Columbia shall be deemed to constitute a single jurisdiction.
- "Fund" means any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.
- "GAAP" means generally accepted accounting principles in the United States of America as are in effect from time to time, subject to the provisions of Section 1.3.
- "Governmental Approvals" means all authorizations, consents, approvals, licenses and exemptions of, registrations and filings with, and reports to, all Governmental Authorities
- "Governmental Authority" means the government of the United States of America or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or

pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank), including, without limitation, the FCC and any applicable PUC.

- "Guaranty" or "Guarantee" means, with respect to any Person, without duplication, any obligation, contingent or otherwise, of such Person pursuant to which such Person has directly or indirectly guaranteed or had the economic effect of guaranteeing any Indebtedness or other obligation or liability of any other Person and, without limiting the generality of the foregoing, any obligation, direct or indirect, contingent or otherwise, of any such Person (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation or liability (whether arising by virtue of partnership arrangements, by agreement to keep well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement condition or otherwise), (b) to purchase or lease property or services for the purpose of assuring another Person's payment or performance of any Indebtedness or other obligations or liabilities, (c) to maintain the working capital of such Person to permit such Person to pay such Indebtedness or other obligations or liabilities or (d) entered into for the purpose of assuring in any other manner the obligee of such Indebtedness or other obligation or liability of the payment thereof or to protect such obligee against loss in respect thereof (in whole or in part); provided, that the term Guaranty/Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. Unless otherwise specified, the amount of any Guaranty shall be deemed to be the lesser of the principal amount of the Indebtedness or other obligations or liabilities guaranteed and still outstanding and the maximum stated or determined amount for which the guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Guaranty or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith.
- "Indebtedness" of any Person means, without duplication, (A) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind (other than customer deposits made in the ordinary course of business), (B) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (C) all obligations of such Person upon which interest charges are customarily paid, (D) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (E) all obligations of such Person issued or assumed as the deferred purchase price of property or services (other than current trade payables incurred, expense accruals and deferred compensation items arising, in each case, in such Person's ordinary course of business), (F) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, provided that, if such Person has not assumed such obligations, then the amount of Indebtedness of such Person for purposes of this clause (F) shall be equal to the lesser of the amount of the obligations of the holder of such obligations and the fair market value of the assets of such Person which secure such obligations as (G) all Capital Lease Obligations of such Person, (H) all obligations of such Person in respect of Swap Contracts (except to the extent such obligations are used as a bona fide hedge of other Indebtedness of such Person), provided the amount of such obligations shall be deemed to be the net termination obligations of such Person thereunder calculated as if such Swap Contracts were terminated on such date of calculation but such net termination shall not be less than zero for purposes of this definition, (I) all obligations of such Person as an account

party in respect of letters of credit and bankers' acceptances (except to the extent any such obligations are incurred in support of other obligations constituting Indebtedness of such Person and other than, to the extent reimbursed if drawn, letters of credit in support of ordinary course performance obligations), (J) obligations under partnership, organizational or other agreements to fund capital contributions or other equity calls with respect to any Person or investment, or to redeem, repurchase or otherwise make payments in respect to capital stock or other securities of such Person, and (K) all Guarantees of such Person in respect of any of the foregoing; provided, however, that the term Indebtedness shall not include endorsements for collection or deposit, in either case in the ordinary course of business.

- "Indemnified Taxes" means (i) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of the Borrower under any Loan Document and (ii) to the extent not otherwise described in the preceding clause (i), Other Taxes.
 - "Indemnitee" has the meaning specified in Section 11.3.
 - "Information" has the meaning specified in Section 11.8.
- "Insolvency Proceeding" means, with respect to any Person, (i) a case, action or proceeding with respect to such Person (a) before any court or any other Governmental Authority under any Debtor Relief Law or other similar Law now or hereafter in effect, or (b) for the appointment of a receiver, liquidator, assignee, custodian, trustee, sequestrator, conservator (or similar official) of the Borrower or otherwise relating to the liquidation, dissolution, winding-up or relief of such Person, or (ii) any general assignment for the benefit of creditors, composition, marshaling of assets for creditors, or other, similar arrangement in respect of such Person's creditors generally or any substantial portion of its creditors; undertaken under any Law.
 - "Interest Payment Date" means the last day of each calendar quarter after the date hereof until the Maturity Date.
- "Interest Period" means the period of time selected by the Borrower in connection with (and to apply to) any election permitted hereunder by the Borrower to have a Term Loan bear interest under the LIBOR Rate Option. Subject to the last sentence of this definition, such period shall be one, two, three, six, or, to the extent made available by all the Lenders, twelve months. Such Interest Period shall commence on the effective date of such LIBOR Rate Loan, which shall be (i) the Borrowing Date if the Borrower is requesting the Term Loans, or (ii) the date of renewal of or conversion to a LIBOR Rate Loan if the Borrower is renewing or converting any Term Loan. Notwithstanding the second sentence hereof: (a) any Interest Period that would otherwise end on a date that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (b) the Borrower shall not select, convert to or renew an Interest Period for any portion of the Term Loans that would end after the applicable Maturity Date and (c) if any Interest Period begins on the last Business Day of a month for which there is no numerically corresponding day in the month in which such Interest Period is to end, such Interest Period shall be deemed to end on the last Business Day of the final month of such Interest Period.

- "Interest Rate Option" means any (i) LIBOR Rate Option or (ii) Base Rate Option.
- "IRS" means the United States Internal Revenue Service.
- " Laws" means, collectively, all international, foreign, Federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law, including, without limitation, the Communications Act, any applicable PUC Laws and Environmental Laws
- "Lenders" means each of the financial institutions from time to time party hereto as a lender and their respective successors and assigns as permitted hereunder, each of which is referred to herein as a Lender.
- "LIBOR Rate" means, with respect to any Interest Period, a rate of interest (rounded upward to the next whole multiple of 1/100th of one percent) reported by Bloomberg Information Services (or on any successor or substitute service providing rate quotations comparable to those currently provided by such service, as determined by the Administrative Agent from time to time, for the purpose of providing quotations of interest rates applicable to dollar deposits in the London interbank market) at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, as the rate for dollar deposits with a maturity comparable to such Interest Period; provided that in the event the Administrative Agent is not able to determine the LIBOR Rate using such methodology, the Administrative Agent shall notify the Borrower and the Administrative Agent and the Borrower will agree upon a substitute basis for obtaining such quotations.
 - "LIBOR Rate Loan" means a Term Loan bearing interest at the LIBOR Rate Option.
 - "LIBOR Rate Option" means the option of the Borrower to have Term Loans bear interest at the rate and under the terms set forth in Section 2.2(a)(ii)_.
- " Licenses" means any cable television franchise or any landline telephone, cellular telephone, microwave, personal communications, or other telecommunications or similar license, authorizations, waiver, certificate of compliance, franchise (including cable television and telecommunications franchise), approval, or permit, whether for the acquisition, construction or operation of any Communications System, granted or issued by the FCC or any applicable PUC or other Governmental Authority and held by the Borrower or any of its Subsidiaries.
- "Lien" means, with respect to any asset, (a) any mortgage, deed of trust, lien, pledge, encumbrance, charge, or security interest in or on such asset, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease, or title retention agreement relating to such asset and (c) in the case of securities, any purchase option, call, or similar right of a third party with respect to such securities.

- "Loan Documents" means this Agreement, the Fee Letter, the Solvency Certificate and the Term Loan Notes.
- "Loan Request" means a request for the Term Loans, substantially in the form of Exhibit C hereto.
- "Margin Regulations" means Regulations T, U and X of the Board.
- "Material Adverse Effect" means a material adverse effect on the business, assets, operations, financial condition or results of operations of the Borrower and its Subsidiaries taken as a whole.
- "Material Transaction" means any acquisition or disposition outside the ordinary course of business of any property or assets that (A) constitute assets comprising all or substantially all of an operating unit of a business or equity interests of a Person representing a majority of the ordinary voting power or economic interests in such Person that are represented by all its outstanding capital stock and (B) involves aggregate consideration in excess of \$50,000,000.
- "Maturity Date" means the earliest of (a) the date of acceleration of the Obligations in accordance with Section 9.2, (b) the fifth anniversary of the Delayed Draw Date and (c) December 15, 2019.
- "Maximum Priority Amount" shall mean, at any time, the sum of (A) 10% of the value of the consolidated total assets of the Borrower and (B) 20% of the sum of the total consolidated current assets and net property, plant and equipment of the Borrower, in each case, as shown on, or computed from, the most recent quarterly or annual consolidated balance sheet of the Borrower delivered by the Borrower pursuant to Section 8.2(a) or 8.2(b).
 - "Maximum Rate" has the meaning specified in Section 11.13.
- "Multiemployer Plan" means any employee benefit plan that is a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA and to which the Borrower or any ERISA Affiliate is then making or accruing an obligation to make contributions or, within the preceding five (5) p lan years of such Multiemployer Plan, has made or had an obligation to make such contributions.
 - "Non-Consenting Lender" has the meaning specified in Section 11.1.
 - "Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.
- "Obligation" means any obligation or liability of any of the Borrower, howsoever created, arising or evidenced, whether direct or indirect, joint or several, absolute or contingent, for payment or performance, now or hereafter existing (and including obligations or liabilities arising or accruing after the commencement of any Insolvency Proceeding with respect to the Borrower or which would have arisen or accrued but for the commencement of such Insolvency Proceeding, even if the claim for such obligation or liability is not enforceable or allowable in

such proceeding), or due or to become due, under or in connection with this Agreement, the Term Loan Notes, the Fee Letter or any other Loan Document (regardless of whether any Term Loan is in excess of the amount committed under or contemplated by the Loan Documents or are made in circumstances in which any condition to any Term Loan is not satisfied) whether to the Administrative Agent, any of the Lenders or their Affiliates or other persons provided for under such Loan Documents.

- "Official Body" means (i) any Governmental Authority and (ii) any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).
- "Organizational Documents" means the certificate or articles of incorporation, bylaws, certificate of limited partnership, partnership agreement, certificate of formation, limited liability company agreement or other organizational documents of any Person.
- "Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Term Loan or Loan Document).
- "Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 3.6 (b)).
 - "Participant" has the meaning specified in Section 11.7(d).
 - "Participant Register" has the meaning specified in Section 11.7(d).
- "Payment In Full" means the payment in full in cash of the Term Loans and other Obligations (other than contingent indemnification Obligations) hereunder and the termination of the Term Loan Commitments.
 - "PBGC" means the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA.
- "Person" means any natural person, corporation, company, partnership, limited liability company, association, joint-stock company, trust, unincorporated organization, joint venture, Official Body, or any other entity.

"Plan" means any pension plan (including a multiemployer plan) subject to the provisions of Title IV of ERISA or Section 412 of the IRC which is maintained for or to which contributions are made for employees of the Borrower or any ERISA Affiliate.

" Pricing Grid" means the table and text set forth below:

Level	Total Leverage <u>Ratio</u>	Applicable Margin for Base Rate Loans	Applicable Margin for LIBOR Rate <u>Loans</u>
Level I	<u>></u> 4.00:1.00	2.875%	3.875%
Level II	≥3.50:1.00 < 4.00:1.00	2.375%	3.375%
Level III	<u>></u> 3.00:1.00 < 3.50:1.00	1.875%	2.875%
Level IV	≥2.50:1.00 < 3.00:1.00	1.375%	2.375%
Level V	< 2.50:1.00	0.875%	1.875%

For purposes of determining the Applicable Margin:

- (i) The initial Applicable Margin shall be set at Level II until the earlier of (x) delivery of the first Compliance Certificate after the Closing Date or (y) the date on which the first Compliance Certificate is due after the Closing Date pursuant to Section 8.2(c). The Applicable Margin shall be adjusted according to the above pricing grid as of the end of each full fiscal quarter after the Closing Date. Any increase or decrease in the Applicable Margin adjusted as of a quarter end shall be effective no later than five (5) Business Days following the date on which the Compliance Certificate evidencing such computation is delivered under Section 8.2(c). If a Compliance Certificate is not delivered when due in accordance with such Section 8.2(c), then the rates at Level I shall apply as of the first Business Day after the date on which such Compliance Certificate was required to have been delivered and shall remain in effect until the date on which such Compliance Certificate is delivered. Notwithstanding anything contained in this definition to the contrary, to the extent that the Applicable Margin shall change as a result of operation of this Section (i), such change shall not apply to any existing LIBOR Rate Loan until such time as the current Interest Period with respect to such LIBOR Rate Loan expires.
- (ii) Notwithstanding anything in clause (i) hereof, until receipt of the first Compliance Certificate delivered at least four fiscal quarters after the Delayed Draw Date, the Applicable Margin will be set at Level II if the Total Leverage Ratio is less than 4.00:1.00 for the applicable fiscal quarter, and Level I if the Total Leverage Ratio is greater than or equal to 4:00:1.00 for the applicable fiscal quarter.

- (iii) If, as a result of any restatement of or other adjustment to the financial statements of the Borrower or for any other reason, the Borrower or the Lenders determine that (i) the Total Leverage Ratio as calculated by the Borrower as of any applicable date was inaccurate and (ii) a proper calculation of the Total Leverage Ratio would have resulted in higher pricing for such period, the Borrower shall immediately and retroactively be obligated to pay to the Administrative Agent for the account of the applicable Lenders, promptly on demand by the Administrative Agent (or, after the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under the Bankruptcy Code, automatically and without further action by the Administrative Agent or any Lender), an amount equal to the excess of the amount of interest and fees that should have been paid for such period over the amount of interest and fees actually paid for such period. This paragraph shall not limit the rights of the Administrative Agent or any Lender, as the case may be, under Section 3.5, or Section VIII.
- "Prime Rate" means a variable rate of interest per annum equal to the "U.S. prime rate" as reported on such day in the Money Rates Section of the Eastern Edition of The Wall Street Journal is not published on such day, such rate as last published in the Eastern Edition of The Wall Street Journal. In the event the Eastern Edition of The Wall Street Journal ceases to publish such rate or an equivalent on a regular basis, the term "Prime Rate" shall be determined on any day by reference to such other regularly published average prime rate for such date applicable to such commercial banks as is acceptable to the Administrative Agent in its reasonable discretion. Any change in Prime Rate shall be automatic, without the necessity of notice provided to the Borrower.
- "Principal Office" means the main banking office of the Administrative Agent in Greenwood Village, Colorado, or such other banking office as may be designated by the Administrative Agent from time to time.
- "Principal Subsidiaries" shall mean any Subsidiary of the Borrower whose Consolidated Tangible Assets comprise in excess of 10% of the Consolidated Tangible Assets of the Borrower and its consolidated Subsidiaries as of the date hereof or at any time hereafter.
- "Pro Rata Share" means as of any date of determination, (i) if any Term Loan Commitments remain in effect, the proportion that a Lender's unused Term Loan Commitments bears to the aggregate amount of Term Loan Commitments of all of the Lenders as of such date, or (ii) if the Term Loan Commitments have been terminated or have expired, the proportion that the outstanding principal amount of a Lender's Term Loans as of such date bears to the aggregate outstanding principal amount of the Term Loans as of such date.
- "PUC" means any state, provincial or other local public utility commission, local franchising authority, or similar regulatory agency or body that exercises jurisdiction over the rates or services or the ownership, construction or operation of any Communications System (and its related facilities) or over Persons who own, construct or operate a Communications System, in each case by reason of the nature or type of the business subject to regulation and not pursuant to laws and regulations of general applicability to Persons conducting business in any such jurisdiction.

- "PUC Laws" means all relevant rules, regulations, and published policies of, and all Laws administered by, any PUC asserting jurisdiction over the Borrower or its Subsidiaries.
 - "Recipient" means (i) the Administrative Agent and (ii) any Lender, as applicable.
 - "Regulation D" means Regulation D of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.
- "Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of such Person and of such Person's Affiliates.
- "Release" means any spilling, emitting, discharging, depositing, escaping, leaching, dumping or other releasing, including the movement of any Specified Substance through the air, soil, surface water, groundwater or property, and when used as a verb has a like meaning.
 - "Removal Effective Date" has the meaning given in Section 10.8.
- "Requisite Lenders" means, to the extent more than one Lender holds any of the Term Loan Commitments or the outstanding principal amount of the Term Loans, at least two Lenders, including Voting Participants, who are not Defaulting Lenders and who have in the aggregate Pro Rata Shares greater than 50.00% (calculated without giving effect to any Term Loans held or deemed to be held by a Defaulting Lender); provided that for purposes hereof, such two (2) or more Lenders (including Voting Participants) may not consist solely of Voting Participants who purchased their participations from the same Lender or of Voting Participants and the Lender who sold such participations to such Voting Participant.
 - "Resignation Effective Date" has the meaning given in Section 10.8.
- "Restricted Payment" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other equity interest of the Borrower or any Subsidiary, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any such capital stock or other equity interest, or on account of any return of capital to the Borrower's stockholders, partners or members (or the equivalent Person thereof).
 - "Sanctioned Country" means, at any time, a country or territory which is the subject or target of any Sanctions.
- "Sanctioned Person" means, at any time, (a) any Person publicly identified in any Sanctions-related list of designated Persons maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury or the U.S. Department of State, (b) any Person that engages in material operations, or is organized or resident, in a Sanctioned Country or (c) any Person controlled by any such Person.

- "Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the U.S. government, including those administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury or the U.S. Department of State.
- "Securitization Transaction" means (A) any transfer of accounts receivable or interests therein (1) to a trust, partnership, corporation or other entity (other than a Subsidiary), which transfer or pledge is funded by such entity in whole or in part by the issuance to one or more lenders or investors of indebtedness or other securities that are to receive payments principally from the cash flow derived from such accounts receivable or interests in accounts receivable, or (2) directly to one or more investors or other purchasers (other than any Subsidiary), or (B) any transaction in which the Borrower or a Subsidiary incurs Indebtedness secured principally by Liens on accounts receivable. The "amount" of any Securitization Transaction shall be deemed at any time to be (i) in the case of a transaction described in clause (A) of the preceding sentence, the aggregate uncollected amount of the accounts receivable transferred pursuant to such Securitization Transaction, net of any such accounts receivable that have been written off as uncollectible, and (ii) in the case of a transaction described in clause (B) of the preceding sentence, the aggregate outstanding principal amount of the Indebtedness secured by Liens on accounts receivable incurred pursuant to such Securitization Transaction.
 - "Solvency Certificate" shall mean the certificate of the Borrower in the form of Exhibit E hereto, or otherwise in form acceptable to the Administrative Agent.
- "Specified Substance" means (i) any chemical, material or substance defined as or included in the definition of "hazardous substances", "hazardous waste", "restricted hazardous waste" or "toxic substances" or words of similar import under any applicable Environmental Laws; (ii) any (A) oil, natural gas, petroleum or petroleum derived substance, any drilling fluids, produced waters and other wastes associated with the exploration, development or production of crude oil, natural gas or geothermal fluid, any flammable substances or explosives, any radioactive materials, any hazardous wastes or substances, any toxic wastes or substances or (B) other materials or pollutants that, in the case of both (A) and (B), (1) pose a hazard to the property of the Borrower or any of its Subsidiaries or any part thereof or to persons on or about such property or to any other property that may be affected by the Release of such materials or pollutants from such property or any part thereof or to persons on or about such other property or (2) cause such property or such other property to be in violation of any Environmental Law; (iii) asbestos, urea formaldehyde foam insulation, toluene, polychlorinated biphenyls and any electrical equipment which contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of fifty parts per million; and (iv) any sound, vibration, heat, radiation or other form of energy and any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any Governmental Authority.
- "Statutory Reserve Rate" means, for the Interest Period for any LIBOR Rate Loan, a fraction (expressed as a decimal), the numerator of which is the number one and the denominator of which is the number one minus the arithmetic mean, taken over each day in such Interest Period, of the aggregate of the maximum reserve percentages (including any marginal, special,

emergency or supplemental reserves) expressed as a decimal established by the Board to which the Administrative Agent is subject for eurocurrency funding (currently referred to as "Eurocurrency liabilities" in Regulation D of the Board). Such reserve percentages shall include those imposed pursuant to such Regulation D. Eurocurrency Loans shall be deemed to constitute eurocurrency funding and to be subject to such reserve requirements without benefit of or credit for proration, exemptions or offsets that may be available from time to time to any Lender under such Regulation D or any comparable regulation. The Statutory Reserve Rate shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.

- "Subsidiary" of any Person at any time means any corporation, trust, partnership, any limited liability company or other business entity (i) of which more than 50% of the outstanding voting securities or other interests normally entitled to vote for the election of one or more directors or trustees (regardless of any contingency that does or may suspend or dilute the voting rights) is at such time owned, or the management of which is controlled, directly or indirectly through one or more intermediaries, or both, by such Person or one or more of such Person's Subsidiaries, or (ii) that is directly or indirectly Controlled by such Person or one or more of such Person's Subsidiaries.
- "Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.
- "Swap Termination Value" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) for any date on or after the date such Swap Contracts have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts (which may include a Lender or any Affiliate of a Lender).
- "Tax Compliance Certificate" means a tax certificate substantially in the form of $\underline{\text{Exhibit } F}$ hereto, prepared and delivered by any Lender in accordance with $\underline{\text{Section } 3.2(g)}$.

- "Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholdings), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.
 - "Term Loans" has the meaning given in Section 2.1(a).
- "Term Loan Commitment" means, as to any Lender at any time, the amount initially set forth opposite its name on Schedule 1.1(B), as such Term Loan Commitment is thereafter assigned or modified and "Term Loan Commitments" means the aggregate Term Loan Commitments of all of the Lenders. As of the Closing Date, the aggregate amount of the Term Loan Commitments of the Lenders is \$350,000,000.
- "Term Loan Commitment Termination Date" means the earliest of (i) the date on which the full amount of the Term Loan Commitments have been advanced to the Borrower, (ii) the date of the acceleration of the Obligations pursuant to Section 9.2., (iii) December 16, 2014 or (iv) the date the Borrower gives notice to the Administrative Agent of the Borrower's election to terminate the Term Loan Commitments; provided, that the Term Loan Commitment Termination Date may be extended to a date no later than June 16, 2015 upon delivery to the Administrative Agent of written notice from the Borrower that the "Termination Date" (as defined in the Acquisition Agreement) has been extended to such date.
 - "Term Loan Facility" means the term loan credit facility extended to the Borrower pursuant to Section 2.1(a).
 - "Term Loan Notes" means the promissory notes of the Borrower substantially in the form of Exhibit D hereto evidencing the Term Loans.
- "Total Indebtedness" means, as of any date, the aggregate principal amount of Indebtedness of the Borrower and its consolidated Subsidiaries outstanding as of such date, in the amount and only to the extent that such Indebtedness would be reflected on a balance sheet prepared as of such date on a consolidated basis in accordance with GAAP, minus the amount of the unrestricted cash and cash equivalents of the Borrower and its consolidated Subsidiaries in excess of \$50,000,000 that would be reflected on such balance sheet.
- "Total Leverage Ratio" means, with respect to any fiscal quarter, as of the date ending such fiscal quarter, the ratio of (a) Total Indebtedness as of such fiscal quarter end to (b) EBITDA for the most recently completed four fiscal quarters (including such fiscal quarter).
 - "Unused Delayed Draw Commitment Fee" has the meaning specified in Section 2.5(a).
- "USA Patriot Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56.
 - "U.S. Borrower" means any Borrower that is a U.S. Person.

- "U.S. Person" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.
- "Voting Participant" has the meaning specified in Section 11.7.
- "Voting Participant Notice" has the meaning specified in Section 11.7.
- "Withholding Agent" means (i) the Borrower and (ii) the Administrative Agent.
- 1.2 Construction. Unless the context of this Agreement otherwise clearly requires, the following rules of construction shall apply to this Agreement and each of the other Loan Documents: (a) references to the plural include the singular, the plural, the part and the whole; (b) the words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation"; (c) the words "hereof," "herein," "herein," "hereto" and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document as a whole; (d) article, section, clause, schedule and exhibit references are to this Agreement or other Loan Document, as the case may be, unless otherwise specified; (e) reference to any Person includes such Person's successors and assigns; (f) reference to any Governmental Authority includes any similar or successor Governmental Authority; (g) reference to any agreement, including this Agreement and any other Loan Document together with the schedules and exhibits hereto or thereto, document or instrument means such agreement, document or instrument as amended, extended, modified, supplemented, replaced, substituted for, superseded, renewed, refinanced, refunded or restated at any time and from time to time; (h) relative to the determination of any period of time, "from" means "from and including;" to "means" to but excluding," and "through" means "through and including"; (i) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights; (j) section headings herein and in each other Loan Document are included for convenience and shall not affect the interpretation of this Agreement or such Loan Document; (k) any pronoun shall include the corresponding masculine, feminine and neuter terms; (l) reference to any Law or regulation shall refer to such Law or regulation as amended, modified or supplemented fro
- 1.3 Accounting Principles. Except as otherwise provided in this Agreement, all computations and determinations as to accounting or financial matters (including financial ratios and other financial covenants) and all financial statements to be delivered pursuant to this Agreement shall be made and prepared in accordance with GAAP (including principles of consolidation where appropriate), applied on a consistent basis and, except as expressly provided herein, in a manner consistent with that used in preparing audited financial statements in accordance with Section 8.2(b) and all accounting or financial terms have the meanings ascribed to such terms by GAAP. Notwithstanding anything to the contrary herein, in the event of any change after the date hereof in GAAP, and if such change would affect the computation of any of the financial covenants set forth in Article VIII, then the parties hereto agree to endeavor, in

good faith, to agree upon an amendment to this Agreement that would adjust such financial covenants in a manner that would preserve the original intent thereof, but would allow compliance therewith to be determined in accordance with the Borrower's financial statements at that time, provided that until so amended such financial covenants shall continue to be computed in accordance with GAAP prior to such change therein. Notwithstanding the foregoing, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, Indebtedness of the Borrower and its Subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded.

- 1.4 Rounding. Any financial ratios required to be maintained pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio or percentage is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- 1.5 <u>Holidays</u>. Whenever payment of a Term Loan to be made or taken hereunder shall be due on a day that is not a Business Day such payment shall be due on the next Business Day (except as provided in <u>Section 2.3</u>) and such extension of time shall be included in computing interest and fees, except that the Term Loans shall be due on the Business Day preceding the Maturity Date if the Maturity Date is not a Business Day. Whenever any payment or action to be made or taken hereunder (other than payment of the Term Loans) shall be stated to be due on a day that is not a Business Day, such payment or action shall be made or taken on the next following Business Day, and such extension of time shall not be included in computing interest or fees, if any, in connection with such payment or action.
- 1.6 Covenant Compliance Generally. For purposes of determining compliance under Article VIII, any amount in a currency other than Dollars will be converted to Dollars in a manner consistent with that used in calculating consolidated net income in the most recent financial statements of the Borrower and its Subsidiaries delivered pursuant to Section 8.2. Notwithstanding the foregoing, for purposes of determining compliance with Article VII, with respect to any covenant with respect to the amount of Indebtedness or investment in a currency other than Dollars, no breach of any basket contained therein shall be deemed to have occurred solely as a result of changes in rates of exchange occurring after the time such Indebtedness or investment is incurred; provided, that for the avoidance of doubt, the result of any changes in rates of exchange occurring after the time such Indebtedness or investment is incurred shall otherwise apply in all other cases, including determining whether any additional Indebtedness or investment may be incurred at any time in accordance with Article VIII and for purposes of calculating financial ratios in accordance with Article VIII.
- 1.7 <u>Administration of Rates</u>. The Administrative Agent does not warrant, nor accept responsibility, nor shall the Administrative Agent have any liability with respect to the administration, submission or any other matter related to the rates in the definition of "LIBOR Rate" or with respect to any comparable or successor rate thereto

II. CREDIT FACILITIES

2.1 Term Loans.

- (a) <u>Term Loan Commitments</u>. Subject to the terms and conditions hereof, and relying upon the representations and warranties herein set forth, each Lender severally agrees to make a delayed draw term loan under its Term Loan Commitment (each such loan, a "**Term Loan**") to the Borrower on the Delayed Draw Date in a single advance in an aggregate amount not to exceed \$350,000,000; <u>provided</u> that the aggregate amount of Term Loans from each Lender shall not exceed such Lender's Term Loan Commitment. Each request by the Borrower for a Term Loan shall be deemed to be a representation by the Borrower that it shall be in compliance with <u>Section 4.2</u> immediately after giving effect to the requested Term Loans. The Term Loan Commitments are not revolving commitments, and the Borrower shall not have the right to repay and reborrow any Term Loan under this <u>Section 2.1</u>.
- (b) Term Loans Request. The Borrower shall request the Lenders to make the Term Loans by delivering to the Administrative Agent, not later than 11:00 a.m., (i) three (3) Business Days prior to the proposed Borrowing Date with respect to LIBOR Rate Loans; and (ii) one (1) Business Day prior to the proposed Borrowing Date with respect to Base Rate Loans (or, in each case, such shorter period as may be agreed to by the Administrative Agent in consultation with the Lenders), a duly completed request therefor substantially in the form of Exhibit C. The Loan Request shall be irrevocable and shall specify the Interest Period; provided that the Borrower may revoke the Loan Request on one (1) occasion due to a delay in the closing of the Acquisition by delivering a written notice to the Administrative Agent received prior to wiring of such advance, such revocation to be without premium or penalty (except as provided in Section 3.5). The Borrowing shall be in an aggregate amount equal to the aggregate Term Loan Commitments.
- (c) <u>Nature of Lenders' Obligations with Respect to Term Loans</u>. Each Lender shall be obligated to participate in each request for a Term Loan pursuant to this <u>Section 2.1</u> in accordance with its Pro Rata Share. The aggregate of each Lender's Term Loans outstanding hereunder to the Borrower at any time shall never exceed its Term Loan Commitment. The obligations of each Lender hereunder are several. The failure of any Lender to perform its obligations hereunder shall not affect the Obligations of the Borrower to any other party nor shall any other party be liable for the failure of such Lender to perform its obligations hereunder. The Lenders shall have no obligation to make the Term Loans after the Term Loan Commitment Termination Date.
- (d) Repayment of Term Loans. In addition to any prepayments or repayments made pursuant to Sections 2.11 and 2.12, the Borrower shall repay the aggregate outstanding principal balance of the Term Loans in equal quarterly principal payments on the last day of each March, June, September and December beginning one full fiscal quarter after the Delayed Draw D ate in an amount equal to \$8,750,000. Notwithstanding

anything herein to the contrary, the entire outstanding principal balance of the Term Loans shall be due and payable in full in cash on the Maturity Date.

- 2.2 Interest Rate Provisions. The Borrower shall pay interest in respect of the outstanding unpaid principal amount of the Base Rate Loans and LIBOR Rate Loans, it being understood that, subject to the provisions of this Agreement, the Borrower may select different Interest Rate Options and different Interest Periods to apply to different Borrowings at any time outstanding and may convert to or renew one or more Interest Rate Options with respect to all or any portion of any Borrowing (subject to minimum amounts in a principal amount equal to \$1,000,000 or any whole multiple of \$500,000 in excess thereof); provided that there shall not be at any one time outstanding more than five (5) Borrowings of LIBOR Rate Loans, and provided, further, that if an Event of Default or Default has occurred and is continuing, the Borrower may not request, convert to, or renew any LIBOR Rate Loans. If at any time the designated rate applicable to any Term Loan made by any Lender exceeds the Maximum Rate, the rate of interest on such Lender's Term Loan shall be limited to the Maximum Rate.
 - (a) <u>Interest Rate Options</u>. All other Obligations not constituting the Term Loans shall bear interest calculated based upon the Base Rate Option. Subject to the limitations set forth in <u>Section 3.4</u>, the Borrower shall have the right to select from the following Interest Rate Options applicable to the Term Loans:
 - (i) Base Rate Option: An option to pay interest at a fluctuating rate per annum equal to the Alternate Base Rate in effect as of any date of determination <u>plus</u> the Applicable Margin as of such date; or
 - (ii) LIBOR Rate Option: An option to pay interest at a fluctuating rate per annum equal to the Adjusted LIBOR Rate with respect to the applicable Interest Period and as in effect as of any date of determination <u>plus</u> the Applicable Margin as of such date.
 - (b) <u>Day Count Basis</u>. Interest on Base Rate Loans shall be calculated on the basis of a 365/366-day year for the actual number of days elapsed (except at times that the Alternate Base Rate is calculated based upon the LIBOR Rate, in which case the interest will be calculated on the basis of a 360-day year for the actual number of days elapsed.) Interest on LIBOR Rate Loans and all other Obligations, including amounts due under <u>Section 3.5</u>, shall be calculated on the basis of a 360-day year for the actual number of days elapsed. The date of funding or conversion of a LIBOR Rate Loan to a Base Rate Loan and the first day of an Interest Period shall be included in the calculation of interest. The date of payment of any Term Loan and the last day of an Interest Period shall be excluded from the calculation of interest; <u>provided</u>, if a Term Loan is repaid on the same day that it is made, one (1) day's interest shall be charged.
- **2.2** Interest Periods . In order to convert a Base Rate Loan or LIBOR Rate Loan or continue a LIBOR Rate Loan, the Borrower shall deliver to the Administrative Agent a duly completed, written request therefor substantially in the form of Exhibit G (each, a "Conversion or Continuation Notice") not later than 11:00 a.m. (i) with respect to a conversion to or

continuation of a LIBOR Rate Loan, at least three (3) Business Days prior to the proposed effective date of such conversion or continuation and (ii) with respect to a conversion to a Base Rate Loan, at least one (1) Business Day prior to the proposed effective date of such conversion. The Conversion or Continuation Notice shall specify (i) which Borrowings (including the principal amount thereof) are subject to such request, and, in the case of any LIBOR Rate Loan to be converted or continued, the last day of the current Interest Period therefor, (ii) the proposed effective date of such conversion or continuation (which shall be a Business Day), (iii) whether the Borrower is requesting a continuation of LIBOR Rate Loans or a conversion of Borrowings from one interest rate option to the other interest rate option, and (iv) if a continuation of or conversion to LIBOR Rate Loans is requested, the requested Interest Period with respect thereto. In addition, the following provisions shall apply to any continuation of or conversion of any Borrowings:

- (a) <u>Amount of Loans</u>. Immediately after giving effect to such conversion or continuation, each Borrowing of Term Loans shall be in an amount no less than the applicable minimum amount for such Term Loans as set forth in <u>Section 2.2</u>.
- (b) <u>Commencement of Interest Period</u>. In the case of any borrowing of, conversion to or continuation of any LIBOR Rate Loan, the Interest Period shall commence on the date of advance or continuation of, or conversion to, any LIBOR Rate Loan and, in the case of immediately successive Interest Periods, each successive Interest Period shall commence on the date on which the immediately preceding Interest Period expires. Upon a conversion from a LIBOR Rate Loan to a Base Rate Loan, interest at the Base Rate Option shall commence on the last day of the existing Interest Period.
- (c) <u>Selection of Interest Rate Options</u>. If the Borrower elects to continue a LIBOR Rate Loan but fails to select a new Interest Period to apply thereto, then a one month Interest Period automatically shall apply. If the Borrower fails to duly request the continuation of any Borrowing consisting of LIBOR Rate Loans on or before the date specified and otherwise in accordance with the provisions of this <u>Section 2.3</u>, then such LIBOR Rate Loan automatically shall be converted to a Base Rate Loan, commencing on the last day of the Interest Period then in effect.

2.4 Making of Loans

(a) Notifications and Payments. The Administrative Agent shall, promptly after receipt by it of a Loan Request (or revocation thereof) pursuant to Section 2.1(b), notify the applicable Lenders of its receipt of such Loan Request (or revocation thereof) specifying the information provided by the Borrower and, in the case of a new Loan Request, the apportionment among the Lenders of the requested Term Loan as determined by the Administrative Agent in accordance with Section 2.1. Each applicable Lender shall remit the principal amount of their Pro Rata Share of the Term Loan to the Administrative Agent such that the Administrative Agent is able to, and the Administrative Agent shall, to the extent the Lenders have made funds available to it for such purpose and subject to the terms and conditions of Section 2.1, fund such Term Loan to the Borrower in U.S. Dollars and immediately available funds to the Borrower's

account specified in the Loan Request prior to 2:00 p.m. on the proposed Borrowing Date.

- (b) <u>Pro Rata Treatment of Lenders</u>. Each selection of, conversion to or renewal of any Interest Rate Option and each payment or prepayment by the Borrower with respect to principal and interest due from the Borrower hereunder to the Lenders shall (except as otherwise may be provided with respect to a Defaulting Lender and except as provided in <u>Section 2.2</u>, <u>Section 3.1</u> or <u>Section 3.6</u>) be payable ratably among the Lenders entitled to such payment in accordance with the amount of principal and interest then due or payable such Lenders as set forth in this Agreement.
- (c) Presumptions by the Administrative Agent. Unless the Administrative Agent shall have received notice from a Lender prior to the proposed Borrowing Date that such Lender will not make available to the Administrative Agent such Lender's Term Loan, the Administrative Agent may assume that such Lender has made such Term Loan available on such date in accordance with Section 2.1., and may, in reliance upon such assumption, make available to the Borrower a corresponding amount. In such event, if a Lender has not in fact made its Term Loans available to the Administrative Agent, then the applicable Lender and the Borrower severally agree to pay to the Administrative Agent forthwith on demand such corresponding amount with interest thereon, for each day from and including the date such amount is made available to the Borrower to but excluding the date of payment to the Administrative Agent, at (i) in the case of a payment to be made by such Lender, the greater of the Federal Funds Effective Rate and a rate as reasonably determined by the Administrative Agent in accordance with banking industry rules on interbank compensation and (ii) in the case of a payment to be made by the Borrower, the interest rate then applicable to Base Rate Loans. If such Lender pays its Term Loans to the Administrative Agent, then the amount so paid shall constitute such Lender's Term Loan. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Lender that shall have failed to make such payment to the Administrative Agent. If the Borrower and such Lender pay such interest for the same period, the Administrative Agent promptly shall remit to the Borrower the amount of interest paid by Borrower for such overlapping period. Nothing in this Section 2.4(c) or elsewhere in this Agreement or the other Loan Documents, including the provisions of Section 2.12, shall be deemed to require the Administrative Agent (or any other Lender) to advance funds on behalf of any Lender or to relieve any Lender as a r

2.5 <u>Fees</u>.

(a) <u>Unused Delayed Draw Commitment Fee</u>. Accruing from the Closing Date until the earlier to occur of (i) the Delayed Draw Date or (ii) Term Loan Commitment Termination Date, the Borrower agrees to pay to the Administrative Agent for the account of each Lender according to its Pro Rata Share of the Term Loan Commitments, a nonrefundable unused commitment fee (each an " **Unused Delayed Draw**

Commitment Fee ") equal to 0.40% (computed on the basis of a year of 360 days, as the case may be, and actual days elapsed) multiplied by the maximum available amount of the Term Loan Commitments; provided, however, that any Unused Delayed Draw Commitment Fee accrued with respect to the Term Loan Commitment of a Defaulting Lender during the period prior to the time such Lender became a Defaulting Lender and unpaid at such time shall not be payable by the Borrower so long as such Lender shall be a Defaulting Lender except to the extent that such Unused Delayed Draw Commitment Fee shall otherwise have been due and payable by the Borrower prior to such time; and provided further that no Unused Delayed Draw Commitment Fee shall accrue with respect to the Term Loan Commitment of a Defaulting Lender so long as such Lender shall be a Defaulting Lender. Subject to the provisos in the directly preceding sentence, all Unused Delayed Draw Commitment Fees shall be payable in arrears on each Interest Payment Date.

(b) Other Fees. The Borrower agrees to pay to the Administrative Agent such other fees as agreed in the Fee Letter.

2.6 Notes. The obligation of the Borrower to repay the aggregate unpaid principal amount of the Term Loans made to it by each Lender, together with interest thereon, shall, at the request of the applicable Lender, be evidenced by a Term Loan Note, dated the Delayed Draw Date payable to the order of such Lender or its registered assigns in a face amount equal to the Term Loan Commitment, as applicable, of such Lender. The Borrower hereby unconditionally promises to pay, to the order of each of the Lenders and the Administrative Agent or their registered assigns, as applicable, the outstanding principal amount of the Term Loans and other Obligations as provided in this Agreement and the other Loan Documents.

2.7 <u>Drawing Payments</u>.

- (a) Payments Generally. All payments and prepayments to be made in respect of principal, interest, Unused Delayed Draw Commitment Fees, other fees referred to in Section 2.5 or other fees or amounts due from the Borrower hereunder shall be payable prior to 11:00 a.m. on the date when due without presentment, demand, protest or notice of any kind, all of which are hereby expressly waived by the Borrower, and without set-off, counterclaim or other deduction of any nature, and an action therefor shall immediately accrue. Such payments shall be made to the Administrative Agent at the Principal Office for the account of Lenders to which they are owed, in each case in U.S. Dollars and in immediately available funds. The Administrative Agent statement of account, ledger or other relevant record shall, in the absence of manifest error, be conclusive as the statement of the amount of principal of and interest on the Term Loans and other amounts owing under this Agreement and shall be deemed an "account stated".
- (b) Payments by the Borrower; Presumptions by the Administrative Agent . Unless the Administrative Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Administrative Agent for the account of the

Lenders hereunder that the Borrower will not make such payment, the Administrative Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders severally agrees to repay to the Administrative Agent forthwith on demand the amount so distributed to such Lender, with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Administrative Agent, at the greater of the Federal Funds Effective Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation.

2.8 Interest Payment Dates. Interest on Base Rate Loans shall be due and payable in arrears on each Interest Payment Date. Interest on LIBOR Rate Loans shall be due and payable on the last day of each Interest Period for those Term Loans and, if such Interest Period is longer than three (3) months, also on the date that is the three-month anniversary of the first day of such Interest Period. Interest on the principal amount of each Term Loan or other monetary Obligation shall be due and payable on demand after such principal amount or other monetary Obligation becomes due and payable (whether on the stated Maturity Date, upon an accelerated Maturity Date or otherwise).

2.9 Voluntary Prepayments and Reduction of Commitments .

- (a) <u>Right to Prepay</u>. The Borrower shall have the right at its option from time to time to prepay the Term Loans in whole or part without premium or penalty (except as provided in <u>Section 3.5</u>). Whenever the Borrower desires to prepay any part of the Term Loans, it shall provide a prepayment notice to the Administrative Agent by 11:00 a.m. at least (A) three (3) Business Days prior to the date of prepayment of LIBOR Rate Loans, or (B) one (1) Business Day prior to the date of prepayment of Base Rate Loans, setting forth the following information:
 - (i) the date, which shall be a Business Day, on which the proposed prepayment is to be made;
 - (ii) a statement indicating the application of the prepayment among Borrowings; and
- (iii) the total principal amount of such prepayment, which shall not be less than \$250,000 (provided, that the amount of any prepayment to which this <u>Section 2.9(a)(iii)</u> applies shall be in integral multiples of \$250,000).

All prepayment notices shall be irrevocable (provided that any such notice of a prepayment in full may be conditioned upon the effectiveness of other credit facilities, the incurrence of other Indebtedness or a change of control and revoked by notice to the Administrative Agent on or before 10:00 a.m. on the date specified in such notice for prepayment if such condition is not satisfied). The principal amount of the Term Loans for which a prepayment notice is given, together with interest on such principal amount

except with respect to Term Loans to which the Base Rate Option applies, shall be due and payable on the date specified in such prepayment notice as the date on which the proposed prepayment is to be made. All prepayments of the Term Loans will be applied pro rata to remaining installments of principal (including, without limitation, the final installment due at maturity.) Except as provided in Section 2.2., if the Borrower prepays a Term Loan but fails to specify the applicable Borrowing that the Borrower intends to prepay, and during the continuance of any Event of Default, then such prepayment shall be applied first, to all outstanding portions of the Term Loans that are Base Rate Loans, and second, to all outstanding portions of the Term Loans that are LIBOR Rate Loans. Any prepayment hereunder shall include all interest and fees due and payable with respect to the Term Loan being prepaid and shall be subject to the Borrower's Obligation to indemnify the Lenders under Section 3.5.

(b) <u>Reduction of the Term Loan Commitment</u>. The Term Loan Commitment shall be permanently reduced and terminated in full on the Term Loan Commitment Termination Date.

2.10 [Intentionally omitted]

- 2.11 Sharing of Payments by Lenders. If any Lender shall, by exercising any right of setoff, counterclaim or banker's lien, by receipt of voluntary payment, by realization upon security, or by any other non-pro rata source or otherwise, obtain payment in respect of any principal of or interest on its Term Loans or other obligations hereunder resulting in such Lender receiving payment of a proportion of the aggregate amount of its Term Loans and accrued interest thereon or other such Obligations greater than its pro-rata share of the amount such Lender is entitled hereunder, then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the Term Loans and such other Obligations of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Term Loans and other Obligations owing them, provided that:
 - (a) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest other than interest or other amounts, if any, required by Law (including court order) to be paid by the Lender or the holder making such purchase; and
 - (b) the provisions of this $\underline{\text{Section 2.11}}$ shall not be construed to apply to (x) any payment (including the application of funds arising from the existence of a Defaulting Lender) made by the Borrower pursuant to and in accordance with the express terms of the Loan Documents or (y) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Term Loans to any assignee or participant, other than to the Borrower or any Subsidiary thereof (as to which the provisions of this $\underline{\text{Section 2.11}}$ shall apply).

The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation. This $\underline{\text{Section 2.11}}$ shall not apply to any action taken by CoBank with respect to any CoBank Equities held by the Borrower, including pursuant to $\underline{\text{Section 9.2(c)}}$.

2.12 <u>Defaulting Lenders</u>.

- (a) <u>Defaulting Lender Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by applicable Law:
- (i) Waivers and Amendments . Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Requisite Lenders.
- (ii) Defaulting Lender Waterfall . Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Article IX or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to Section 9.2(c) shall be applied at such time or times as may be determined in good faith by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Term Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined in good faith by the Administrative Agent; third, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to satisfy such Defaulting Lender's potential future funding obligations with respect to its Term Loans under this Agreement; fourth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by the Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; fifth, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and sixth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of such Defaulting Lender's Term Loan was made at a time when the conditions set forth in Section 4.3 were satisfied or waived, such payment shall be applied solely to pay t

the Lenders pro rata in accordance with the Term Loan Commitments under the Term Loan Facility. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this <u>Section 2.12(a)</u> shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(iii) Certain Fees .

- (A) No Defaulting Lender shall be entitled to receive any Unused Delayed Draw Commitment Fee for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender).
- (B) With respect to any Unused Delayed Draw Commitment Fee not required to be paid to any Defaulting Lender pursuant to clause (A) above, the Borrower shall not be required to pay the remaining amount of any such fee.
- (b) <u>Defaulting Lender Cure</u>. If the Borrower and the Administrative Agent agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, that Lender will, to the extent applicable, purchase at par that portion of outstanding Term Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Term Loans to be held pro rata by the Lenders in accordance with the Term Loan Commitments under the Term Loan Facility, whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.

III. INCREASED COSTS; TAXES; ILLEGALITY; INDEMNITY

3.1 Increased Costs

- (a) Increased Costs Generally. If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement reflected in the Adjusted LIBOR Rate);
- (ii) subject any Recipient to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (ii) through (iv) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit,

commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto; or

(iii) impose on any Lender or the London interbank market any other condition, cost or expense (other than Taxes) affecting this Agreement or Term Loans made by such Lender;

and the result of any of the foregoing shall be to increase the cost to such Lender or such other Recipient of making, converting to, continuing or maintaining any Term Loan or of maintaining its obligation to make any such Term Loan, or to increase the cost to such Lender, or to reduce the amount of any sum received or receivable by such Lender then, upon request of such Lender or other Recipient, the Borrower will pay to such Lender or other Recipient, as the case may be, such additional amount or amounts as will compensate such Lender or other Recipient, as the case may be, for such additional costs incurred or reduction suffered.

- (b) <u>Capital Requirements</u>. If any Lender determines in good faith that any Change in Law affecting such Lender or any lending office of such Lender or such Lender's holding company, if any, regarding capital or liquidity requirements, has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of this Agreement, the Term Loan Commitments of such Lender or the portions of the Term Loans made by such Lender to a level below that which such Lender's such Lender's could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to liquidity and capital adequacy), then from time to time the Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender's holding company for any such reduction suffered.
- (c) <u>Certificates for Reimbursement</u>. A certificate of a Lender setting forth the amount or amounts necessary to compensate such Lender or its holding company, as the case may be, as specified in this <u>Section 3.1</u> and delivered to the Borrower shall be conclusive absent manifest error. The Borrower shall pay such Lender the amount due under this <u>Section 3.1</u> and stated on any such certificate within fifteen (15) days after receipt thereof.
- (d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender to demand compensation pursuant to this <u>Section 3.1</u> shall not constitute a waiver of such Lender's right to demand such compensation, <u>provided</u> that the Borrower shall not be required to compensate a Lender pursuant to this <u>Section 3.1</u> for any increased costs incurred or reductions suffered more than six (6) months prior to the date that such Lender, as the case may be, notifies the Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the six (6) month period referred to above shall be extended to include the period of retroactive effect thereof).

3.2 Taxes.

- (a) <u>Defined Terms</u>. For purposes of this <u>Section 3.2</u>, the term "Law" includes FATCA.
- (b) <u>Payments Free of Taxes</u>. Any and all payments by or on account of any obligation of the Borrower hereunder or under any other Loan Document shall be made free and clear of and without deduction or withholding for any Taxes, except as required by applicable Law. If any applicable Law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable Law and, if such Tax is an Indemnified Tax, then the sum payable by the Borrower shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this <u>Section 3.2</u>) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.
- (c) <u>Payment of Other Taxes by the Borrower</u>. The Borrower shall timely pay to the relevant Governmental Authority in accordance with applicable Law, or at the option of the Administrative Agent timely reimburse it for the payment of, any Other Taxes.
- (d) <u>Indemnification by the Borrower</u>. The Borrower shall jointly and severally indemnify each Recipient, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this <u>Section 3.2</u>) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (e) <u>Indemnification by the Lenders</u>. Each Lender shall severally indemnify the Administrative Agent, within ten (10) days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that the Borrower has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Borrower to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of <u>Section 11.7(d)</u> relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or

liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (e).

(f) Evidence of Payments. As soon as practicable after any payment of Taxes by the Borrower to a Governmental Authority pursuant to this Section 3.2, the Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

(g) Status of Lenders .

- (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by applicable Law, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.2(g)(ii)(A), (g)(ii)(B), (g)(ii)(C) and (g)(ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
 - (ii) Without limiting the generality of the foregoing, in the event that the Borrower is a U.S. Borrower:
- (A) any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding Tax;
- (B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies

as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), whichever of the following is applicable:

- (1) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed originals of IRS Form W-8BEN or IRS Form W-8 BEN-E, as applicable (or successor form) establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN or IRS Form W-8 BEN-E, as applicable (or successor form) establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;
 - (2) executed originals of IRS Form W-8ECI (or successor form);
- (3) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881 (c) of the Code, (x) a Tax Compliance Certificate to the effect that such Foreign Lender is not (A) a "bank" within the meaning of Section 881(c)(3)(A) of the Code, (B) a "10 percent shareholder" of the Borrower within the meaning of Section 881(c)(3)(B) of the Code, or (C) a "controlled foreign corporation" described in Section 881 (c)(3)(C) of the Code and (y) executed originals of IRS Form W-8BEN or IRS Form W-8 BEN-E, as applicable (or successor form); or
- (4) to the extent a Foreign Lender is not the beneficial owner, executed originals of IRS Form W-8IMY (or successor form), accompanied by IRS Form W-8ECI, IRS Form W-8BEN, IRS Form W-8BEN-E, a Tax Compliance Certificate, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a Tax Compliance Certificate on behalf of each such direct and indirect partner;
- (C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of any other form prescribed by applicable Law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable Law to permit the

Borrower or the Administrative Agent to determine the withholding or deduction required to be made; and

(D) if a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent such documentation prescribed by applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or the Administrative Agent as may be necessary for the Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of its legal inability to do so.

(h) <u>Treatment of Certain Refunds</u>. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this <u>Section 3.2</u> (including by the payment of additional amounts pursuant to this <u>Section 3.2</u>), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this <u>Section 3.2</u> with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (h) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this paragraph (h), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph (h) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.

3.3 <u>Illegality</u>. If any Lender determines in good faith that any Change in Law has made it unlawful for any Lender to make, maintain or fund LIBOR Rate Loans, or to determine or charge interest rates based upon the LIBOR Rate Option, or if any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the London interbank market, then, on written notice thereof by such Lender to the Borrower through the Administrative Agent, any obligation of such Lender to make or continue LIBOR Rate Loans or to convert Base Rate Loans to LIBOR Rate Loans shall be suspended until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrowers shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all LIBOR Rate Loans of such Lender to Base Rate Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBOR Rate Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBOR Rate Loans. Upon any such prepayment or conversion, the Borrowers shall also pray accrued and unpaid interest and all other amounts payable by Borrower under this Agreement (including amounts payable under Section 3.5) on the amount so prepaid or converted.

3.4 <u>LIBOR Rate Option Unavailable; Interest After Default</u>.

- (a) Adjusted LIBOR Rate Unavailable. If prior to the commencement of any Interest Period for any Borrowing proposed to be subject to the LIBOR Rate Option:
- (i) the Administrative Agent determines in good faith (which determination shall be conclusive and binding absent manifest error) that either Dollar deposits are not being offered to banks in the London interbank LIBOR Rate market or that adequate and reasonable means do not exist for ascertaining the Adjusted LIBOR Rate for such Interest Period; or
- (ii) the Requisite Lenders determine in good faith (which determination shall be conclusive and binding absent manifest error) that the Adjusted LIBOR Rate for such Interest Period will not adequately and fairly reflect the cost to the lenders of making or maintaining the Term Loans for such Interest Period:

then the Administrative Agent shall give notice thereof to the Borrower and the Lenders as promptly as practicable thereafter and, until the Administrative Agent notifies the Borrower and the Lenders that the circumstances giving rise to such notice no longer exist, (x) any request to convert any Base Rate Loan to, or continue any LIBOR Rate Loan at, the LIBOR Rate Option shall be ineffective, and (y) the Base Rate Option shall apply to any and all Borrowings upon the expiration of the Interest Period applicable thereto.

(b) <u>Default Rate</u>. To the extent permitted by Law, immediately upon the occurrence and during the continuation of an Event of Default under clause (g) or (h) of <u>Section 9.1</u>, or immediately after written demand by the Requisite Lenders to the Administrative Agent after the occurrence and during the continuation of an Event of

Default under clause (a) of Section 9.1, then the principal amount of all Obligations shall bear interest at the Default Rate. The Borrower acknowledges that the increase in rates referred to in this Section 3.4(b) reflects, among other things, the fact that such Term Loans or other amounts have become a substantially greater risk given their default status and that the Lenders are entitled to additional compensation for such risk; and all such interest shall be payable by the Borrower upon demand by the Administrative Agent.

- 3.5 <u>Indemnity</u>. Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrower shall promptly compensate such Lender for and hold such Lender harmless from any loss (excluding any Applicable Margin), cost or expense incurred by it as a result of:
 - (a) any continuation, conversion, payment or prepayment of any Term Loan other than a Base Rate Loan on a day other than the last day of the Interest Period for such Term Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);
 - (b) any failure by the Borrower (for a reason, including the revocation of a Loan Request, other than the failure of such Lender to make a Term Loan) to prepay, borrow, continue or convert any Term Loan other than a Base Rate Loan on the date or in the amount notified by the Borrower; or
 - (c) any assignment of a LIBOR Rate Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Borrower pursuant to Section 3.6;

including interest paid but excluding any Applicable Margin on LIBOR Rate Loans and including any loss or expense arising from the cost, liquidation or reemployment of funds obtained by it to maintain such Term Loan or from fees payable to terminate the deposits from which such funds were obtained.

For purposes of calculating amounts payable by the Borrower to the Lenders under this <u>Section 3.5</u>, each Lender shall be deemed to have funded each LIBOR Rate Loan made by it at the LIBOR Rate for such Term Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such LIBOR Rate Loan was in fact so funded.

3.6 Mitigation Obligations; Replacement of Lenders .

(a) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.1</u>, or requires the Borrower to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.2</u>, or suspends making or continuing LIBOR Rate Loans pursuant to <u>Section 3.3</u>, then such Lender shall (at the request of the Borrower) use reasonable efforts to designate a different lending office for funding or booking its Term Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the good faith judgment of such Lender, such designation or

assignment (i) would eliminate or reduce amounts payable pursuant to Section 3.1 or Section 3.2 or eliminate the cause of the suspension pursuant to Section 3.3, as the case may be, in the future, and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.

- (b) Replacement of Lenders. If any Lender requests compensation under Section 3.1, or if the Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.2, or if any Lender suspends making or continuing LIBOR Rate Loans pursuant to Section 3.3 and, in each case, such Lender has declined or is unable to designate a different lending office in accordance with Section 3.6(a) above or if any Lender is a Defaulting Lender or a Non-Consenting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 11.7), all of its interests, rights (other than its existing rights to payments pursuant to Section 3.1 or 3.2) and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided that:
 - (i) the Borrower or applicable assignee shall have paid to the Administrative Agent the assignment fee (if any) specified in Section 11.7;
- (ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Term Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>Section 3.5</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
- (iii) in the case of any such assignment resulting from a claim for compensation under Section 3.1 or payments required to be made pursuant to Section 3.2, such assignment will result in a reduction in such compensation or payments thereafter;
 - (iv) such assignment does not conflict with applicable Law; and
- (v) in the case of any assignment resulting from a Lender becoming a Non-Consenting Lender, the applicable assignee shall have consented to the applicable amendment, waiver or consent.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

3.7 <u>Survival</u>. Each party's obligations under this Article III shall survive the resignation of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Term Loan Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.

IV. CONDITIONS OF LENDING

The effectiveness of this Agreement and the obligation of each Lender to make its Term Loans hereunder are subject to (in addition to, in the case of the Term Loans, the performance by the Borrower of its obligations to be performed hereunder at or prior to the making of any such Term Loans) the satisfaction of the following conditions:

- 4.1 Effectiveness. The effectiveness of this Agreement on the Closing Date is subject to the satisfaction of the following conditions:
- (a) Executed Loan and Other Documents. This Agreement shall have been duly authorized and executed by the Borrower or other Persons, as applicable, in form and substance reasonably satisfactory to the Administrative Agent.
 - (b) Closing Certificates; Opinions .
- (i) Officer's Certificate. The Administrative Agent shall have received a certificate from a Financial Officer of the Borrower, in form and substance reasonably satisfactory to the Administrative Agent, confirming compliance with the conditions precedent set forth in Sections 4.1(g)(i) and (ii).
- (ii) <u>Certificate of Secretary of the Borrower</u>. The Administrative Agent shall have received a certificate of the secretary or assistant secretary of the Borrower certifying that attached thereto is a true and complete copy of the articles of incorporation of the Borrower, and all amendments thereto, certified as of a recent date by the appropriate Governmental Authority in its jurisdiction of incorporation or organization, that attached thereto is a true and complete copy of the bylaws of the Borrower as in effect on the date of such certification; that attached thereto is a true and complete copy of the resolutions of the board of directors of the Borrower, authorizing the borrowings contemplated hereunder, the execution, delivery and performance of this Agreement and the other Loan Documents, as applicable; and as to the incumbency and genuineness of the signature of each officer of the Borrower executing Loan Documents.
- (iii) Certificates of Good Standing . The Administrative Agent shall have received certificates as of a recent date of the good standing of the Borrower under the laws of its jurisdiction of organization.
- (iv) Opinion(s) of Counsel. The Administrative Agent shall have received a favorable opinion of counsel to the Borrower addressed to the Administrative Agent and the Lenders with respect to the Borrower and the Loan Documents, reasonably satisfactory in form and substance to the Administrative Agent.

(c) Consents .

- (i) <u>Governmental and Third Party Approvals</u>. The Borrower shall have delivered to the Administrative Agent all necessary approvals, authorizations and consents, if any, of all Persons, Governmental Authorities, including the FCC and all applicable PUCs, and courts having jurisdiction with respect to the execution and delivery of this Agreement and the other Loan Documents, and all such approvals shall be in form and substance reasonably satisfactory to the Administrative Agent.
- (ii) No Injunction, Etc. No action, proceeding, investigation, regulation or legislation shall have been instituted or threatened in writing before, nor any adverse ruling received from, any Governmental Authority to enjoin, restrain or prohibit, or to obtain substantial damages in respect of, or which is related to or arises out of this Agreement or the other Loan Documents or the consummation of the transactions contemplated hereby or thereby, or which, as determined by the Administrative Agent in its reasonable discretion, would make it inadvisable to consummate the transactions contemplated by this Agreement and such other Loan Documents.
- (d) Fees, Expenses, Etc. There shall have been paid, or shall substantially concurrently with the closing be paid, by the Borrower to the Administrative Agent the fees and other amounts set forth or referenced in Section 2.5 due and payable on or prior to the Closing Date, to the extent invoiced.
- (e) <u>Litigation, Investigations, Audits, Etc.</u> There shall be no action, suit, proceeding or investigation pending against, or, to the knowledge of the Borrower, threatened in writing against or in any other manner relating adversely to, the Borrower or any of its respective properties, in any court or before any arbitrator of any kind or before or by any Governmental Authority (including the FCC), except (i) such as affect the telecommunications industry generally or (ii) as disclosed in the Borrower's Annual Report on Form 10-K for the fiscal year ended December 31, 2013, that would reasonably be expected to have a Material Adverse Effect.
- (f) <u>Know-Your-Customer</u>. At least three (3) Business Days prior to the Closing Date, the Administrative Agent shall have received all documentation and other information in order to comply with requirements of Anti-Terrorism Laws which has been requested by the Administrative Agent of the Borrower at least five (5) Business Days prior to the Closing Date.

(g) Other Conditions .

(i) The representations and warranties contained in <u>Article V</u> of this Agreement shall be (and such request by the Borrower for the Term Loan shall constitute a representation and warranty by the Borrower that such representations and warranties are) true, correct and complete in all material respects on and as of the Closing Date, except that such representations and warranties that are qualified in this Agreement by

reference to materiality or a Material Adverse Effect shall be true and correct in all respects, as of such date (except to the extent such representations and warranties expressly relate to an earlier date, in which case they shall be true, correct and complete in all material respects as of such earlier date).

- (ii) No event shall have occurred and be continuing or would result from the execution and delivery of this Agreement or the consummation of the borrowing contemplated that would constitute an Event of Default or a Default.
- (iii) Since December 31, 2013, there shall not have occurred any event or condition that has had or would reasonably be expected to have a Material Adverse Effect.
- **4.2** <u>Term Loans</u>. The obligation of each Lender to make the Term Loans on the Delayed Draw Date, as applicable, is subject to the satisfaction of the following conditions:
 - (a) Term Loan Commitment Termination Date. The Term Loan Commitment Termination Date shall not have occurred.
 - (b) <u>Deliveries</u>. The Administrative Agent shall have received each of the following in form and substance reasonably satisfactory to the Administrative Agent:
 - (i) a certificate of the Borrower signed by a Financial Officer of the Borrower, dated the Delayed Draw Date stating that (a) all representations and warranties of the Borrower set forth in Article V of this Agreement are true and correct in all material respects, except that such representations and warranties that are qualified in this Agreement by reference to materiality or a Material Adverse Effect shall be true and correct in all respects, as of the Delayed Draw Date (or, if such representation or warranty makes reference to an earlier date, as of such earlier date), (b) no Event of Default or Default exists, (c) all material consents of any Governmental Authority or third party necessary for the Acquisition shall have been obtained and be in full force and effect without condition or provision that would reasonably be expected to have a "Buyer Adverse Condition" (as defined in the Acquisition Agreement as on effect on the Closing Date), (d) no event or condition has occurred since December 16, 2013 that individually or in the aggregate has had, or is reasonably likely to have, a "Seller Material Adverse Effect" (as defined in the Acquisition Agreement in effect on the Closing Date), and (e) attached thereto is a true, correct and complete copy of the Acquisition Agreement, together with all schedules, exhibits, appendices, attachments and amendments thereto, as in effect on the Delayed Draw Date;
 - (ii) a duly executed Compliance Certificate dated the Delayed Draw Date and evidencing a Total Leverage Ratio of no greater than 3.95:1.00 on a proforma basis after giving effect to the Term Loans and the Acquisition;
 - (iii) a certificate dated the Delayed Draw Date and signed by the Secretary or an Assistant Secretary of the Borrower certifying that (a) all information

certified to in the certificate delivered to the Administrative Agent and the Lenders on the Closing Date pursuant to Section 4.1(b)(ii) remains true and correct in all respects as of the Delayed Draw Date, or providing updates to any changes therein, (b) each of the Organizational Documents attached thereto has not been modified since the Closing Date and remains in full force and effect or providing updates to any changes therein and (c) all corporate actions taken by the Borrower in connection with this Agreement and the other Loan Documents remain in full force and effect and authorize the transactions contemplated as of the Delayed Draw Date;

- (iv) a duly completed, executed Loan Request for the Term Loans requested to be made on the Delayed Draw Date, including notice of election as to Interest Periods (if applicable);
 - (v) a duly completed, executed Solvency Certificate signed by a Financial Officer of the Borrower;
- (vi) evidence, in form and substance reasonably satisfactory to the Administrative Agent, that (A) the Acquisition has been or is being consummated on terms and conditions consistent with the Acquisition Agreement in the form attached to the certificate delivered to the Administrative Agent pursuant to Section 4.2(b)(i), in all material respects, substantially concurrently with the making of the Term Loans, other than any amendments, consents or waivers thereto that are not materially adverse to the Lenders (absent the prior consent of the Lenders, which consent shall not be unreasonably withheld or delayed; provided that an increase or reduction in the purchase price by 10% or less shall not be deemed materially adverse to the Lenders), and (B) all material governmental and third-party consents, subordinations or waivers, as applicable, required to consummate the Acquisition have been obtained and are in full force and effect, including any required material permits, authorizations or consents, of all applicable Governmental Authorities, including the FCC and all applicable PUCs, in each case without condition or provision that would reasonably be expected to have a "Buyer Adverse Condition" (as defined) in the Acquisition Agreement in effect on the Closing Date;
- (vii) an executed letter from the Borrower with respect to any proceeds of the Term Loans being disbursed to third parties authorizing the Administrative Agent to distribute such proceeds on behalf of the Borrower in accordance with the instructions set forth in such letter; and
 - (viii) any Term Loan Note requested by any Lender.
- (c) Payment of Fees. The Borrower shall have paid, or shall concurrently pay, all fees and expenses related to the Term Loans and the Loan Documents payable on or before the Delayed Draw Date as required by this Agreement, any Fee Letter or any other Loan Document.

(d) Representations and Warranties; No Defaults. At the time of and immediately after giving effect to the making of any Term Loan on the Delayed Draw Date: (i) the representations and warranties of the Borrower set forth in Article V of this Agreement shall then be true and correct in all material respects, except that such representations and warranties that are qualified in this Agreement by reference to materiality or a Material Adverse Effect shall be true and correct in all respects, as of such date (except for any such representation and warranty that by its terms is made only as of an earlier date, which representation and warranty shall remain true and correct in all material respects as of such earlier date) and (ii) no Event of Default or Default shall have occurred and be continuing or would result from the making of such Term Loans.

V. REPRESENTATIONS AND WARRANTIES

In order to induce the Administrative Agent and the Lenders to enter into this Agreement and to make the Term Loans, the Borrower hereby represents and warrants to the Administrative Agent and each Lender that:

5.1 Organization, Powers, Governmental Approvals.

(a) The Borrower and each Principal Subsidiary (1) is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (2) has all requisite power and authority to own its property and assets and to carry on its business as now conducted and (3) is qualified to do business in every jurisdiction where such qualification is required, except where the failure so to qualify would not have a Material Adverse Effect. The Borrower's execution, delivery and performance of the Loan Documents are within its corporate powers, have been duly authorized by all necessary action and do not violate or create a default under (i) law, (ii) its constituent documents, or (iii) any contractual provision binding upon it, except to the extent (in the case of violations or defaults described under clauses (i) or (iii)) such violation or default would not reasonably be expected to result in a Material Adverse Effect and would not have an adverse effect on the validity, binding effect or enforceability of this Agreement or any other Loan Documents and would not materially adversely affect any of the rights of the Administrative Agent or any Lender under or in connection with this Agreement or any other Loan Documents. Each of the Loan Documents constitutes the legal, valid and binding obligation of the Borrower enforceable against it in accordance with its terms (except as such enforceability may be limited by applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting the rights of creditors generally and general principles of equity, including an implied covenant of good faith and fair dealing).

(b) Except for (1) any Governmental Approvals required in connection with the funding of the Term Loans (such approvals being "Borrowing Approvals") and (2) any Governmental Approvals the failure to obtain which could not reasonably be expected to result in a Material Adverse Effect or affect the validity or enforceability of this Agreement or any other Loan Document, all Governmental Approvals required in connection with the execution and delivery by the Borrower of this Agreement and the

other Loan Documents and the performance by the Borrower of its obligations hereunder and thereunder have been, and, prior to the time of any Borrowing, all Borrowing Approvals will be, duly obtained, are (or, in the case of Borrowing Approvals, will be) in full force and effect without having been amended or modified in any manner that may impair the ability of the Borrower to perform its obligations under this Agreement, and are not (or, in the case of Borrowing Approvals, will not be) the subject of any pending appeal, stay or other challenge.

- 5.2 Financial Statements. The Borrower has furnished or otherwise made available to the Administrative Agent, for itself and its Subsidiaries, its most recent filings with the Securities and Exchange Commission on Forms 10-K and 10-Q. Such Forms 10-K and 10-Q do not contain any untrue statement of a material fact or omit to state a material fact necessary to make any statement therein, in light of the circumstances under which it was made, not misleading. Each of the financial statements in such Forms 10-K and 10-Q has been, and each of the financial statements to be furnished pursuant to Section 8.2 will be, prepared in accordance with GAAP applied consistently with prior periods, except as therein noted, and fairly presents or will fairly present in all material respects the consolidated financial position of the Borrower and its Subsidiaries as of the date thereof and the results of the operations of the Borrower and its Subsidiaries for the period then ended.
- **5.3** No Material Adverse Change. Since the date of the Borrower's most recent financial statements contained in its Annual Report on Form 10-K for the fiscal year ended December 31, 2013, there has been no material adverse change in, and there has occurred no event or condition which is likely to result in a material adverse change in, the financial condition, results of operations, business, assets or operations of the Borrower and the Subsidiaries taken as a whole (it being understood that the consummation of an Asset Exchange shall not constitute such a material adverse change).

5.4 Title to Properties; Possession Under Leases .

- (a) To the best of the Borrower's knowledge, each of the Borrower and the Principal Subsidiaries has good and marketable title to, or valid leasehold interests in, or other rights to use or occupy, all its material properties and assets, except for minor defects in title that do not interfere with its ability to conduct its business as currently conducted or to utilize such properties and assets for their intended purposes. All such material properties and assets are free and clear of Liens, other than Liens expressly permitted by Section 7.1.
- (b) Each of the Borrower and the Principal Subsidiaries has complied with all obligations under all material leases to which it is a party and all such leases are in full force and effect, except where such failure to comply or maintain such leases in full force and effect would not have a Material Adverse Effect. Each of the Borrower and the Subsidiaries enjoys peaceful and undisturbed possession under all such material leases except where such failure would not have a Material Adverse Effect.

5.5 Ownership of Subsidiaries. The Borrower owns, free and clear of any Lien (other than Liens expressly permitted by Section 7.1), all of the issued and outstanding shares of common stock of each of the Principal Subsidiaries.

5.6 <u>Litigation; Compliance with Laws</u>.

- (a) There is no action, suit, or proceeding, or any governmental investigation or any arbitration, in each case pending or, to the knowledge of the Borrower, threatened against the Borrower or any of the Subsidiaries or any material property of any thereof before any court or arbitrator or any governmental or administrative body, agency, or official which (1) challenges the validity of this Agreement or any other Loan Document, (2) may reasonably be expected to have a material adverse effect on the ability of the Borrower to perform any of its obligations under this Agreement or any other Loan Document or on the rights of or benefits available to the Lenders under this Agreement or any other Loan Document or (3) except as disclosed in the Borrower's Annual Report on Form 10-K for the fiscal year ended December 31, 2013, may reasonably be expected to have a Material Adverse Effect.
- (b) Neither the Borrower nor any of the Subsidiaries is in violation of any law, rule, or regulation, or in default with respect to any judgment, writ, injunction or decree of any Governmental Authority, where such violation or default could reasonably be anticipated to result in a Material Adverse Effect.
- (c) Except as set forth in or contemplated by the financial statements or other reports referred to in Section 5.2 and which have been delivered or otherwise made available to the Administrative Agent on or prior to the date hereof, (1) the Borrower and each of its Subsidiaries have complied with all Environmental Laws, except to the extent that failure to so comply is not reasonably likely to have a Material Adverse Effect, (2) neither the Borrower nor any of its Subsidiaries has failed to obtain, maintain or comply with any permit, license or other approval under any Environmental Law, except where such failure is not reasonably likely to have a Material Adverse Effect, (3) neither the Borrower nor any of its Subsidiaries has received notice of any failure to comply with any Environmental Law or become subject to any liability under any Environmental Law, except where such failure or liability is not reasonably likely to have a Material Adverse Effect, (4) no facilities of the Borrower or any of its Subsidiaries are used to manage any Specified Substance in violations of any law, except to the extent that such violations, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect, and (5) the Borrower is aware of no events, conditions or circumstances involving any Release of a Specified Substance that is reasonably likely to have a Material Adverse Effect.

5.7 Agreements

- (a) Neither the Borrower nor any of the Subsidiaries is a party to any agreement or instrument or subject to any corporate restriction that has resulted, or could reasonably be anticipated to result, in a Material Adverse Effect.
- (b) Neither the Borrower nor any of the Subsidiaries is in default in any manner under any provision of any indenture or other agreement or instrument evidencing Indebtedness, or any other material agreement or instrument to which it is a party or by which it or any of its properties or assets are or may be bound, where such default could reasonably be anticipated to result in a Material Adverse Effect.
- 5.8 <u>Federal Reserve Regulations</u>. No part of the proceeds of the Term Loans will be used, whether directly or indirectly, and whether immediately, incidentally, or ultimately, for any purpose which entails a violation of, or which is inconsistent with, the provisions of the Margin Regulations.
- 5.9 <u>Investment Company Act</u>. Neither the Borrower nor any of the Subsidiaries is an "investment company" as defined in, or subject to regulation under, the Investment Company Act of 1940.
 - 5.10 <u>Use of Proceeds</u>. The Borrower will use the proceeds of the Term Loans solely for the purposes described in the recital paragraphs to this Agreement.
- **5.11** Tax Returns. Each of the Borrower and each of the Subsidiaries has filed or caused to be filed all Federal, state and local and non-U.S. tax returns required to have been filed by it and has paid or caused to be paid all taxes (whether or not shown in such tax returns) and satisfied all of its withholding tax obligations, except (A) taxes that are being contested in good faith by appropriate proceedings and for which the Borrower shall have set aside on its books adequate reserves in accordance with GAAP and (B) where such failure to file or pay would not reasonably be expected to result in a Material Adverse Effect.
- 5.12 No Material Misstatements. No statement, information, report, financial statement, exhibit or schedule furnished by or on behalf of the Borrower to the Administrative Agent or any Lender in connection with the syndication or negotiation of this Agreement or any other Loan Document or included herein or therein or delivered pursuant hereto or thereto contained, contains, or will contain any material misstatement of fact or intentionally omitted, omits, or will omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were, are, or will be made, not materially misleading.

5.13 Employee Benefit Plans.

(a) Each Plan is in compliance with ERISA, except for such noncompliance that has not resulted, and could not reasonably be anticipated to result, in a Material Adverse Effect.

- (b) No Plan has an accumulated or waived funding deficiency within the meaning of Section 412 or Section 418B of the IRC and no failure to satisfy the minimum funding standard under Section 412 of the IRC has occurred, whether or not waived, with respect to any Plan, except for any such deficiency or failure that has not resulted, and could not reasonably be anticipated to result, in a Material Adverse Effect.
- (c) No proceedings have been instituted to terminate any Plan, except for such proceedings where the termination of a Plan has not resulted, and could not reasonably be anticipated to result, in a Material Adverse Effect.
- (d) Neither the Borrower nor any Subsidiary or ERISA Affiliate has incurred any liability to or on account of a Plan under ERISA (other than obligations to make contributions in accordance with such Plan), and no condition exists which presents a material risk to the Borrower or any Subsidiary of incurring such a liability, except for such liabilities that have not resulted, and could not reasonably be anticipated to result, in a Material Adverse Effect.
- **5.14** Insurance. Each of the Borrower and the Principal Subsidiaries maintains insurance with financially sound and reputable insurers, or self-insurance, with respect to its properties and business against loss or damage of the kind customarily insured against by reputable companies in the same or similar business and of such types and in such amounts (with such deductible amounts) as is customary for such companies under similar circumstances.
- 5.15 Patriot Act; FCPA. Each of the Borrower and its Subsidiaries is in compliance in all material respects with the USA Patriot Act. The Borrower has implemented and maintains in effect policies and procedures intended to ensure compliance in all material respects by the Borrower, its Subsidiaries and their respective directors, officers and employees (in their capacities as such) with the FCPA and applicable Sanctions, and the Borrower and its Subsidiaries, and to the knowledge of the Borrower or such Subsidiary, its respective officers, employees and directors (in their capacities as such), are in compliance with the FCPA and applicable Sanctions in all material respects. None of the Borrower, any Subsidiary or, to the knowledge of the Borrower or such Subsidiary, any of their respective directors, officers or employees, is a Sanctioned Person. No Term Loan, use of proceeds or other transaction contemplated by the Loan Documents will violate the FCPA or applicable Sanctions.

VI. AFFIRMATIVE COVENANTS

The Borrower hereby covenants and agrees that so long as this Agreement is in effect and until payment in full of all Obligations (other than contingent obligations in respect of which no claim has been made), unless the Requisite Lenders shall otherwise give their written consent, the Borrower shall perform and comply with all covenants in this Article VI.

6.1 Existence; Businesses and Properties .

- (a) Preserve and maintain, cause each of the Principal Subsidiaries to preserve and maintain, and cause each other Subsidiary to preserve and maintain (where the failure by any such other Subsidiary to so preserve and maintain would likely result in a Material Adverse Effect), its corporate existence, rights and franchises, except in connection with an Asset Exchange, provided, however, that the corporate existence of any Principal Subsidiary may be terminated if such termination is not disadvantageous to the Administrative Agent or any Lender;
 - (b) continue to own all of the outstanding shares of common stock of each Principal Subsidiary, except in connection with an Asset Exchange;
- (c) comply, and cause each of the Subsidiaries to comply, in all material respects, with all applicable laws, rules, regulations and orders, including all Environmental Laws;
- (d) pay, and cause each of the Subsidiaries to pay, before any such amounts become delinquent, (i) all Taxes imposed upon it or upon its property, and (ii) all claims (including claims for labor, materials, supplies, or services) which might, if unpaid, become a Lien upon its property, unless, in each case, the validity or amount thereof is being disputed in good faith, and the Borrower has maintained adequate reserves with respect thereto, in each case where the failure to so pay would be reasonably expected to cause a Material Adverse Effect;
- (e) keep, and cause each of the Subsidiaries to keep, proper books of record and account, containing complete and accurate entries of all financial and business transactions of the Borrower and such Subsidiary in all material respects;
- (f) continue to carry on, and cause each Principal Subsidiary to continue to carry on, substantially the same type of business as the Borrower or such Principal Subsidiary conducted as of the date hereof and business reasonably related thereto, except for changes in such business that result from an Asset Exchange; and
- (g) maintain or cause to be maintained insurance with financially sound and reputable insurers, or self-insurance, with respect to its properties and business and the properties and business of the Subsidiaries against loss or damage of the kinds customarily insured against by reputable companies in the same or similar businesses, such insurance to be of such types and in such amounts (with such deductible amounts) as is customary for such companies under similar circumstances;
- $\underline{provided}$, $\underline{however}$, that the foregoing shall not limit the right of the Borrower or any of its Subsidiaries to engage in any transaction not otherwise prohibited by $\underline{Section 7.2}$, $\underline{7.3}$ or $\underline{7.4}$.
- **6.2** <u>M aintaining Records</u>. Maintain all financial records in accordance with GAAP and, upon reasonable notice, permit the Administrative Agent and each Lender to visit and

inspect the financial records of the Borrower at reasonable times and as often as reasonably requested and to make extracts from and copies of such financial records, and permit any representatives designated by the Administrative Agent or any Lender to discuss the affairs, finances and condition of the Borrower with the appropriate officers thereof and, with the Borrower's consent (which shall not be unreasonably withheld), the independent accountants therefor; provided, however, that if the Borrower shall so require, a single representative shall be appointed by the Requisite Lenders to exercise the rights granted to the Lenders under this Section 6.2; provided, further, that when an Event of Default exists the Administrative Agent or any Lender may do any of the foregoing, upon reasonable notice, at any time during normal business hours (without appointment of a single representative by the Lenders).

- 6.3 <u>Use of Proceeds</u>. The Borrower will use the proceeds of the Term Loans solely for the purposes described in the recital paragraphs to this Agreement.
- 6.4 <u>CoBank Equity</u>. So long as CoBank is a Lender hereunder, the Borrower will acquire equity in CoBank in such amounts and at such times as CoBank may require in accordance with CoBank's Bylaws and Capital Plan (as each may be amended from time to time), except that the maximum amount of equity that the Borrower may be required to purchase in CoBank in connection with the Term Loans made by CoBank hereunder may not exceed the maximum amount permitted by CoBank's Bylaws and Capital Plan at the time this Agreement is entered into. The Borrower acknowledges receipt of a copy of (i) CoBank's most recent annual report, and if more recent, CoBank's latest quarterly report, (ii) CoBank's Notice to Prospective Stockholders and (iii) CoBank's Bylaws and Capital Plan, which describe the nature of all of the Borrower's stock and other equities in CoBank acquired in connection with its patronage loan from CoBank (the "CoBank Equities") as well as capitalization requirements, and agrees to be bound by the terms thereof.
 - (a) Each party hereto acknowledges that CoBank's Bylaws and Capital Plan (as each may be amended from time to time) shall govern (x) the rights and obligations of the parties with respect to the CoBank Equities and any patronage refunds or other distributions made on account thereof or on account of the Borrower's patronage with CoBank, (y) the Borrower's eligibility for patronage distributions from CoBank (in the form of CoBank Equities and cash) and (z) patronage distributions, if any, in the event of a sale of a participation interest. CoBank reserves the right to assign or sell participations in all or any part of its Term Loans or Term Loan Commitments on a non-patronage basis.
 - (b) Each party hereto acknowledges that CoBank has a statutory first lien pursuant to the Farm Credit Act of 1971 (as amended from time to time) on all CoBank Equities that the Borrower may now own or hereafter acquire, which statutory lien shall be for CoBank's sole and exclusive benefit. The CoBank Equities shall not constitute security for the Obligations due to any other Lender. To the extent that any of the Loan Documents create a Lien on the CoBank Equities or on patronage accrued by CoBank for the account of the Borrower (including, in each case, proceeds thereof), such Lien shall be for CoBank's sole and exclusive benefit and shall not be subject to pro rata sharing hereunder. Neither the CoBank Equities nor any accrued patronage shall be offset

against the Obligations except that, in the event of an Event of Default, CoBank may elect to apply the cash portion of any patronage distribution or retirement of equity to amounts due under this Agreement. The Borrower acknowledges that any corresponding tax liability associated with such application is the sole responsibility of the Borrower. CoBank shall have no obligation to retire the CoBank Equities upon any Event of Default, Default or any other default by the Borrower or at any other time, either for application to the Obligations or otherwise.

VII. NEGATIVE COVENANTS

The Borrower hereby covenants and agrees that so long as this Agreement is in effect and until payment in full of all Obligations (other than contingent obligations in respect of which no claim has been made), unless the Requisite Lenders shall otherwise give their written consent, the Borrower shall perform and comply with all covenants in this Article VII, and the Borrower will not:

7.1 Liens; Restrictions on Sales of Receivables. Create, incur, assume, or suffer to exist, or permit any of the Subsidiaries to create, incur, assume, or suffer to exist, any Lien on any of its property now owned or hereafter acquired to secure any Indebtedness of the Borrower or any such Subsidiary, or sell or assign any accounts receivable (other than in the ordinary course of business substantially in accordance with the Borrower's past practice), other than: (A) Liens incurred or deposits made in the ordinary course of business to secure surety and appeal bonds, leases, return-of-money bonds and other similar obligations (exclusive of obligations of the payment of borrowed money); (B) pledges or deposits to secure the utility obligations of the Borrower incurred in the ordinary course of business; (C) Liens upon or in property now owned or hereafter acquired to secure Indebtedness incurred (1) solely for the purpose of financing the acquisition, construction or improvement of such property, provided that such Indebtedness shall not exceed the fair market value of the property being acquired, constructed or improved or (2) to refinance, refund, renew or extend any Indebtedness described in subclause (1) that does not increase the principal amount thereof except by the amount of accrued and unpaid interest and premium thereon and reasonable fees and expense in connection with such refinancing, refunding, renewal or extension so long as the Liens securing such Indebtedness shall be limited to all or part of the same property that secured the Indebtedness refinanced, refunded, renewed or extended; (D) Liens on the assets of any Person merged or consolidated with or into (in accordance with Section 7.4) the Borrower or any Principal Subsidiary that were in effect at the time of such merger or consolidation; (E) Liens for taxes, assessments and governmental charges or levies, which are not yet due or which are being contested in good faith by appropriate proceedings; (F) Liens securing Indebtedness of the Borrower or any Subsidiary to the Rural Electrification Administration or the Rural Utilities Service (or any successor to any such agency) in an aggregate principal amount outstanding at any time not to exceed \$25,000,000; (G) carriers', warehousemen's, mechanics', materialmen's, repairmen's, supplier's or other like Liens arising in the ordinary course of business relating to obligations not overdue for a period of more than 60 days or which are bonded or being contested in good faith by appropriate proceedings; (H) pledges or deposits in connection with workers' compensation laws or similar legislation or to secure public or statutory obligations; (I) Liens or deposits to secure the performance of bids, trade contracts, leases, statutory obligations, surety and performance bonds and other obligations of a like nature incurred in the ordinary course of business; (J) easements, rights of way, restrictions and other encumbrances incurred which, in the aggregate, do not materially interfere with the ordinary conduct of business; (K) restrictions by Governmental Authorities on the operations, business or assets of the Borrower or its Subsidiaries that are customary in the Borrower's and its Subsidiaries' businesses; (L) sales of accounts receivable pursuant to, and Liens existing or deemed to exist in connection with, any Securitization Transactions, provided that the aggregate amount of all such Securitization Transactions shall not at any time exceed \$150,000,000; and (M) other Liens securing Indebtedness in an aggregate principal amount, when aggregated, without duplication, with the amount of Indebtedness of Subsidiaries outstanding pursuant to Section 7.7(3), not to exceed \$200,000,000 at any one time outstanding; provided. however, that the Borrower or any Subsidiary may create, incur, assume or suffer to exist other Liens (in addition to Liens excepted by the foregoing clauses (A) through (M)) on its assets so long as (1) such Liens equally and ratably secure the Obligations pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent and (2) at the time of any incurrence of Indebtedness secured by Liens in reliance on this proviso, the aggregate principal amount of all such Indebtedness incurred in reliance on this proviso (including the Obligations), when aggregated, without duplication, with the amount of Indebtedness of Subsidiaries outstanding pursuant to Section 7.7 (other than clauses (1) through (4) of Section 7.7), shall not exceed the Maximum Priority Amount at such time.

- 7.2 Ownership of the Principal Subsidiaries. Sell, assign, pledge, or otherwise transfer or dispose of any shares of common stock, voting stock, or stock convertible into voting or common stock of any Principal Subsidiary, except (A) to another Subsidiary, or (B) in connection with an Asset Exchange; provided, however, that the Borrower may pledge any shares of common stock, voting stock, or stock convertible into voting or common stock of any Principal Subsidiary so long as such pledge equally and ratably secures the Obligations pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent.
- 7.3 Asset Sales. Except in connection with an Asset Exchange, permit any Principal Subsidiary to sell, assign, or otherwise dispose of telecommunications assets (whether in one transaction or a series of transactions), if the net, after-tax proceeds thereof are used by the Borrower or any Subsidiary to prepay (other than a mandatory prepayment in accordance with the terms of the applicable governing documents, including pursuant to any put provision) Indebtedness incurred after the date hereof which Indebtedness has a maturity later than the Maturity Date (other than bridge or other financings incurred in connection with an asset purchase or sale, including acquisition indebtedness or indebtedness of an acquired entity or indebtedness incurred to refinance indebtedness outstanding as of the date hereof).
- 7.4 Mergers. Merge or consolidate with, or sell, assign, lease, or otherwise dispose of (whether in one transaction or a series of transactions) all or substantially all of its assets (whether now owned or hereafter acquired); except in connection with an Asset Exchange, to any Person, or permit any Principal Subsidiary to do so, except that any Subsidiary may merge into or, subject to Section 7.3, transfer assets to the Borrower or any other Subsidiary and the Borrower may merge with any Person; provided that, immediately thereafter and after giving effect thereto, no event shall occur or be continuing which constitutes an Event of Default or a

Default and, in the case of any such merger to which the Borrower is a party, either the Borrower is the surviving corporation or the surviving entity (if not the Borrower) has a consolidated net worth (as determined in accordance with GAAP) immediately subsequent to such merger at least equal to the Consolidated Net Worth of the Borrower immediately prior to such merger and expressly assumes the obligations of the Borrower hereunder; provided, however, that, notwithstanding the foregoing, the Borrower and any of the Principal Subsidiaries may sell assets in the ordinary course of its business and may sell or otherwise dispose of worn out or obsolete equipment on a basis consistent with good business practices.

7.5 Restrictions on Dividends

- (a) Enter into or permit any Principal Subsidiary to enter into, any contract or agreement (other than with a governmental regulatory authority having jurisdiction over the Borrower or such Principal Subsidiary) restricting the ability of such Principal Subsidiary to pay dividends or make distributions to the Borrower in any manner that would impair the ability of the Borrower to meet its present and future obligations hereunder.
- (b) In the case of the Borrower only, declare or make, directly or indirectly, any Restricted Payment, or incur any obligation (contingent or otherwise) to do so, in each case if any Event of Default has occurred and is continuing at the time of such action or will result therefrom (but excluding the payment of dividends declared and announced by the Board of Directors at a time when no Event of Default existed).
- 7.6 Transactions with Affiliates. Except in connection with an Asset Exchange, sell or transfer any property or assets to, or purchase or acquire any property or assets from, or otherwise engage in any other transactions with, any of its Affiliates (or permit any of its Subsidiaries to do any of the foregoing), except that as long as no Default or Event of Default shall have occurred and be continuing, the Borrower or any Subsidiary may engage in any of the foregoing transactions (A) in the ordinary course of business at prices and on terms and conditions not less favorable to the Borrower or such Subsidiary than could be obtained on an arm's-length basis from unrelated third parties, (B) as otherwise may be required by any Federal or state Governmental Authority, (C) so long as such transactions are not materially disadvantageous to the Borrower or (D) so long as such transactions are solely among the Borrower and one or more of its Subsidiaries.
- 7.7 Subsidiary Indebtedness. Permit any Subsidiary to enter into, directly or indirectly, issue, incur, assume or Guarantee any Indebtedness unless (A) the Obligations are guaranteed by such Subsidiary on a pari passu basis pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent and (B) at the time of any incurrence of such Indebtedness, the aggregate principal amount of such Indebtedness of Subsidiaries (including any Guarantee of the Obligations but excluding Indebtedness permitted by clauses (1) through (4) below), when aggregated with the principal amount of Indebtedness secured by Liens in reliance on the final proviso to Section 7.1., shall not exceed the Maximum Priority Amount at such time, except (1) Indebtedness in effect at the time such Subsidiary becomes a Subsidiary of the Borrower, so long as such Indebtedness was not entered into solely

in contemplation of such Person becoming a Subsidiary of the Borrower (and any refinancing, refunding, renewal or extension of such Indebtedness that does not increase the principal amount thereof except by the amount of accrued and unpaid interest and premium thereon and reasonable fees and expenses in connection with such refinancing, refunding, renewal or extension), (2) any Indebtedness in effect as of the Closing Date that is listed on Schedule 7.7 (and any refinancing, refunding, renewal or extension of such Indebtedness to the extent not increasing the principal amount thereof except by the amount of accrued and unpaid interest and premium thereon and reasonable fees and expenses in connection with such refinancing, refunding, renewal or extension), (3) additional Indebtedness, when aggregated, without duplication, with the principal amount of Indebtedness secured by Liens in reliance on Section 7.1(M), not to exceed \$200,000,000 at any one time outstanding and (4) Indebtedness of a Subsidiary to the Borrower or another Subsidiary.

7.8 Anti-Terrorism Laws. The Borrower covenants and agrees that it shall not, and shall not permit any of its Subsidiaries to, knowingly, directly or indirectly, (a) conduct any business or engage in making or receiving any contribution of funds, goods or services to or for the benefit of any Person subject to Executive Order No. 13224, (b) deal in, or otherwise engage in any material transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224 or any other Anti-Terrorism Law, or (c) engage in any transaction that violates any of the prohibitions set forth in any Anti-Terrorism Law.

VIII. FINANCIAL COVENANTS AND REPORTING

The Borrower hereby covenants and agrees that so long as this Agreement is in effect and until payment in full of all Obligations (other than contingent obligations in respect of which no claim has been made), unless the Requisite Lenders shall otherwise give their written consent, the Borrower shall perform and comply with all covenants in this Article VIII. For the purposes of this Article VIII, all covenants calculated for the Borrower shall be calculated on a consolidated basis for the Borrower and its Subsidiaries.

- **8.1** <u>Total Leverage Ratio</u>. The Borrower shall maintain at all times, measured at each fiscal quarter end, a Total Leverage Ratio less than or equal to 4.5:1.0; <u>provided</u>, that if after the Closing Date the Borrower provides any other holder of Indebtedness with additional or more restrictive financial covenants than set forth in this <u>Section 8.1</u>, then this Agreement shall be deemed to include, and this Agreement shall be amended to contain, such additional and more restrictive financial covenants
- **8.2** Financial Statements and Other Reports. The Borrower will and will cause its Subsidiaries to maintain a system of accounting established and administered in accordance with sound business practices to permit preparation of financial statements in conformity with GAAP consistently applied (it being understood that quarterly financial statements are not required to have footnote disclosures and are subject to normal year-end audit adjustments). The Borrower will deliver each of the financial statements and other reports described below to the Administrative Agent.

- (a) Quarterly Financials. Within 65 days after the end of each of the first three fiscal quarters of each fiscal year of the Borrower, consolidated balance sheets and related statements of income and cash flows of the Borrower and its Subsidiaries as of the close of such fiscal quarter and the then elapsed portion of the fiscal year (which requirement shall be deemed satisfied by the delivery of the Borrower's Quarterly Report on Form 10-Q (or any successor form) for such quarter), each certified by a Financial Officer as fairly presenting in all material respects the financial condition and results of operations of the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP consistently applied, subject to the absence of footnotes and normal year-end audit adjustments;
- (b) Year-End Financials. As soon as available and in any event within 110 days after the end of each fiscal year of the Borrower, consolidated balance sheets and the related statements of income and cash flows of the Borrower and its Subsidiaries as of the close of such fiscal year (which requirement shall be deemed satisfied by the delivery of the Borrower's Annual Report on Form 10-K (or any successor form) for such year), all audited by KPMG LLP or other independent public accountants of recognized national standing, certified without any material qualification or exception as to the scope of such audit or any "going concern" or like qualification by such accountants, and accompanied by an opinion of such accountants to the effect that such consolidated financial statements fairly present in all material respects the financial condition and results of operations of the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP consistently applied;
- (c) The Borrower Compliance Certificate. Together with each delivery of financial statements of the Borrower and its Subsidiaries pursuant to Sections 8.2(a) or 8.2(b), the Borrower will deliver a Compliance Certificate of the Borrower (1) certifying as to whether a Default has occurred and, if a Default has occurred, specifying the details thereof and any action taken or proposed to be taken with respect thereto and (2) setting forth reasonably detailed calculations (including with respect to any pro forma effect given to a Material Transaction) demonstrating compliance with Section 8.1 as of the last day of the most recent fiscal quarter covered by such financial statements.
- (d) <u>Budget.</u> As soon as available and in any event within 65 days after the beginning of the Borrower's fiscal years, the Borrower will deliver a Budget of the Borrower and its consolidated Subsidiaries for such fiscal year.
- (e) <u>SEC Filings.</u> Promptly upon the mailing or filing thereof, copies of all financial statements, reports and proxy statements mailed to the Borrower's public shareholders, and copies of all registration statements (other than those on Form S-8) and Form 8-K's (to the extent that such Form 8-K's disclose actual or potential adverse developments with respect to the Borrower or any of its Subsidiaries that constitute, or could reasonably be anticipated to constitute, a Material Adverse Effect) filed with the Securities and Exchange Commission (the "<u>SEC</u>") (or any successor thereto) or any national securities exchange.

- (f) <u>Litigation and Other Notices.</u> Promptly upon any Financial Officer of the Borrower becoming aware of the following, Borrower will provide written notice thereof to the Administrative Agent:
- (i) any Event of Default or Default, specifying the nature and extent thereof and the corrective action (if any) proposed to be taken with respect thereto:
- (ii) the filing or commencement of, or any written notice of intention of any Person to file or commence, any action, suit or proceeding, whether at law or in equity or by or before any Governmental Authority, against the Borrower or any of the Subsidiaries which is reasonably likely to be adversely determined and which, if adversely determined, could reasonably be anticipated to result in a Material Adverse Effect; and
- (iii) any development with respect to the Borrower or any Subsidiary that has resulted in, or could reasonably be anticipated to result in, a Material Adverse Effect.
- (g) ERISA Events. Promptly after (1) the occurrence thereof, notice of any ERISA Termination Event or "prohibited transaction", as such term is defined in Section 4975 of the IRC, with respect to any Plan that results, or could reasonably be anticipated to result, in a Material Adverse Effect, which notice shall specify the nature thereof and the Borrower's proposed response thereto, and (2) actual knowledge thereof, copies of any notice of PBGC's intention to terminate or to have a trustee appointed to administer any Plan.
- (h) Other Information. Promptly, from time to time, such other information, regarding the Borrower's operations, business affairs and financial condition, or compliance with the terms of this Agreement, as the Administrative Agent or any Lender (through the Administrative Agent) may reasonably request.
- (i) <u>Delivery of Information.</u> Documents required to be delivered pursuant to <u>Section 8.2(a)</u>, (b) or (c) (to the extent any such documents are included in materials otherwise filed with the SEC (or any successor thereto)) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date (1) on which the Borrower posts such documents, or provides a link thereto at www.frontier.com; or (2) on which such documents are posted on the Borrower's behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether sponsored by the Administrative Agent); <u>provided</u> that: (i) the Borrower shall deliver paper copies of such documents to the Administrative Agent or any Lender that requests the Borrower to deliver such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender and (ii) the Borrower shall notify the Administrative Agent (by telecopier or electronic mail) of the posting of any such documents and provide to the Administrative Agent by electronic mail electronic versions (i.e. , soft copies) of such documents. Notwithstanding anything contained

herein, in every instance the Borrower shall be required to provide paper copies of the Compliance Certificates required by <u>Section 8.2(c)</u> to the Administrative Agent. Except for Compliance Certificates, the Administrative Agent shall have no obligation to request the delivery or to maintain copies of the documents referred to above, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery, and each Lender shall be solely responsible for requesting delivery to it or maintaining its copies of such documents.

IX. EVENTS OF DEFAULT

- 9.1 Events of Default. An Event of Default means the occurrence or existence of any one or more of the following:
- (a) <u>Payment.</u> The Borrower shall fail to pay (1) any principal of the Term Loans when and as the same shall become due and payable, whether at the due date thereof or at a date fixed for prepayment thereof or otherwise, or (2) any interest on the Term Loans or any fee or any other amount (other than an amount referred to in clause (1) of this <u>Section 9.1(a)</u>) payable under this Agreement or under any other Loan Document, when and as the same shall become due and payable, and such failure shall continue unremedied for a period of five Business Days; or
- (b) <u>Default in Other Agreements.</u> Any breach by the Borrower or any of its Principal Subsidiaries of any agreement or instrument relating to Indebtedness occurs that results in any Indebtedness of any one or more of the Borrower and its Principal Subsidiaries in an aggregate principal amount exceeding \$50,000,000 becoming due prior to its scheduled maturity or that enables or permits the holder or holders of any such Indebtedness or any trustee or agent on its or their behalf to cause any such Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity, in each case after giving effect to any applicable grace period; or, as a result of any such breach, any such Indebtedness shall be required to be prepaid (other than by a regularly scheduled required prepayment, pursuant to any put right (or similar right) of the holder thereof, or by the exercise by the Borrower or any Principal Subsidiary of its right to make a voluntary prepayment) in whole or in part prior to its stated maturity; or there occurs under any Swap Contract an Early Termination Date (as defined in such Swap Contract) resulting from (1) any event of default under such Swap Contract as to which the Borrower or any Principal Subsidiary is the Defaulting Party (as defined in such Swap Contract) or (2) any Termination Event (as defined in such Swap Contract) under such Swap Contract as to which the Borrower or any Principal Subsidiary is an Affected Party (as defined in such Swap Contract) and, in either event, the Swap Termination Value owed by the Borrower or such Subsidiary as a result thereof is greater than \$50,000,000; provided that this Section 9.1(b) shall not apply to any Indebtedness that becomes due as a result of a voluntary redemption or repayment of such Indebtedness effected in accordance with the terms of the agreement governing such Indebtedness and which is not prohibited by this Agreement; or

- (c) Breach of Warranty. A ny representation or warranty made or deemed made by or on behalf of the Borrower or any of its Subsidiaries in or in connection with this Agreement or any other Loan Document or any amendment or modification hereof or thereof, or any waiver hereunder or thereunder, or in any report, certificate, financial statement or other document furnished pursuant to or in connection with this Agreement or any other Loan Document or any amendment or modification hereof or thereof, or any waiver hereunder or thereunder, shall prove to have been incorrect when made or deemed made in any material respect: or
- (d) Breach of Certain Provisions. Failure of the Borrower or any of its Subsidiaries to perform or comply with any term or condition contained in Sections $\underline{6.1(f)}$, $\underline{6.3}$ or $\underline{8.1}$, or in Article VII; or
- (e) Other Defaults Under Loan Documents. The Borrower shall fail to observe or perform any covenant, condition or agreement contained in this Agreement (other than those specified in Sections 9.1(a) or 9.1(d)) or any other Loan Document and such failure shall continue unremedied for a period of 30 days after the earlier to occur of (1) the Borrower obtaining knowledge thereof and (2) the date that notice thereof shall have been given to the Borrower by the Administrative Agent or any Lender; or
- (f) <u>Default in Other Indebtedness</u>. The Borrower or any Principal Subsidiary shall fail to make any payment of any amount in respect of Indebtedness in an aggregate principal amount of \$50,000,000 or more, when and as the same shall become due and payable after giving effect to any applicable grace periods; or
- (g) Involuntary Bankruptcy; Appointment of Receiver; Etc. An involuntary proceeding shall be commenced or an involuntary petition shall be filed in a court of competent jurisdiction seeking (1) liquidation, reorganization or other relief in respect of the Borrower or any of its Principal Subsidiaries or its debts, or of a substantial part of its assets, under any Federal or state bankruptcy, insolvency, receivership or similar law now or hereafter in effect or (2) the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for the Borrower or any of its Principal Subsidiaries or for a substantial part of its assets, and, in any such case, such proceeding or petition shall continue undismissed for a period of 60 or more days or an order or decree approving or ordering any of the foregoing shall be entered; or
- (h) Voluntary Bankruptcy; Appointment of Receiver; Etc. The Borrower or any of its Principal Subsidiaries shall (1) voluntarily commence any proceeding or file any petition seeking liquidation, reorganization or other relief under any Federal or state bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (2) consent to the institution of, or fail to contest in a timely and appropriate manner, any proceeding or petition described in Section 9.1(g), (3) apply for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for the Borrower or any of its Principal Subsidiaries or for a substantial part of its assets, (4) file an answer admitting the material allegations of a petition filed against it in any such proceeding,

- (5) make a general assignment for the benefit of creditors or (6) take any action for the purpose of effecting any of the foregoing; or
- (i) <u>Judgment and Attachments</u>. One or more judgments for the payment of money in an aggregate amount in excess of \$50,000,000 shall be rendered against the Borrower or any of its Subsidiaries or any combination thereof and the same shall remain undischarged for a period of 30 consecutive days during which execution shall not be effectively stayed, or any action shall be legally taken by a judgment creditor to attach or levy upon any assets of the Borrower or any of its Subsidiaries to enforce any such judgment; or
- (j) ERISA; Pension Plans. A Plan shall fail to maintain the minimum funding standard required by Section 412(a) of the IRC for any plan year or a waiver of such standard is sought or granted under Section 412(c), or a Plan is or shall have been terminated or the subject of termination proceedings under ERISA, or the Borrower or an ERISA Affiliate has incurred a liability to or on account of a Plan under Section 4062, 4063, 4064, 4201 or 4204 of ERISA, and there shall result from any such event or events a Material Adverse Effect; or
 - (k) Change in Control . A Change in Control shall occur.

9.2 Consequences of Event of Default.

- (a) Events of Default Other Than Bankruptcy, Insolvency or Reorganization Proceedings. If an Event of Default specified under Section 9.1 (other than Section 9.1(g) or (h)) shall occur and be continuing, the Lenders and the Administrative Agent shall be under no further obligation to make Term Loans and the Administrative Agent may, and upon the request of the Requisite Lenders, shall by written notice to the Borrower, declare the unpaid principal amount of the Term Loans then outstanding and all interest accrued thereon, any unpaid fees and all other Indebtedness of the Borrower to the Lenders hereunder and thereunder to be forthwith due and payable, and the same shall thereupon become and be immediately due and payable to the Administrative Agent for the benefit of each Lender without presentment, demand, protest or any other notice of any kind, all of which are hereby expressly waived; and
- (b) <u>Bankruptcy, Insolvency or Reorganization Proceedings</u>. If an Event of Default specified under <u>Section 9.1(g)</u> or <u>(h)</u> shall occur, the Lenders shall be under no further obligations to make Term Loans hereunder and the unpaid principal amount of the Term Loans then outstanding and all interest accrued thereon, any unpaid fees and all other Indebtedness of the Borrower to the Lenders hereunder and thereunder automatically shall be immediately due and payable, without presentment, demand, protest or notice of any kind, all of which are hereby expressly waived; and
- (c) <u>Set-off</u>. If an Event of Default shall have occurred and be continuing, each Lender and each of their respective Affiliates is hereby authorized at any time and from time to time, to the fullest extent permitted by applicable law, to set off and apply any

and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held, and other obligations (in whatever currency) at any time owing, by such Lender or any such Affiliate, to or for the credit or the account of the Borrower against any and all of the obligations of the Borrower now or hereafter existing under this Agreement or any other Loan Document to such Lender or its respective Affiliates, irrespective of whether or not such Lender or Affiliate shall have made any demand under this Agreement or any other Loan Document and although such obligations of the Borrower may be contingent or unmatured or are owed to a branch, office or Affiliate of such Lender different from the branch, office or Affiliate holding such deposit or obligated on such indebtedness; provided that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.12 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff. The rights of each Lender and its respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff) that such Lender or its respective Affiliates may have. Each Lender agrees to notify the Borrower and the Administrative Agent promptly after any such setoff and application; provided that the failure to give such notice shall not affect the validity of such setoff and application.

(d) <u>Application of Proceeds</u>. After the exercise of remedies provided for in this <u>Section 9.2</u> (or after the Term Loans have automatically become immediately due and payable), any amounts received on account of the Obligations shall be applied by the Administrative Agent in the following order:

First, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts (including fees, charges and disbursements of counsel to the Administrative Agent) payable to the Administrative Agent in its capacity as such;

Second, to payment of that portion of the Obligations constituting indemnities, expenses, and other amounts (other than principal, interest and fees) payable to the Lenders (including fees, charges and disbursements of counsel to the respective Lenders and amounts payable under $\underline{\text{Article } X}$), ratably among them in proportion to the amounts described in this clause Second payable to them;

Third, to payment of that portion of the Obligations constituting unpaid principal of the Term Loans, ratably among the Lenders in proportion to the respective amounts described in this clause Third held by them;

Fourth, to payment of all other Obligations, ratably among the Lenders in proportion to the respective amounts described in this clause Fourth held by them;

X. THE ADMINISTRATIVE AGENT

- **Appointment and Authority**. Each of the Lenders (on behalf of itself and each of its Affiliates) hereby irrevocably appoints CoBank to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article X are solely for the benefit of the Administrative Agent and the Lenders, and the Borrower shall not have rights as a third party beneficiary of any of such provisions. It is understood and agreed that the use of the term "agent" herein or in any other Loan Documents (or any other similar term) with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties.
- 10.2 Rights as a Lender. The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lenders" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, own securities of, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.
- 10.3 No Fiduciary Duty. The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents and its duties hereunder shall be administrative in nature. Without limiting the generality of the foregoing, the Administrative Agent:
 - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Event of Default has occurred and is continuing;
 - (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Requisite Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents); provided that the Administrative Agent shall not be required to take any action that, in its good faith opinion or the good faith opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable Law, including for the avoidance of doubt any action that may be in violation

of the automatic stay under any Debtor Relief Law or that may effect a forfeiture, modification or termination of property of a Defaulting Lender in violation of any Debtor Relief Law: and

- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.
- 10.4 Exculpation. The Administrative Agent shall not be liable to the Lenders for any action taken or not taken by it (i) with the consent or at the request of the Requisite Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 11.1 and 9.2) or (ii) in the absence of its own gross negligence or willful misconduct as determined by a court of competent jurisdiction by final and non-appealable judgment. The Administrative Agent shall be deemed not to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Administrative Agent by the Borrower or a Lender.
 - (a) The Administrative Agent shall not be responsible to the Lenders for or have any duty to the Lenders to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition set forth in Article IV or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.
- 10.5 Reliance by the Administrative Agent. The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Term Loan that by its terms must be fulfilled to the satisfaction of a Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent shall have received notice to the contrary from such Lender prior to the making of such Term Loan. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable

for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

- 10.6 Delegation of Duties. The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub –agents appointed by the Administrative Agent. The Administrative Agent and any such sub –agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article X shall apply to any such sub –agent and to the Related Parties of the Administrative Agent and any such sub –agent, and shall apply to their respective activities in connection with the syndication of the Term Loan Facility as well as activities as Administrative Agent. The Administrative Agent shall not be responsible for the negligence or misconduct of any sub-agents except to the extent that a court of competent jurisdiction determines in a final and non-appealable judgment that the Administrative Agent acted with gross negligence or willful misconduct in the selection of such sub agents.
- 10.7 Filing Proofs of Claim. In case of the pendency of any proceedings under any Debtor Relief Law or any other judicial proceeding relating to the Borrower, the Administrative Agent (irrespective of whether the principal of any Term Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand therefor) shall be entitled and empowered (but not obligated) by intervention in such proceeding or otherwise:
 - (a) to file and prove a claim for the whole amount of the owing and unpaid principal and interest in respect to the Obligations and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and the Administrative Agent under Sections 2.7 and 3.5) allowed in such proceeding;
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same; and
 - (c) any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Lenders, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.7, 2.9(b) and 3.5.
- 10.8 Resignation or Removal of the Administrative Agent. The Administrative Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Requisite Lenders shall have the right, in

consultation with the Borrower, to appoint a successor Administrative Agent that is reasonably acceptable to the Borrower. If no such successor shall have been so appointed by the Requisite Lenders and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation (or such earlier date as the Requisite Lenders may approve)(the "Resignation Effective Date"), then the retiring Administrative Agent may (but shall not be obligated to) on behalf of the Lenders, appoint a successor Administrative Agent meeting the qualifications set forth above; provided, that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then the Administrative Agent's resignation shall nonetheless become effective in accordance with such notice on the Resignation Effective Date. If the Person serving as Administrative Agent is a Defaulting Lender pursuant to clause (iv) of the definition thereof, the Requisite Lenders may, to the extent permitted by applicable law, by notice in writing to the Borrower and such Person remove such Person as Administrative Agent and, in consultation with the Borrower, appoint a successor reasonably acceptable to the Borrower. If no such successor shall have been so appointed by the Requisite Lenders and shall have accepted such appointment within 30 days (or such earlier day as shall be agreed by the Requisite Lenders) (the "Removal Effective Date"), then such removal shall nonetheless become effective in accordance with such notice on the Removal Effective Date.

With effect from the Resignation Effective Date or the Removal Effective Date, as applicable, (i) the retiring or removed Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by the Administrative Agent on behalf of the Lenders under any of the Loan Documents, the retiring or removed Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (ii) except for any indemnity payments owed to the retiring or removed Administrative Agent, all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender directly, until such time as the Requisite Lenders appoint a successor Administrative Agent as provided for above in this Section 10.8. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired or removed) Administrative Agent (other than any rights to indemnity payments owed to the retiring or removed Administrative Agent), and the retiring or removed Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the Resignation Effective Date or the Removal Effective Date, as applicable, the provisions of this Article X and Section 11.3 shall continue in effect for the benefit of such retiring or removed Administrative Agent, its sub –agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while th

10.9 Non-Reliance on the Administrative Agent and Other Lenders. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as

it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

- 10.10 Enforcement. By its acceptance of the benefits of this Agreement and the other Loan Documents, each Lender agrees that (i) the Loan Documents may be enforced only by the Administrative Agent, acting upon the instructions or with the consent of Requisite Lenders as provided in this Agreement and (ii) no Lender shall have any right individually to enforce or seek to enforce this Agreement or the other Loan Documents or to realize upon any collateral or other security given to secure the payment and performance of the Obligations.
- 10.11 No Other Duties, etc. Anything herein to the contrary notwithstanding, the Administrative Agent shall not have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent or a Lender hereunder.
- 10.12 No Reliance on the Administrative Agent's Customer Identification Program. Each Lender acknowledges and agrees that neither such Lender, nor any of its Affiliates, participants or assignees, may rely on the Administrative Agent to carry out such Lender's, Affiliate's, participant's or assignee's customer identification program, or other obligations required or imposed under or pursuant to the USA Patriot Act or the regulations thereunder, including the regulations contained in 31 CFR 103.121 (as hereafter amended or replaced, the "CIP Regulations"), or any other Anti-Terrorism Law, including any programs involving any of the following items relating to or in connection with any of the Borrower, their Affiliates or their agents, the Loan Documents or the transactions hereunder or contemplated hereby: (i) any identity verification procedures, (ii) any recordkeeping, (iii) comparisons with government lists, (iv) customer notices or (v) other procedures required under the CIP Regulations or such other Laws.

XI. MISCELLANEOUS

- 11.1 Modifications, Amendments or Waivers. With the written consent of the Requisite Lenders, the Administrative Agent, acting on behalf of all the Lenders, and the Borrower may from time to time enter into written agreements amending or changing any provision of this Agreement or any other Loan Document or the rights of the Lenders or the Borrower hereunder or thereunder, or may grant written waivers or consents hereunder or thereunder. Any such agreement, waiver or consent made with such written consent shall be effective to bind all the Lenders and the Borrower; provided, that no such agreement, waiver or consent may be made that will:
 - (a) extend or increase the Term Loan Commitment of any Lender (or reinstate any obligation to make the Term Loans terminated pursuant to Section 9.2) without the written consent of such Lender whose Term Loan Commitment is being extended or

increased (it being understood and agreed that a waiver of any condition precedent set forth in Section 4.2 or 4.3 or of any Default, Event of Default, mandatory prepayment or a mandatory reduction in Term Loan Commitments is not considered an extension or increase in Term Loan Commitments of any Lender);

- (b) postpone any date fixed by this Agreement or any other Loan Document for any payment (excluding mandatory prepayments of principal, interest, fees or other amounts due to the Lenders (or any of them) or any scheduled or mandatory reduction of the Term Loan Commitments hereunder or under any other Loan Document) without the written consent of each Lender entitled to receive such payment or whose Term Loan Commitments are to be reduced, it being understood that the waiver of any mandatory prepayment of the Term Loans (or any definition relating thereto) shall not constitute a postponement of any date scheduled for the payment of principal or interest;
- (c) reduce the principal of, or the rate of interest specified herein on, any Term Loan or any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby; provided, however, that only the consent of the Requisite Lenders shall be necessary (A) to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest at the Default Rate or (B) to amend any financial covenant hereunder (or any defined term used therein) even if the effect of such amendment would be to reduce the rate of interest on the Term Loans or to reduce any fee payable hereunder;
- (d) change Section 2.11 in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender directly affected thereby;
- (e) change any provision of this <u>Section 11.1</u> or the definition of "Requisite Lenders" or otherwise change the percentage of the Lenders which shall be required for the Lenders to take any action hereunder without the written consent of each Lender directly affected thereby; or
 - (f) release the Borrower without the consent of each Lender;

provided that (i) no agreement, waiver or consent that would modify the interests, rights or obligations of the Administrative Agent may be made without the written consent of such Administrative Agent, and (ii) only the consent of the Administrative Agent and the Borrower shall be required for any amendment to the Fee Letter; and provided, further that, if in connection with any proposed waiver, amendment or modification referred to in Sections 11.1(a) through 11.1(f) above, the consent of the Requisite Lenders is obtained but the consent of one or more of such other Lenders whose consent is required is not obtained (each a "Non-Consenting Lender"), then the Borrower shall have the right to replace any such Non-Consenting Lender with one or more replacement Lenders pursuant to Section 3.6.

Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder (and any amendment,

waiver or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than Defaulting Lenders), except that (x) the Term Loan Commitment of such Defaulting Lender may not be increased or extended without the consent of such Lender or amounts due to it permanently reduced without the consent of such Lender, and (z) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that by its terms affects such Defaulting Lender disproportionately adversely relative to other affected Lenders (other than solely as a result of differing amounts of the Term Loans held by such Defaulting Lender vis-à-vis the other Lenders) shall require the consent of such Defaulting Lender.

11.2 No Implied Waivers; Cumulative Remedies. No course of dealing and no delay or failure of the Administrative Agent or any Lender in exercising any right, power, remedy or privilege under this Agreement or any other Loan Document shall affect any other or future exercise thereof or operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any further exercise thereof or of any other right, power, remedy or privilege. The rights and remedies of the Administrative Agent and the Lenders under this Agreement and any other Loan Documents are cumulative and not exclusive of any rights or remedies that they would otherwise have.

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Borrower shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent for the benefit of the Lenders; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) any Lender from exercising setoff rights in accordance with Section 9.2 (subject to the terms of Section 2.12), or (c) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to the Borrower under any Debtor Relief Law.

11.3 Expenses; Indemnity; Damage Waiver.

(a) Costs and Expenses. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Administrative Agent and its Affiliates (including the reasonable fees, charges and disbursements of a single external counsel for the Administrative Agent) in connection with the syndication of the Term Loan Facility, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents, or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), and (ii) all out of pocket expenses incurred by the Administrative Agent or any Lender (including the reasonable fees, charges and disbursements of a single counsel for the Administrative Agent and the Lenders (selected by the Administrative Agent) plus one additional counsel if any Lender reasonably determines that due to a conflict its interests are better represented by separate counsel), in connection with the enforcement or protection of its rights (A) in connection with this

Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with the Term Loans made hereunder, including all such out of pocket expenses incurred during any workout, restructuring or negotiations in respect of such Term Loans.

- (b) Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof) and each Lender, and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, liabilities and related expenses (including the reasonable fees, charges and disbursements of a single counsel for the Indemnitees (selected by the Administrative Agent) plus one additional counsel if any Indemnitee reasonably determines that due to a conflict its interests are better represented by separate counsel) incurred by any Indemnitee or asserted against any Indemnitee by any Person (including the Borrower) other than such Indemnitee and its Related Parties arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereby, the performance by the parties hereto of their respective obligations hereunder or the consummation of the transactions contemplated hereby or thereby, (ii) any Term Loan or the use or proposed use of the proceeds therefrom, (iii) any actual or alleged presence or Release of Specified Substances on or from any property owned or operated by the Borrower or any of its Subsidiaries, or any liability under Environmental Laws related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by the Borrower, and regardless of whether any Indemnitee is a party thereto; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by the Borrower against an Indemnitee for breach in bad faith of such Indennitee's obligations hereunder or under any other Loan Document, if the Borrower has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction. This Section 11.3(b) shall not apply with respect to Taxes other than any Taxes that represent losses, claims, damages and other similar amounts arising from any non-Tax claim.
- (c) <u>Reimbursement by Lenders</u>. To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under paragraph (a) or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof) or any Related Party of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent) or such Related Party, as the case may be, such Lender's pro rata share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought based on each Lender's Pro Rata Share at such time) of such unpaid amount (including any such unpaid amount in respect of a claim asserted by such Lender); <u>provided</u>, that the unreimbursed expense or indemnified loss, claim, damage,

liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent) in connection with such capacity.

- (d) <u>Waiver of Consequential Damages, Etc.</u> To the fullest extent permitted by applicable Law, the Borrower shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Term Loan or the use of the proceeds thereof. No Indemnitee referred to in <u>Section 11.3</u> shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby.
 - (e) Payments. All amounts due under this Section shall be payable not later than ten (10) days after demand therefor.
- (f) <u>Survival</u>. Each party's obligations under this <u>Section 11.3</u> shall survive the resignation of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Term Loan Commitments, the termination of this Agreement and the repayment, satisfaction or discharge of all obligations under any Loan Document.

11.4 Notices; Effectiveness; Electronic Communication .

- (a) Notices Generally. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in paragraph (b) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile (i) if to a Lender, at its address (or facsimile number) set forth in its Administrative Questionnaire or (ii) if to any other Person, to it at its address (or facsimile number) set forth on Schedule 1.1(B). Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications, to the extent provided in paragraph (b) below, shall be effective as provided in said paragraph (b).
- (b) <u>Electronic Communications</u>. Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communication (including e –mail and Internet or intranet websites) pursuant to procedures approved by the

Administrative Agent, <u>provided</u> that the foregoing shall not apply to notices to any Lender pursuant to <u>Article II if</u> such Lender has notified the Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient, at its e-mail address as described in the foregoing clause (i), of notification that such notice or communication is available and identifying the website address therefor; provided that, for both clauses (i) and (ii) above, if such notice, email or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient.

(c) $\underline{\text{Change of Address, etc}}$. Any party hereto may change its address or facsimile number for notices and other communications hereunder by notice to the other parties hereto.

(d) Pl atform.

- (i) The Borrower agrees that the Administrative Agent may, but shall not be obligated to, make the Communications (as defined below) available to the other Lenders by posting the Communications on Debt Domain, Intralinks, Syndtrak or a substantially similar electronic transmission system (the "Platform").
- (ii) The Platform is provided "as is" and "as available." The Agent Parties (as defined below) do not warrant the adequacy of the Platform and expressly disclaim liability for errors or omissions in the Communications. No warranty of any kind, express, implied or statutory, including, without limitation, any warranty of merchantability, fitness for a particular purpose, non-infringement of third-party rights or freedom from viruses or other code defects, is made by any Agent Party in connection with the Communications or the Platform. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to the Borrower, any Lender or any other Person or entity for damages of any kind, including, without limitation, direct or indirect, special, incidental or consequential damages, losses or expenses (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of communications through the Platform. "Communications" means, collectively, any notice, demand, communication, information, document or other material provided by or on behalf the Borrower pursuant to any Loan Document or the transactions contemplated therein which is distributed to

the Administrative Agent or any Lender by means of electronic communications pursuant to this Section, including through the Platform.

- 11.5 Severability. The provisions of this Agreement are intended to be severable. If any provision of this Agreement shall be held invalid or unenforceable in whole or in part in any jurisdiction, such provision shall, as to such jurisdiction, be ineffective to the extent of such invalidity or unenforceability without in any manner affecting the validity or enforceability thereof in any other jurisdiction or the remaining provisions hereof in any jurisdiction.
- 11.6 <u>Duration; Survival</u>. All representations and warranties of the Borrower contained herein or made in connection herewith shall survive the execution and delivery of this Agreement, the completion of the transactions hereunder, the termination of this Agreement and Payment In Full. All covenants and agreements of the Borrower contained herein relating to the payment of principal, interest, premiums, additional compensation or expenses and indemnification, including those set forth in the Term Loan Notes, <u>Sections 2.9.</u> 2.10. 2.11 and 11.3 or in any other provision of any Loan Document, shall survive Payment In Full and shall protect the Administrative Agent, the Lenders and any other Indemnitees against events arising after such termination as well as before. All other covenants and agreements of the Borrower shall continue in full force and effect from and after the date hereof and until Payment In Full.

11.7 Successors and Assigns.

- (a) <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of this Section, (ii) by way of participation in accordance with the provisions of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Term Loan Commitment and the Term Loan at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:
 - (i) Minimum Amounts.

(A) in the case of an assignment of the entire remaining amount of the assigning Lender's Term Loan Commitment and/or the Term Loans at the time owing to it or contemporaneous assignments to related Approved Funds that equal at least the amount specified in clause (B) below in the aggregate or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and

- (B) in any case not described in clause (i)(A) of this clause (b), the aggregate amount of the Term Loan Commitments (which for this purpose includes Term Loans outstanding thereunder) or, if the applicable Term Loan Commitment is not then in effect, the principal outstanding balance of the Term Loan owed to the assignment subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "**Trade Date**" is specified in the Assignment and Assumption, as of the Trade Date) shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed).
- (ii) Proportionate Amounts . Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Term Loan or the Term Loan Commitment assigned.
- $(iii) \textit{Required Consents} \ . \ No \ consent \ shall \ be \ required \ for \ any \ assignment \ except \ to \ the \ extent \ required \ by \ paragraph \ (b)(i)(B) \ of \ this \ Section \ and \ in \ addition:$
- (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (x) an Event of Default has occurred and is continuing at the time of such assignment or (y) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; provided that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof and provided, further, that the Borrower's consent shall not be required during the primary syndication of the Term Loan Facility; and
- (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of (i) any unfunded Term Loan Commitments if such assignment is to a Person that is not a Lender with a Term Loan Commitment, an Affiliate of such Lender or an Approved Fund with respect to such Lender, or (ii) any Term Loan to a Person who is not a Lender, an Affiliate of a Lender or an Approved Fund.
- (iv) Assignment and Assumption . The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together

with a processing and recordation fee of \$3,500; provided that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.

- (v) No Assignment to Certain Persons. No such assignment shall be made to (i) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (ii) to any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute any of the foregoing Persons described in this clause (v).
 - (vi) No Assignment to Natural Persons . No such assignment shall be made to a natural Person.
- (vii) Certain Additional Payments. In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Term Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent and each other Lender hereunder (and interest accrued thereon), and (y) acquire (and fund as appropriate) its full pro rata share of all Term Loans in accordance with its Pro Rata Share. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 2.4, 3.1, 3.2 (subject to the requirements and limitations of Section 3.2) and 11.3 with respect to facts and circumstances occurring prior to the effective date of such assignment; provided, that except to the extent otherwise expressly agreed by the affected parties, no assignment by a Defaulting Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does

comply with this Section shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 11.7(d) below.

- (c) <u>Register</u>. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at one of its offices in Greenwood Village, Colorado a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Term Loan Commitments of, and principal amounts of (and stated interest on) the Term Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Term Loan Commitment and/or the Term Loans owing to it); <u>provided</u> that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent and the Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. For the avoidance of doubt, each Lender shall be responsible for the indemnity under <u>Section 11.3(c)</u> with respect to any payments made by such Lender to its Participant(s). A sale of a participation interest may include certain voting rights, which unless such participant is a Voting Participant, shall be limited to significant matters such as (A) increases in the commitment of such participant, (B) reductions of interest rates, principal or fees, (C) extensions of scheduled maturities or times for payment or (D) reductions in voting percentages, in each case directly affecting such participant.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver with respect to Section 11.1 that affects such Participant. The Borrower agrees that each Participant shall be entitled to the benefits of Section 3.1, 3.5 and 3.2 (subject to the requirements and limitations therein, including the requirements under Section 3.2 (it being understood that the documentation required under Section 3.1 shall be delivered to the participating Lender)) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to paragraph (b) of this Section 11.7; provided that such Participant (A) agrees to be subject to the provisions of Section 3.6 as if it were an assignee under paragraph (b) of this Section 11.7; and (B) shall not be entitled to receive any greater payment under Section 3.1 or 3.2, with respect to any participation, than its

participating Lender would have been entitled to receive, except in the case of a Voting Participant only, to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation. Each Lender that sells a participation agrees, at the Borrower's request and expense, to use reasonable efforts to cooperate with the Borrower to effectuate the provisions of Section 3.6 with respect to any Participant. To the extent permitted by Law, each Participant also shall be entitled to the benefits of Section 9.2(c) as though it were a Lender; provided that such Participant agrees to be subject to Section 2.12 as though it were a Lender. Each Lender that sells a participation shall, acting solely for this purpose as an agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts of (and stated interest on) each Participant's interest in the Term Loans or other obligations under the Loan Documents (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any Term Loan Commitments, the Term Loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such Term Loan Commitment, the Term Loans or other obligation is in registered form under Section 5f.103 –1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maint

Notwithstanding the preceding paragraph, any Participant that is a Farm Credit Lender that (i) has purchased a participation in a minimum amount of \$5,000,000, (ii) has been designated as a voting Participant (a "Voting Participant") in a notice (a "Voting Participant Notice") sent by the relevant Lender (including any existing Voting Participant) to the Borrower and the Administrative Agent and (iii) receives, prior to becoming a Voting Participant, the prior written consent of the Borrower (unless an Event of Default has occurred and is continuing) the consent of the Administrative Agent (such consent to be required only to the extent and under the circumstances it would be required if such Voting Participant were to become a Lender pursuant to an assignment in accordance with Section 11.7(b) and such consent is not required for an assignment to an existing Voting Participant), shall be entitled to vote as if such Voting Participant were a Lender on all matters subject to a vote by Lenders, and the voting rights of the selling Lender (including any existing Voting Participant) shall be correspondingly reduced, on a dollar-for-dollar basis. Each Voting Participant Notice shall include, with respect to each Voting Participant, the information that would be included by a prospective Lender in an Assignment and Assumption. Notwithstanding the foregoing, each Farm Credit Lender designated as a Voting Participant in Schedule 11.7 shall be a Voting Participant of the Administrative Agent. The selling Lender (including any existing Voting Participant) and the purchasing Voting Participant shall notify the Administrative Agent within three (3) Business Days of any termination, reduction or increase of the amount of, such participant Notices and all other

notices delivered pursuant hereto. The voting rights of each Voting Participant are solely for the benefit of such Voting Participant and shall not inure to any assignee or participant of such Voting Participant that is not a Farm Credit Lender.

- (e) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank or other central bank; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto.
- 11.8 Confidentiality. Each of the Administrative Agent and the Lenders agree to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its Related Parties (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (b) to the extent required or requested by any regulatory authority purporting to have jurisdiction over such Person or its Related Parties (including any self-regulatory authority, such as the National Association of Insurance Commissioners); (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process; (d) to any other party hereto; (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder; (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights and obligations under this Agreement, or (ii) any actual or prospective party (or its Related Parties) to any swap, derivative or other transaction under which payments are to be made by reference to the Borrower and its obligations, this Agreement or payments hereunder; (g) on a confidential basis to (i) any rating agency in connection with rating the Borrower or its Subsidiaries or the Term Loan Facility, (ii) the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the Term Loan Facility or (iii) to any credit insurance provider relating to the Borrower and its Obligations; (h) with the consent of the Borrower; or (i) to the extent such Information (x) becomes publicly available other than as a result of a breach of

For purposes of this Section, "Information" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any of its Subsidiaries; provided that, in the case of information received from the Borrower or any of its Subsidiaries after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

11.9 Counterparts; Integration; Effectiveness.

- (a) This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Article IV, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Agreement.
- (b) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

11.10 CHOICE OF LAW; SUBMISSION TO JURISDICTION; WAIVER OF VENUE; SERVICE OF PROCESS; WAIVER OF JURY TRIAL.

- (a) <u>Governing Law</u>. This Agreement and the other Loan Documents and any claims, controversy, dispute or cause of action (whether in contract or tort or otherwise) based upon, arising out of or relating to this Agreement or any other Loan Document (except, as to any other Loan Document, as expressly set forth therein) and the transactions contemplated hereby and thereby shall be governed by, and construed in accordance with, the law of the State of New York without regard to conflicts of law principles that require or permit application of the laws of any other state or jurisdiction.
- (b) <u>SUBMISSION TO JURISDICTION</u>. THE BORROWER IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NONEXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT,

AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH NEW YORK STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR IN ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT THE ADMINISTRATIVE AGENT OR ANY LENDER MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST THE BORROWER OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.

- (c) <u>WAIVER OF VENUE</u>. THE BORROWER IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY COURT REFERRED TO IN THIS <u>SECTION 11.10</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT AND AGREES NOT TO ASSERT ANY SUCH DEFENSE.
- (d) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 11.4</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.
- (e) <u>WAIVER OF JURY TRIAL</u>. EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, ADMINISTRATIVE AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN

DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

- 11.11 USA Patriot Act Notice. Each Lender that is subject to the USA Patriot Act and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act, it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow such Lender or Administrative Agent, as applicable, to identify the Borrower in accordance with the USA Patriot Act. The Borrower shall, promptly following a request by the Administrative Agent or any Lender, provide all documentation and other information that the Administrative Agent or such Lender requests in order to comply with its ongoing obligations under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act.
- 11.12 Payments Set Aside. To the extent the Borrower makes a payment or payments to the Administrative Agent for the ratable benefit of the Lenders or the Administrative Agent receives any such payment which payments or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside and/or required to be repaid to a trustee, receiver or any other party under any proceeding under any Debtor Relief Law, other applicable Law or equitable cause, then, to the extent of such payment or proceeds repaid, the Obligations or part thereof intended to be satisfied shall be revived and continued in full force and effect as if such payment or proceeds had not been received by the Administrative Agent.
- 11.13 Interest Rate Limitation. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law (the "Maximum Rate"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Term Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

IN WITNESS WHEREOF, the parties hereto, by their officers thereunto duly authorized, have executed this Agreement as of the day and year first above written.

BORROWER:

FRONTIER COMMUNICATIONS CORPORATION, as Borrower

By: /s/ John M. Jureller
Name: John M. Jureller
Title: Executive Vice President and Chief Financial Officer

COBANK, ACB , as Administrative Agent and as a Lender

By: /s/ Gary Franke
Name: Gary Franke
Title: Vice President

BNP PARIBAS, as a Lender

By: /s/ Kwang Kyun Choi Name: Kwang Kyun Choi Title: Vice President

By: /s/ Maria Mulic Name: Maria Mulic, CFA Title: Vice President

FIFTH THIRD BANK, AN OHIO BANKING CORPORATION, as a Lender

By: <u>/s/ Valerie Schanzer</u> Name: Valerie Schanzer Title: V.P.

COMPASS BANK, as a Lender

By: /s/ Raj Nambiar Name: Raj Nambiar Title: Vice President

KEYBANK NATIONAL ASSOCIATION, as a Lender

By: /s/ Philip Muscato
Name: Philip Muscato
Title: Senior Vice President

TD BANK, N.A., as a Lender

By: <u>/s/ David Perlman</u> Name: David Perlman Title: Senior Vice President

THE ROYAL BANK OF SCOTLAND PLC, as a Lender

By: /s/ Alex Daw Name: Alex Daw Title: Director

SCHEDULE 1.1(B) C OMMITMENTS OF LENDERS AND ADDRESSES FOR NOTICES

Part 1 - Commitments of Lenders and Addresses for Notices to Lenders

LENDER AND ADDRESS	AMOUNT OF COMMITMENT FOR TERM LOANS	PRO RATA SHARE
COBANK, ACB S500 SOUTH QUEBEC STREET GREENWOOD VILLAGE COLORADO 80111 ATTENTION: COMMUNICATIONS BANKING GROUP TELEPHONE: 303-740-4000 TELECOPY: 303-224-2718	\$135,000,000	38.57%
BNP PARIBAS 787 SEVENTH AVENUE NEW YORK, NY 10019 ATTENTION: MARIA MULIC TELECOPY: 212-471-6368	\$50,000,000	14.29%
FIFTH THIRD BANK, AN OHIO BANKING CORPORATION 38 FOUNTAIN SQUARE PLAZA CINCINNATI, OH 45263 ATTENTION: VALERIE SCHANZER TELECOPY: 212-299-3036	\$50,000,000	14.29%
COMPASS BANK 2200 POST OAK BLVD., #1500 HOUSTON, TX 77056 ATTENTION: DENISE WOLFENBERGER TELECOPY: 866-984-8668	\$30,000,000	8.57%
KEYBANK NATIONAL ASSOCIATION 1700 BAUSCH & LOMB PLACE ROCHESTER, NY 14604 ATTENTION: PHIL MUSCATO TELECOPY: 585-238-4142	\$30,000,000	8.57%

Exhibit -F - 3

TD BANK, N.A.
444 MADISON AVE., 2 ND
FLOOR
NEW YORK, NY 10022
ATTENTION: SHIVANI
AGARWAL
TELECOPY: 212-308-0486

\$30,000,000

8.57%

THE ROYAL BANK OF SCOTLAND PLC 600 WASHINGTON BLVD. STAMFORD, CT 06901 ATTENTION: ALEX DAW TELECOPY: 203-873-3426

\$25,000,000 7.14%

¹The above amount reflects the gross commitment of CoBank, ACB, of which \$59,165,000 will be participated within the Farm Credit System to eleven voting and non-Voting Farm Credit System lending institutions.

$\begin{array}{c} \textbf{SCHEDULE 1.1(B)} \\ \textbf{C OMMITMENTS OF LENDERS AND ADDRESSES FOR NOTICES} \end{array}$

Part 2 - Addresses for Notices to Borrower and Guarantors:

A DMINISTRATIVE AGENT : CoBank, ACB 5500 South Quebec Street Greenwood Village Colorado 80111

Attention: Communications Banking Group Telephone: (303) 740-4000 Telecopy: (303) 224-2718

BORROWER:
Frontier Communications Corporation
3 High Ridge Park
Stamford, CT 06905
Attention: John M. Jureller, Executive Vice President and Chief Financial Officer
Telephone: (203) 614-5995
Telecopy: (203) 614-4661

SCHEDULE 7.7

Subsidiary Indebtedness

As of March 31, 2014

(\$ in Thousands)

Hron	itier	Ind	eht	tedness

Citizens Rural Company Rural FFB 6.052% due Jan. 3, 2028 2,999 5,827 FFB 6.206% due Jan. 3, 2028

Total Frontier Subsidiary Indebtedness 8,826

Spinco Assumed Indebtedness

Verizon North Inc.

6.73% Debentures, Series G, due Feb. 15, 2028 200,000

Verizon West Virginia Inc.

8.40% Debentures due Oct. 15, 2029 50,000

Total Assumed Spinco Subsidiary Indebtedness 250,000

Capital Lease Obligations
Citizens Telecommunications Company of Minnesota,
LLC

NESC Minnesota (fiber ring) Frontier Communications Northwest Inc.

1800 41st, Everett, Washington 18,446

6,269

Total Capital Lease Obligations 24,715

Finance Lease Obligations

Obligations in connection with the contributions to limited liability companies owned by the Frontier Communications Pension Plan (a) made in 2011 by Frontier North Inc., Commonwealth Telephone Company LLC and Frontier Communications Corporation Corporate Services Inc., and consisting of properties located at 100 Communications Drive, Sun Prairie, WI; Country Club Road, Dallas, PA; 6430 Oakbrook Parkway, Fort Wayne IN and 8001 West Jefferson Blvd, Fort Wayne, IN and (b) made in 2013 by Frontier North Inc., Frontier Communications of the Carolina LLC, Frontier West Virginia Inc. and Citizens Telecommunications Company of California Inc., and consisting of properties located at 601 N US 131, Three Rivers, MI; 400 S Pike RD W, Sumter, SC; 1500 Maccorkle Ave SE, Charleston, WV and 9260 E. Stockton Blvd., Elk Grove, CA

79,698

Total Finance Lease Obligations

79,698

Total Subsidiary Indebtedness

363,239

SCHEDULE 11.7 VOTING PARTICIPANTS

AgStar Financial Services, FLCA 1921 Premier Drive Mankato, MN 56001

Badgerland Financial, FLCA 1430 North Ridge Dr. Prairie du Sac, WI 53578-1092

Farm Credit Bank of Texas 4801 Plaza on the Lake Drive Austin, TX 78746

Farm Credit Services of America, FLCA 5015 S. 118 $^{\rm th}$ Street Omaha, NE 68137

Farm Credit Mid-America, FLCA 1601 Ups Drive Louisville, KY 40223

United FCS, FLCA d/b/a FCS Commercial Finance Group 600 South Highway 169, Suite 850 Minneapolis, MN 55426





January 15, 2014

Mark Nielsen 3 Parley Lane Ridgefield CT 06877

Dear Mark,

It is my pleasure to confirm our offer of employment for the position of Senior Vice President, General Counsel & Secretary. The work location for this position will be Stamford, CT and you will be reporting to Maggie Wilderotter, Chairman and Chief Executive Officer. Your start date will be March 4, 2014.

Your executive compensation program includes four principle components:

- 1) Annual base salary of \$ 325,000 (less applicable taxe s) paid on a semi-monthly basis.
- 2) Cash bonus under the Frontier Bonus Plan ("FBP") with an annual target incentive of 75 % of your annual base salary (initially \$ 243,750) which will be paid out based on company and personal performance. Your first year payout will be prorated based on your start date.
- 3) Restricted stock awards , which are generally granted in the first quarter of each year. Restricted shares vest in three equal annual installments (33.3 % per year) , commencing one year from the date of grant , with one- third of the r estricted s hares vesting and becoming unrestricted upon each annual anniversary date . The annual target for your position is currently \$450,000 . You will receive dividends , to the extent dividends are declared by the Board of Directors, on all shares , whether vested or un vested . The company's current annual dividend policy (which is subject to change at the discretion of the Board of Directors) is \$0.40 per share , declared quarterly in an amount of \$0.10 per share per quarter. Your first year payout will be prorated based on your start date.
- 4) **Performance Shares** under the Frontier Long Term Incentive Plan (LTIP), with an annual performance share target valued at one-third of your annual target for restricted stock awards. The LTIP target is an annual grant that will be paid out based on company performance over a three year period (initially, 2014 2016). The performance metrics (which are subject to change for future grants) are currently Operating Cash. Flow and Total Shareholder Return. You will earn performance shares at the end of each three year period based on the company's performance over the three-year measurement period on the metrics for that award. Dividends that would have been payable on any earned shares during a three year period will be paid at the end of the three year period. Your award for the 2014-2016 measurement period will be prorated based on your start date.

You will be eligible for a cash bonus and restricted stock award for 201 4 with awards paid in the first quarter of 201 5. You will be eligible for an award under the LTIP commencing with the 201 4-201 6 performance period. Frontier reserves the right to implement or discontinue executive compensation plans at its own discretion. Eligibility for any given plan does not guarantee award values since Frontier's

Executive Compensation Program is based on performance of the company and the executive. Further, awards are subject to the terms and conditions of Frontier's compensation plans

As a sign-on bonus, you will be granted **3 0,000 restricted shares** (subject to the approval of the Compensation Committee) as part of your new hire compensation. These shares will vest in three equal annual installments (33.3 % per year), commencing one year from your start date.

If, within one year following a "Change in Control" (as defined below) of the Company, you have a "Separation from Service" (as defined below) either because (a) your employment is terminated by the Company without "Cause" (as defined below) or (b) you terminate your employment as a result of (i) a material decrease in your base salary, target bonus or long term incentive compensation target from those in effect immediately prior to the Change in Control for any reason other than Cause; (ii) a material relocation of your principal office location (for this purpose, a relocation more than 50 miles from the Company's Stamford, Connecticut headquarters will be automatically deemed material) or (iii) a material decrease in your responsibilities or authority for any reason other than Cause (and prior to your terminating your employment you provide the Company with notice of the decrease or relocation within 90 days of the occurrence of such conditions), you shall be entitled to a lump sum payment equal to one year's base salary and 100% of your bonus target prorated for the plan year (based on the then current level of salary and bonus target or, if greater, that in effect immediately prior to the Change in Control), all restrictions on restricted shares held by you shall immediately lapse and such restricted shares shall become fully-vested and non-forfeitable and all performance shares granted to you under the Long-Term Incentive Plan (LTIP) or other performance incentive plan pursuant to a performance-based vesting schedules shall immediately be granted. The lump sum payment will be made on the Expiration Date, as defined below.

A "Change in Control" shall be deemed to have occurred:

- (A) When any "person" as defined in Section 3(a)(9) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and as used in Section 13(d) and 14(d) thereof, including a "group" as defined in Section 13(d) of the Exchange Act (but excluding the Company and any subsidiary and any employee benefit plan sponsored or maintained by the Company or any subsidiary (including any trustee of such plan acting as trustee)), directly or indirectly, becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), of securities of the Company representing 50% or more of the combined voting power of the Company's then outstanding securities; or
- (B) Upon the consummation of any merger or other business combination involving the Company, a sale of substantially all of the Company's assets, liquidation or dissolution of the Company or a combination of the foregoing transactions (the "Transactions") other than a Transaction immediately following which the shareholders of the Company immediately prior to the Transaction own, in the same proportion, at least 51% of the voting power, directly or indirectly, of (i) the surviving corporation in any such merger or other business combination; (ii) the purchaser of or successor to the Company's assets; (iii) both the surviving corporation and the purchaser in the event of any combination of Transactions; or (iv) the parent company owning 100% of such surviving corporation, purchaser or both the surviving corporation and the purchaser, as the case may be.

"Cause" shall mean your (a) willful and continued failure (other than as a result of physical or mental illness or injury) to perform your material duties in effect immediately prior to the Change in Control which continues beyond 10 days after a written demand for substantial performance is delivered to you by the Company, which demand shall identify and describe each failure with sufficient specificity to allow

you to respond, (b) willful or intentional conduct that causes material and demonstrable injury, monetary or otherwise, to the Company or (c) conviction of, or a plea of nolo contendere to, a crime constituting (i) a felony under the laws of the United States or any State thereof, or (ii) a misdemeanor involving moral turpitude. For these purposes, no act or failure to act on your part shall be considered "willful" or "intentional" unless it is done, or omitted to be done by you in bad faith and without reasonable belief that your action or inaction was in the best interests of the Company. Any act or failure to act based upon authority given pursuant to a resolution duly adopted by the Board of Directors or based upon the advice of counsel for the Company shall be conclusively presumed to be done, or omitted to be done, by you in good faith and in the best interests of the Company.

If it is determined (as hereafter provided) that any payment or distribution by the Company to or for your benefit, whether paid or payable or distributed or distributable pursuant to the terms of this letter agreement or otherwise pursuant to or by reason of any other agreement, policy, plan, program or arrangement, including without limitation any stock option, restricted stock award, stock appreciation right or similar right, or the lapse or termination of any restriction on or the vesting or exercisability of any of the foregoing (a "Severance Payment"), would be subject to the excise tax imposed by Section 4999 of the Code (or any successor provision thereto) by reason of being "contingent on a change in ownership or control" of the Company, within the meaning of Section 280G of the Code (or any successor provision thereto) or to any similar tax imposed by state or local law, or any interest or penalties with respect to such excise tax (such tax or taxes, together with any such interest and penalties, are hereafter collectively referred to as the "Excise Tax"), then the Severance Payment shall be payable either (i) in full or (ii) as to such lesser amount which would result in no portion of the Severance Payment being subject to the Excise Tax ("Capped Payment"), whichever of the foregoing amounts, taking into account the applicable federal, state and local income taxes and the Excise Tax, results in your receipt on an after-tax basis, of the greatest amount of economic benefits to you, notwithstanding that all or some portion of such benefits may be taxable under Section 4999 of the Code.

Subject to the provisions of immediately preceding paragraph, all determinations required to be made pursuant to this letter agreement, including whether an Excise Tax is payable by you and the amount of such Excise Tax, shall be made by the nationally recognized firm of certified public accountants (the "Accounting Firm") used by the Company prior to the Change in Control (or, if such Accounting Firm declines to serve, the Accounting Firm shall be a nationally recognized firm of certified public accountants selected by you). The Accounting Firm shall be directed by the Company or you, as applicable, to submit its preliminary determination and detailed supporting calculations to both the Company and you within 15 calendar days after the date of your termination of employment, if applicable, and any other such time or times as may be requested by the Company or you. If the Accounting Firm determines that any Excise Tax is payable by you, the Company shall either (x) make payment of the Severance Payment, or (y) reduce the Severance Payment by the amount which, based on the Accounting Firm's determination and calculations, would provide you with the Capped Payment (except that any portion of the Severance Payment that constitutes deferred compensation that is subject to Section 409A shall not be reduced, and its time and form of payment shall not be altered as a result of this process), and pay to you such reduced amount, in each case, less any Excise Taxes, federal, state, and local income and employment withholding taxes and any other amounts required to be deducted or withheld by the Company under applicable statute or regulation. If the Accounting Firm hon Excise Tax is payable by you, it shall, at the same time as it makes such determination, furnish you with an opinion that you have substantial authority not to report any Excise Tax on your federal, state, local income or other tax return. All fees and expenses of the Accounting Firm and opinion letter shall be paid by the Company in connection with the cal

The provisions in this letter regarding lapsing of restrictions on restricted stock and the earning of performance shares in certain circumstances in the event of a Change in Control will remain in effect as

long as you are a member of the Senior Leadership Team ("SLT"). If at any time you are no longer a member of the SLT, such provisions will not apply.

You shall not receive any payments or benefits to which you may be entitled hereunder unless you agree to execute a release of all then existing claims against the Company, its subsidiaries, affiliates, shareholders, directors, officers, employees and agents in relation to claims relating to or arising out of your employment or the business of the Company; provided, however, that any such release shall not bar or prevent you from responding to any litigation or other proceeding initiated by a released party and asserting any claim or counterclaim you have in such litigation or other proceeding as if no such release had been given as to such party, nor shall it bar you from claiming rights that arise under, or that are preserved by, this letter agreement. To comply with this paragraph, you must sign and return the release within 45 days of the termination of your employment, and you must not revoke it during a seven-day revocation period that begins when the release is signed and returned to the Company. Then following the expiration of this revocation period, there shall occur the "Expiration Date," which is the 53rd day following the date of termination of your employment.

To the extent that a payment of Section 409A compensation under this letter agreement is based upon your having a termination of employment, "termination of employment" shall have the same meaning as "Separation from Service" under Section 409A(a)(2)(A)(i) of the Code. In addition, to avoid having such a separation from service occur after your termination of employment, you shall not have (after your termination of employment) any duties or responsibilities that are inconsistent with the termination of employment being treated as such a separation from service as of the date of such termination.

Please be advised that your health and welfare benefits will begin on your 9 1st day of employment and as a Frontier employee, you will be eligible to participate in a full range of benefits. Please bring all of the original paperwork with you on your first day of work.

This offer and subsequent employment is contingent upon Frontier's receipt of acceptable results of a background check and reference checks including, criminal record check, drug screening, and verification of education, employment and professional references. Certain positions will also require a motor vehicle or credit check. All drug screens must be completed within 48 hours from when you receive the chain of custody form. The drug test will be registered by Frontier.

Federal law requires that you provide documentation (I -9) confirming your eligibility to work in the United States. A list of documents that you may use to establish your identity and employment eligibility can be found through our electronic on-boarding tool, Red Carpet. Please bring the appropriate documents with you when you report to work on your first day.

Assuming the contingencies noted above are met and you commence employment with Frontier, as a condition of accepting this offer of employment with Frontier you agree that should you leave employment with Frontier at any time in the future for any reason, you will not solicit, either directly or indirectly, any Frontier employee for employment with any other employer for a period of one (1) year after you leave employment with Frontier; provided, however, that nothing shall prohibit you from performing, or having performed on your behalf, a general solicitation for employees not specifically focused at Frontier's employees through the use of media, advertisement, electronic job boards or other general, public solicitations.

This offer is not an express or implied contract, promise or guarantee of employment, of any particular position, or of any particular term or condition of employment. Your employment by Frontier is at will and is subject to the conditions set forth in Frontier's Code of Conduct as well as all other Frontier policies and applicable Federal, State and local laws.

On behalf of Frontier, I welcome you to our team! Please do not hesitate to contact me with any questions regarding this offer. To acknowledge your acceptance of this offer, please sign the bottom of this offer letter and fax (203-614-5054) or email a scanned copy back to me directly. Please return the original signed offer letter with your original new hire paperwork as soon as convenient.

Sincerely,

/s/ Jim Oddo

Jim Oddo VP, Talent Acquisition & Development Frontier Communications Corporation

Acceptance of Offer

By signing below, I hereby accept the Frontier's contingent offer of employment. I understand that I will not have a contract of employment with Frontier for a specified period of time. I further agree to abide by policie's and procedures established by Frontier.

/s/ Mark Nielsen Mark Nielsen

February 10, 2014 Date





June 9, 2014

D onald Daniels 47 Woodchuck Lane Ridgefield, CT 06877

Dear Donald,

It is my pleasure to confirm our offer of employment for the position of S enior V ice P resident and Controller. The work location for this position will be Stamford, CT and you will be reporting to John Jureller, E xecutive V ice P resident and C hief F inancial O fficer. Your start date will be determined upon acceptance.

Your executive compensation program includes four principle components:

- 1) Annual base salary of \$27 0,000 (less applicable taxe s) paid on a semi-monthly basis.
- 2) Cash bonus under the Frontier Bonus Plan ("FBP") with an annual target incentive of 50% of your annual base salary (initially \$135,000) which will be paid out based on company and personal performance. Your first year payout will be prorated based on your start date.
- 3) Restricted stock awards , which are generally granted in the first quarter of each year. Restricted shares vest in three equal annual installments (33.3 % per year) , commencing one year from the date of grant , with one-quarter of the r estricted shares vesting and becoming unrestricted upon each annual anniversary date . The annual target for your position is currently \$11 5 , 0 00 . Your first year payout will be prorated based on your start date. You will receive dividends , to the extent dividends are declared by the Board of Directors, on all shares , whether vested or un vested . The company's current annual dividend policy (which is subject to change at the discretion of the Board of Directors) is \$0.40 per share , declared quarterly in an amount of \$0.10 per share per quarter.
- 4) **Performance Shares** under the Frontier Long Term Incentive Plan (LTIP), with an annual performance share target of \$35,000. The LTIP target is an annual grant that will be paid out based on company performance over a three year period (initially, 201 5 201 7). The performance metrics (which are subject to change for future gran t s) are currently Operating Cash Flow and Total Shareholder Return. You will earn performance shares at the end of each three year period based on the company's performance over the three-year measurement period on the metrics for that award. Dividends that would have been payable on any earned shares during a three year period will be paid at the end of the three year period.

You will be eligible for a cash bonus and restricted stock award for 201 4 with awards paid in the first quarter of 201 5. You will be eligible for an award under the LTIP commencing with the 201 5 -201 7 performance period. Frontier reserves the right to implement or discontinue executive compensation plans at its own discretion. Eligibility for any given plan does not guarantee award values since Frontier's

Executive Compensation Program is based on performance of the company and the executive. Further, awards are subject to the terms and conditions of Frontier's compensation plans

As a sign-on bonus, you will be granted 10,000 restricted shares (subject to the approval of the Compensation Committee) as part of your new hire compensation. These shares will vest in three equal annual installments (33.3 % per year), commencing one year from your start date.

You will also receive a \$25,000 cash sign on bonus (subject to applicable taxes) contingent upon execution of a twelve (12) month prorated promissory note (attached). The sign on bonus is paid after thirty (30) days of employment.

Please be advised that your health and welfare benefits will begin on your 9 1st day of employment and as a Frontier employee, you will be eligible to participate in a full range of benefits. Please bring all of the original paperwork with you on your first day of work.

This offer and subsequent employment is contingent upon Frontier's receipt of acceptable results of a background check and reference checks including, criminal record check, drug screening, and verification of education, employment and professional references. Certain positions will also require a motor vehicle or credit check. All drug screens must be completed within 48 hours from when you receive the chain of custody form. The drug test will be registered by Frontier.

Federal law requires that you provide documentation (I -9) confirming your eligibility to work in the United States. A list of documents that you may use to establish your identity and employment eligibility can be found through our electronic on-boarding tool, Red Carpet. Please bring the appropriate documents with you when you report to work on your first day.

Assuming the contingencies noted above are met and you commence employment with Frontier, as a condition of accepting this offer of employment with Frontier you agree that should you leave employment with Frontier at any time in the future for any reason, you will not solicit, either directly or indirectly, any Frontier employee for employment with any other employer for a period of one (1) year after you leave employment with Frontier; provided, however, that nothing shall prohibit you from performing, or having performed on your behalf, a general solicitation for employees not specifically focused at Frontier's employees through the use of media, advertisement, electronic job boards or other general, public solicitations.

This offer is not an express or implied contract, promise or guarantee of employment, of any particular position, or of any particular term or condition of employment. Your employment by Frontier is at will and is subject to the conditions set forth in Frontier's Code of Conduct as well as all other Frontier policies and applicable Federal, State and local laws.

On behalf of Frontier, I welcome you to our team! Please do not hesitate to contact me with any questions regarding this offer. To acknowledge your acceptance of this offer, please sign the bottom of this offer letter and fax (203-614-5054) or email a scanned copy back to me directly. Please return the original signed offer letter with your original new hire paperwork as soon as convenient.

Sincerely,

/s/ Jim Oddo

Jim Oddo VP, Talent Acquisition & Development Frontier Communications Corporation

Acceptance of Offer

By signing below, I hereby accept the Frontier's contingent offer of employment. I understand that I will not have a contract of employment with Frontier for a specified period of time. I further agree to abide by policie s and procedures established by Frontier.

/s/ Donald Daniels Donald Daniels June 10, 2014 Date

CERTIFICATIONS

- I, Mary Agnes Wilderotter, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Frontier Communications Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2014 /s/ Mary Agnes Wilderotte

/s/ Mary Agnes Wilderotter

Mary Agnes Wilderotter

Chairman and Chief Executive Officer

CERTIFICATIONS

- I, John M. Jureller, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Frontier Communications Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the c ircumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2014 /s/ John M. Jureller

John M. Jureller

Executive Vice President and Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly Report of Frontier Communications Corporation (the "Company") on Form 10- Q for the period ended June 30, 2014 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Mary Agnes Wilderotter, Chairman and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the S ecurities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Mary Agnes Wilderotter
Mary Agnes Wilderotter
Chairman and Chief Executive Officer
August 7, 2014

This certification is made solely for purpose of 18 U.S.C. Section 1350, subject to the knowledge standard contained therein, and not for any other purpose.

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Frontier Communications Corporation and will be retained by Frontier Communications Corporation and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly Report of Frontier Communications Corporation (the "Company") on Form 10- Q for the period ended June 30, 2014 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, John M. Jureller, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ John M. Jureller

John M. Jureller Executive Vice President and Chief Financial Officer August 7, 2014

This certification is made solely for purpose of 18 U.S.C. Section 1350, subject to the knowledge standard contained therein, and not for any other purpose.

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Frontier Communications Corporation and will be retained by Frontier Communications Corporation and furnished to the Securities and Exchange Commission or its staff upon request.